UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (date of earliest event reported): June 10, 2022 Electro-Sensors, Inc.

(Exact name of Registrant as Specified in its Charter)

	N	Iinnesota		
	(State Or Other Juri	sdiction Of Incorporation)		
000-09587			41-0943459	
(Commission File Number)			(I.R.S. Employer Identification No.)	
6111 Blue Circle Drive				
Minnetonka, MN			55343-9108	
(Address of Principal Executive Offices))		(Zip Code)	
	(95	2) 930-0100		
R	Registrant's Telephone	Number, Including Area Co	de	
Check the appropriate box below if the Form 8-following provisions (see General Instruction A		nultaneously satisfy the filing o	bligation of the registrant under any of the	
Written communications pursuant to Rule 425 under the Securities Act ■				
☐ Soliciting material pursuant to Rul	e 14a-12 under the Excha	ange Act		
☐ Pre-commencement communication	ons pursuant to Rule 14d-	2(b) under the Exchange Act		
Pre-commencement communication	ons pursuant to Rule 13e-	4(c) under the Exchange Act		
Securities registered pursuant to Section 12(b)	of the Act			
Title of Each Class	Trading Symbol	Name of each exchange	on which registered	
Common stock	ELSE	The Nasdaq Capital Marl	ket LLC	
Indicate by check mark whether the registrant is or Rule 12b-2 of the Securities Exchange Act of			f the Securities Act of 1933 (17 CFR §230.405)	
Emerging Growth Company □				
If an emerging growth company, indicate by che	eck mark if the registrant	has elected not to use the exter	ided transition period for complying with any	

new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01 Entry Into a Material Definitive Agreement

On June 10, 2022, Electro-Sensors, Inc. ("ELSE" or "Electro-Sensors") entered into an Agreement and Plan of Merger (the "Merger Agreement") with Mobile X Newco, Inc., a Delaware corporation and wholly owned subsidiary of ELSE (the "Merger Sub"), and Mobile X Global, Inc., a Delaware corporation ("Mobile X") (together with ELSE and Merger Sub, the "Parties"), pursuant to which Merger Sub will merge with and into Mobile X, with Mobile X surviving the merger as a wholly owned subsidiary of ELSE (the "Merger").

Mobile X Global, Inc. is a new entrant in the global mobile industry founded by Peter Adderton. Mobile X plans to launch a new mobile brand called Mobile X in the U.S. in 2022, enabled by a network access agreement with a major carrier.

The Merger is structured as a statutory reverse triangular merger under Delaware and Minnesota law, under which a newly formed subsidiary of ELSE will be merged with and into Mobile X Global, Inc., with Mobile X Global, Inc. surviving the Merger and becoming a wholly owned subsidiary of ELSE. In connection with the Merger, ELSE will reincorporate in Delaware, be re-named Mobile X Global, Inc., and operate both the new MobileX wireless business and the Electro-Sensors business. The Merger Agreement also provides that Electro-Sensors, Inc. will effect a four-for-one reverse stock split shortly before completion of the Merger, unless the Parties agree on a different reverse split ratio.

In connection with the execution of the Merger Agreement, a third-party institutional investor has entered into a commitment letter with Mobile X Global, Inc. to provide equity financing of up to \$20.0 million upon closing of the Merger in the form of convertible preferred stock. The commitment is subject to diligence and definitive agreements satisfactory to the third-party institutional investor, including an agreement for a \$50 million equity line of credit to be provided by the investor. The equity line of credit would provide significant additional liquidity, at the option of Mobile X. The commitment letter terminates on October 31, 2022.

It is anticipated that, immediately after the Merger, former Mobile X stockholders will own approximately 76% of the combined company, legacy ELSE shareholders will own approximately 11% of the combined company, and that the third-party institutional investor noted above (or an alternative investor agreed to by the Parties to the Merger Agreement) and certain other investors will own approximately 13% of the combined company, all based on current ownership of ELSE and Mobile X, and assuming the closing of \$20 million of equity financing on the terms contained in the commitment letter from the third-party institutional investor noted above.

ELSE expects the approximately 332,500 shares subject to currently outstanding options to be exercised in connection with the Merger prior to the record date of the cash dividend discussed below. Assuming this exercise, and based on the relative valuations agreed to by Mobile X and ELSE, and after giving effect to the reverse split at a ratio of four-for-one, the legacy ELSE shareholders would continue to own approximately 932,005 shares, the Mobile X stockholders would own approximately 6,668,294 shares, the third-party institutional investor noted above (or an alternative investor agreed to by the Parties to the Merger Agreement) would own approximately 1,066,860 shares (on an as-converted to common basis) assuming the closing of \$20 million of equity financing on the terms in the commitment letter, and approximately 75,851 shares would be held by others.

In addition to their continuing interest in the combined company, legacy Electro-Sensors shareholders as of a record date to be determined before the closing of the Merger will receive special cash dividends expected to be approximately \$18.0 million in the aggregate, with the amount of the dividends possibly adjusted based on the amount of ELSE transaction expenses and its working capital balance at the closing of the Merger, and further adjusted for indebtedness, if any, and transaction bonuses, if any, approved by the ELSE board of directors. Aggregate cash dividends of \$18.0 million would be approximately \$4.83 per share based on the current, pre-reverse split, fully diluted shares of Electro-Sensors.

3 of 6

Closing of the Merger is subject to specified conditions, including, among other matters: (i) the approval by Mobile X stockholders and ELSE shareholders of the Merger; (ii) a registration statement becoming effective under the Securities Act of 1933, as amended, related to the shares being issued to the Mobile X stockholders in the Merger and the clearance of the proxy statement related to the approval by the ELSE shareholders of the Merger; (iii) receipt of \$20 million in third party equity financing; (iv) listing of the combined company's common stock on Nasdaq; and (v) the filing of an amendment to ELSE 's Articles of Incorporation to increase the number of shares of common stock authorized for issuance to a number at least large enough to consummate the Merger, as discussed below. Certain shareholders of each company have entered Voting Agreements in connection with the Merger, as discussed below.

The Merger Agreement contains customary representations and warranties from Mobile X and ELSE. It also contains customary covenants, including (i) requiring each of the Parties to use reasonable best efforts to cause the Merger to be consummated, (ii) requiring each of the Parties to exert reasonable efforts to obtain at least \$20 million of equity financing, and (iii) for Mobile X and ELSE to carry on their respective businesses in the ordinary course of business consistent with past practice during the period between the execution of the Merger Agreement and the closing of the Merger. ELSE also agreed not to solicit, seek or initiate or knowingly take any action to facilitate or encourage a competing transaction, subject to certain exceptions. ELSE is required to seek shareholder approval of the issuance of the shares of common stock to be issued in the Merger pursuant to Nasdaq listing rules, the required amendments to its Articles of Incorporation and other matters specified in the Merger Agreement.

The Merger Agreement contains termination rights for each of Mobile X and ELSE, including, without limitation, in the event that (i) any governmental entity issues a non-appealable final order permanently enjoining the Merger, (ii) the Merger is not consummated by January 31, 2023 or (iii) the other party breaches its representations, warranties or covenants under the Merger Agreement, which breach would give rise to the failure of a closing condition and such breach is not cured within 10 days of receipt of written notice of such breach. In addition, ELSE has certain rights related to potential Superior Proposals (as defined in the Merger Agreement), including termination rights, which, if it exercises in accordance with the Merger Agreement, would entitle it to terminate the Merger Agreement by paying a fee of \$1.1 million and reimbursing certain expenses of Mobile X up to \$575,000. The Merger Agreement also provides for the sharing of certain expenses in connection with terminations where the aforementioned termination fee is not triggered.

The foregoing description of the Merger Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and is incorporated herein by reference.

The Merger Agreement contains representations, warranties, covenants and other terms, provisions and conditions that the Parties made to each other as of specific dates. The assertions embodied therein were made solely for purposes of the Merger Agreement and may be subject to important qualifications and limitations agreed to by the Parties in connection with negotiating their respective terms. Moreover, they may be subject to a contractual standard of materiality that may be different from what may be viewed as material to shareholders, or may have been used for the purpose of allocating risk between the Parties rather than establishing matters as facts. For the foregoing reasons, no person should rely on such representations, warranties, covenants or other terms, provisions or conditions as statements of factual information at the time they were made or otherwise. Unless required by applicable law, ELSE undertakes no obligation to update this information.

Electro-Sensors and Mobile X Voting Agreements and Electro-Sensors Special Meeting of Shareholders

In connection with the execution of the Merger Agreement, Electro-Sensors' directors, officers, and certain major shareholders, who collectively own a majority of Electro-Sensors' outstanding shares, have entered into agreements with Mobile X to vote their shares in favor of the Merger at a special meeting of shareholders to be held before the closing of the Merger on a date to be announced. In addition, directors, officers and certain major stockholders of Mobile X, who collectively own a majority of Mobile X's outstanding shares, have entered into similar voting agreements. No written consents have been granted nor have any votes been cast. The Voting Agreements may be terminated in connection with a termination of the Merger Agreement. Closing will follow the special meeting of shareholders of Electro-Sensors, consent of stockholders of Mobile X Global, and satisfaction of other customary and specified closing conditions, including the U.S. Securities and Exchange Commission ("SEC") having declared effective a registration statement, and The Nasdaq Stock Market having approved the listing of the common stock of the combined company.

The foregoing descriptions of the Voting Agreements do not purport to be complete and are qualified in their entirety by reference to the full text of the Voting Agreements, which are filed as Exhibit 10.1 and 10.2 to this Current Report on Form 8-K and are incorporated herein by reference.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Amendment to Bylaws

electro220769 8k.htm

On June 10, 2022, the Board of Directors Electro-Sensors, Inc. adopted a new Section 3.12 to its Bylaws, effective immediately. A copy of Section 3.12 is filed as Exhibit 3.2 to this Form 8-K.

In 2018, the State of Minnesota added Section 302A.191 to the Minnesota Business Corporation Act. Section 302A.191 explicitly authorizes a Minnesota corporation to include a provision in its articles of incorporation or bylaws making any or all of the courts in the State of Minnesota (subject to jurisdictional requirements) the exclusive forum for internal corporate claims. Section 302A.191 defines "internal corporate claim," as any

- claim that is based upon a violation of a duty under the laws of the state of Minnesota by a current or former director, officer or shareholder in such capacity;
- derivative action or proceeding brought on behalf of the corporation; or
- action asserting a claim arising under any provision of the MBCA or the corporation's articles or bylaws.

Pursuant to the statutory authority in MBCA Section 302A.191, Electro-Sensors, Inc. new bylaw establishes federal courts (where jurisdiction exists) and state courts located in Hennepin County, Minnesota as the exclusive for Electro-Sensors, Inc. internal corporate claims.

Item 8.01. Other Events

On June 13, 2022 ELSE issued a press release attached hereto as Exhibit 99.1 relating to its proposed merger transaction with Mobile X.

Item 9.01 Financial Statements and Exhibits.

Exhibit No.	Description
2.1	Agreement and Plan of Merger dated June 10, 2022 between Electro-Sensors, Inc., Mobile X Newco, Inc., a Delaware corporation and wholly owned subsidiary of Electro-Sensors, Inc, and Mobile X Global, Inc.
3.2	Section 3.12 of Electro-Sensors, Inc. Bylaws, effective June 10, 2022
10.1	Form of Voting Agreement dated June 10, 2022 among Mobile X Global, Inc, Electro-Sensors, Inc, and officers, directors and shareholders of Electro-Sensors, Inc. named on Schedule I-A
10.2	Form of Mobile X Voting Agreement dated June 10, 2022 among Mobile X Global, Inc. Electro-Sensors, Inc., and the officers, directors and Shareholders of Mobile X Global, Inc. named on Schedule I-B
99.1	Press release issued by Electro-Sensors, Inc. on June 13, 2022

All schedules to the Merger Agreement have been omitted pursuant to Item 601(b)(2) of Regulation S-K. Electro-Sensors, Inc. hereby agrees to furnish supplementally a copy of any omitted schedule to the SEC upon request.

Important Information and Where to Find It

A full description of the terms of the transaction will be provided in a proxy statement/prospectus/consent solicitation statement included in a Form S-4 Registration Statement that Electro-Sensors will file with the SEC. ELECTRO-SENSORS URGES INVESTORS, SHAREHOLDERS AND OTHER INTERESTED PERSONS TO READ, WHEN AVAILABLE, THE PROSPECTUS/PROXY STATEMENT/CONSENT SOLICITATION STATEMENT AS WELL AS OTHER DOCUMENTS FILED WITH THE SEC BECAUSE THESE DOCUMENTS WILL CONTAIN IMPORTANT INFORMATION ABOUT ELECTRO-SENSORS, MOBILE X GLOBAL, AND THE PROPOSED TRANSACTIONS. When final, the definitive proxy statement will be mailed to Electro-Sensors' shareholders as of a record date to be established for voting on the proposed transaction, and a definitive consent solicitation statement will be sent to the Mobile X stockholders. Shareholders will also be able to obtain a copy of the documents (when available), without charge, by directing a request to: Electro-Sensors, Inc., 6111 Blue Circle Drive, Minnetonka, MN 55343. These documents, once available, can also be obtained, without charge, at the SEC's website (www.sec.gov).

Participants in the Solicitation

Electro-Sensors, Inc., Mobile X Global, Inc., and their respective directors and executive officers may be considered participants in the solicitation of proxies by Electro-Sensors, Inc. in connection with the proposed transaction. Information about the directors and executive officers of Electro-Sensors, Inc. is set forth in its Annual Report on Form 10-K for the fiscal year ended December 31, 2021, and its 2022 Proxy Statement, which were filed with the SEC on March 31, 2022. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies from Electro-Sensors' shareholders in connection with the proposed Merger will be included in the definitive proxy statement/prospectus that Electro-Sensors intends to file with the SEC.

Non-Solicitation

This Form 8-K does not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which the offer, solicitation or sale would be unlawful prior to the registration or qualification under the securities laws of that jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended, or an exemption therefrom.

Forward Looking Statements

This Form 8-K includes certain forward-looking statements concerning Electro-Sensors, Mobile X Global and the proposed transactions within the meaning of the Private Securities Litigation Reform Act of 1995, including, but not limited to, statements regarding future financial performance, future growth, and the development of future products and services; the benefits of the proposed transactions, including anticipated growth and synergies; the combined company's plans, objectives and expectations and intentions; the expected timing of the proposed transactions; and future acquisitions. These statements are based on current expectations or beliefs and are subject to uncertainty and changes in circumstances. There can be no guarantee that the proposed transactions described in this press release will be completed, or that they will be completed as currently proposed, or at any particular time. Actual results may vary materially from those expressed or implied by the statements here due to changes in economic, business, competitive or regulatory factors, and other risks and uncertainties affecting the operation of Electro-Sensors as well as the business of Mobile X Global. Many of these risks, uncertainties and contingencies related to Electro-Sensors are presented in Electro-Sensors' Annual Report on Form 10-K and, from time to time, in Electro-Sensors' other filings with the SEC. These and other risks related to the business of Mobile X Global will be presented in the proxy statement/prospectus/consent solicitation statement to be filed with the SEC.

The information here should be read considering these risks and the following considerations: the ability of the merger parties to obtain definitive investment documents and close on the equity investments necessary to complete the Merger; the ability of MobileX to successfully launch its business, attract subscribers, and achieve the levels of customer service, revenues and costs that it currently expects; the ability of the combined company to successfully maintain a Nasdaq Capital Market listing; the ability of the combined company to successfully access the capital markets to finance expansion and acquisitions; the ability of the combined company to identify and acquire appropriate acquisition targets and successfully integrate these companies into its operations; the ability of the combined company to achieve synergies between its legacy sensor business and its new MobileX business; the conditions to the

closing of the Merger may not be satisfied or an event, change or other circumstance could occur that could give rise to the termination of the Merger Agreement; the Merger may involve unexpected costs, liabilities or delays, resulting in the Merger not being consummated within the expected time period; risks that the announced merger may disrupt current Electro-Sensors plans and operations or that the business or stock price of Electro-Sensors may suffer as a result of uncertainty surrounding the Merger; the outcome of any legal proceedings related to the Merger; and Electro-Sensors or Mobile X Global may be adversely affected by other economic, business, or competitive factors.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

ELECTRO-SENSORS, INC.

By: /s/ David L. Klenk

David L. Klenk, Chief Executive Officer and Chief Financial Officer

Date: June 14, 2022

electro220769_ex2-1.htm

Agreement and Plan of Merger dated June 10, 2022
between Electro-Sensors, Inc., Mobile X Newco, Inc.,
a Delaware corporation and wholly owned subsidiary
of ELSE and Mobile X Global, Inc.*

EX-2.1

electro220769_8k_FINAL

1 of 96 06/14/2022 04:19 PM

Exhibit 2.1

AGREEMENT AND PLAN OF MERGER

dated as of June 10, 2022

by and among

ELECTRO-SENSORS, INC.,

MOBILE X NEWCO, INC.,

and

MOBILE X GLOBAL, INC.,

TABLE OF CONTENTS

		<u>PAGE</u>
ARTICLE I DI	EFINED TERMS	2
1.1. 1.2. 1.3.	Specific Definitions Other Terms Interpretation	2 3 3
ARTICLE II T	HE MERGER	4
2.1. 2.2. 2.3. 2.4. 2.5. 2.6. 2.7. 2.8. 2.9. 2.10. 2.11. 2.12. 2.13. 2.14. 2.15. 2.16. 2.17. 2.18.	The Merger Effective Time Effects of the Merger Organizational Documents of the Surviving Company Management and Officers of the Surviving Company and Parent Effect on Securities Anti-Dilution Adjustments; Exchange Ratio Calculation; Freeze Date No Further Ownership Rights in Company Shares or Transfers Closing Payment of Merger Consideration Net Working Capital; Adjustments to Cash Dividend; Payment of Cash Dividend No Liability Withholding Acknowledgement of Company Stockholders Regarding the Merger Consideration Tax Consequences Parent Equity Awards Further Assurances Dissenting Shares	4 4 4 4 4 5 5 6 7 8 8 8 9 11 11 11 11 11 11
	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	14
3.1. 3.2. 3.3. 3.4. 3.5. 3.6. 3.7. 3.8. 3.9. 3.10. 3.11.	Organization; Power and Enforceability Authority Capitalization No Subsidiaries No Conflicts No Consents Financial Statements Information Provided Brokers Access to Information; Disclaimer Ownership of Parent Common Stock	14 14 15 15 15 16 16 16 17 17
4.1. 4.2. 4.3.	ADDITIONAL REPRESENTATIONS AND WARRANTIES OF COMPANY Absence of Certain Changes or Events Material Contracts Assets and Properties	18 18 18

4.4.	Intellectual Property	19
4.5.	Indebtedness	20
4.6.	Legal Proceedings; Governmental Orders	20
4.7.	Compliance With Laws	20
4.8.	Permits	20
4.9.	Affiliate Interests	20
4.10.	Taxes	21
4.11.	Books and Records	21
4,12,	Undisclosed Liabilities	21
ARTICLE V R	REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	22
5.1.	Organization, Power and Enforceability	22
5.2.	Authority	23
5.3.	Capital Structure	23
5.4.	No Conflicts	26
5.5.	Consents	26
5.6.	SEC Filings; Financial Statements; Information Provided	26
5.7.	Brokers	29
5.8.	Absence of Certain Changes or Events	29
5.9.	Material Contracts	29
5.10.	Taxes	30
5.11.	Assets and Properties	32
5.12.	Intellectual Property	32
5.13.	Legal Proceedings	33
5.14.	Indebtedness	33
5.15.	Compliance With Laws	33
5.16.	Merger Sub	33
5.17.	Fairness Opinion	33
5.18.	Employee Benefit Matters	33
5.19.	Permits	36
5.20.		36
5.21.	Insurance	37
5.22.	Affiliate Interests	37
5.23.	Access to Information; Disclaimer	37
ARTICLE VI	CONDUCT OF BUSINESS	38
6.1.	Pre-Closing Covenants of Company	38
6.2.	Pre-Closing Covenants of Parent	38
6.3.	Confidentiality	40
6.4.	No Control	40
ARTICLE VII	ADDITIONAL COVENANTS	40
7.1.	No Solicitation	40
7.1.	Access to Information; Cooperation; Financial Statements	45
7.2.	Proxy Statement/Consent/Prospectus; Registration Statement; Written Consent	46
7.3. 7.4.	Nasdaq Listing	48
7.5.	Shareholder Approval	48
7.6.	Fulfillment of Conditions	50
,		50

4 of 96

electro220769_ex2-1.htm

7.7. 7.8. 7.9. 7.10. 7.11. 7.12. 7.13.	Public Disclosure Notification of Breaches Succession of Officers and Directors State Takeover Laws Section 16 Matters Cash Dividend Termination of Certain Agreements and Rights	50 51 51 51 51 51
7.14. 7.15. 7.16. 7.17. 7.18. 7.19.	Certain Tax Matters PIPE Investment D&O Insurance and Indemnification Parent Transaction Expenses Statement; Indebtedness Reverse Split Litigation Continued Employment	52 52 54 55 55 56 56
ARTICLE VIII	CONDITIONS TO MERGER	57
8.2.	Conditions to Each Party's Obligation to Effect the Merger Conditions to the Obligations of the Parent and the Merger Sub Conditions to the Obligations of Company	57 58 59
ARTICLE IX T	ERMINATION AND AMENDMENT	60
9.2. 9.3. 9.4. 9.5.	Termination Effect of Termination Fees and Expenses Amendment Extension; Waiver Procedure for Termination, Amendment, Extension or Waiver	60 62 62 64 64
ARTICLE X M	ISCELLANEOUS	65
10.11. 10.12. 10.13.	Non-Survival of Representations and Warranties Notices Entire Agreement No Third-Party Beneficiaries Assignment Severability Counterparts and Signatures Governing Law Remedies; Specific Performance Submission to Jurisdiction WAIVER OF JURY TRIAL Disclosure Schedule No Recourse Legal Representation	65 65 66 66 66 67 67 67 67 68 68 68

EXHIBITS

Exhibit A-1 Form of Parent Voting Agreement

Exhibit A-2 Form of Company Voting Agreement

Exhibit B Certain Defined Terms

Exhibit C Terms of Parent Articles of Incorporation

SCHEDULES

Schedule 1 Parties to Parent Voting Agreement

COMPANY DISCLOSURE SCHEDULE

Section 3.1 Organization; Power and Enforceability

Section 3.3(a) Capitalization

Section 3.3(c) Options, Warrants, Convertible Securities or other Rights

Section 3.4 No Conflicts

Section 3.7(a) Financial Statements

Section 3.8(d) Broker

Section 4.1 Absence of Certain Changes

Section 4.2 Material Contracts

Section 4.2(b) Assets and Properties

Section 4.4 Indebtedness

Section 4.6(a) Legal Proceedings; Governmental Orders

Section 4.8 Permits

Section 4.9 Affiliate Interests

Section 4.10 Taxes

Section 4.12 Undisclosed Liabilities

Article VI Pre-Closing; Ordinary Course

Section 7.12 Termination of Certain Agreements and Rights

Section K Knowledge Group

Section P Permitted Indebtedness

PARENT DISCLOSURE SCHEDULE

Section 5.5 Consents

Section 5.7 Broker

Section 5.9(a) Material Contracts

Section 5.11(b) Parent Owned Real Property

6 of 96

Section 5.12 Intellectual Property Section 5.14 Indebtedness Section 5.18(a) Parent Plans Section 5.18(e) Employment; Noncontravention Section 5.19 Permits Insurance Policies Section 5.21 Section 5.22 Affiliate Interests Section 6.2 Exceptions to Pre-Closing Covenants of Parent Section 7.9 Appointment of Officers Section 7.9(b) Resignation of Directors Section K Knowledge Group

electro220769_ex2-1.htm

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of June 10, 2022, is entered into by and among ELECTRO-SENSORS, INC., a Minnesota corporation (the "Parent"), MOBILE X NEWCO, INC., a Delaware corporation and a wholly owned Subsidiary of Parent (the "Merger Sub"), and MOBILE X GLOBAL, INC., a Delaware corporation (the "Company," and together with Parent and the Merger Sub, the "Parties" and each a "Party").

RECITALS

WHEREAS, upon the terms and subject to the conditions set forth in this Agreement and in accordance with the Delaware General Corporation Law, as amended (the "<u>DGCL</u>"), and the Minnesota Business Corporation Act, as amended (the "<u>MBCA</u>"), the Parties intend that Merger Sub shall be merged with and into the Company (the "<u>Merger</u>"), with the Company surviving the Merger as a wholly owned Subsidiary of the Parent, in a transaction intended to qualify as a tax free reorganization under Section 368(a)(1)(A) of the Code by reason of Section 368(a)(2)(E) of the Code;

WHEREAS, in connection with and in furtherance of the Merger, (a) the Parent shall effect the Reverse Split of the Parent Common Stock and (b) the shareholders of the Parent as of the close of the Business Day immediately preceding the Effective Time (or such other time prior to the Effective Time as the Parent and Company may otherwise determine) shall (i) receive from Parent, on a post-Reverse Split basis, a pro rata share of a cash dividend totaling \$18.0 million (the "Provisional Cash Dividend"), subject to potential adjustment in accordance with Section 2.11 of this Agreement (as so adjusted, the "Cash Dividend") and (ii) retain a continuing ownership interest in the Parent equal to approximately 10.66% of the Parent Diluted Shares as of immediately after the Effective Time (subject to future dilution);

WHEREAS, the Parties intend to enter into other transactions and agreements contemplated by this Agreement, including the Parent Voting Agreement, the Parent Stock Issuance, the Reverse Split, the Merger and the effectuation of the Reincorporation and Charter Amendment and the PIPE Investment (collectively, the "<u>Transactions</u>");

WHEREAS, the Board of Directors of the Company (the "Company Board") has unanimously (i) determined that this Agreement and the Transactions are fair to and in the best interests of the Company and the Company Stockholders, (ii) approved, adopted and declared advisable this Agreement and the Transactions, (iii) directed that the adoption of this Agreement be submitted to a vote at a meeting of the Company Stockholders or action by written consent in lieu of a meeting and (iv) recommended the adoption of this Agreement and approval of the Transactions by the Company Stockholders;

WHEREAS, the Board of Directors of the Parent (the "Parent Board") has (i) determined that this Agreement and the Transactions (including the issuance by the Parent of its voting common stock, par value \$0.10 per share, in connection with the Merger, such issuance, the "Parent Stock Issuance," and such stock, the "Parent Common Stock") are fair to and in the best interests of the Parent and its shareholders, (ii) approved, adopted and declared advisable this Agreement and the Transactions, (iii) directed that the approval of the Transactions be submitted

to a vote at a meeting of the Parent's shareholders and (iv) recommended the adoption of this Agreement and the approval of the Transactions by the Parent's shareholders at the Parent Shareholders Meeting as required by the MBCA, the DGCL, the rules of the Nasdaq Capital Market ("Nasdaq") and the Parent's Organizational Documents;

WHEREAS, the Parent, in its capacity as the sole stockholder of the Merger Sub, has (i) approved the execution, delivery and performance of this Agreement and the consummation of the Transactions, and (ii) adopted this Agreement;

WHEREAS, the Parties desire to make certain representations, warranties, covenants and agreements specified herein in connection with this Agreement;

WHEREAS, on or prior to the date hereof, the Company has obtained and delivered to Parent a commitment letter dated May 17, 2022 from an institutional investor (the "Commitment Letter") for a private placement of shares of Parent preferred stock (the "PIPE Investment") and each Party intends to cooperate to obtain binding subscription agreements for the PIPE Investment (each, a "Subscription Agreement"), from the institutional investor or such other investors as the Parties may mutually agree upon, such private placements to be consummated concurrently with or prior to the consummation of the Transactions and resulting in an equity financing to the Parent with minimum gross proceeds of \$20.0 million in the aggregate (the "Minimum PIPE Investment Amount");

WHEREAS, as a condition and inducement to the Company's willingness to enter into this Agreement, concurrently with the execution and delivery of this Agreement, the members of the Parent Board and certain shareholders of the Parent identified on Schedule 1 are entering into the voting agreement with the Company (the "Parent Voting Agreement"), in substantially the form attached hereto as Exhibit A-1; and

WHEREAS, as a condition and inducement to the Parent's willingness to enter into this Agreement, concurrently with the execution and delivery of this Agreement, the certain stockholders of the Company sufficient to constitute the Company Stockholder Approval are entering into the voting agreement with the Parent (the "Company Voting Agreement"), in substantially the form attached hereto as Exhibit A-2.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the representations, warranties, covenants and agreements set forth herein, and for other good and valuable consideration the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINED TERMS

1.1. Specific Definitions. As used in this Agreement, the terms identified on Exhibit B shall have the meanings set forth or referred to in Exhibit B.

- 1.2. Other Terms. Other terms may be defined elsewhere in the text of this Agreement and, unless otherwise indicated or the context requires otherwise, shall have such meaning throughout this Agreement.
 - 1.3. Interpretation.
- (a) The words "hereof," "herein," "hereby," "hereto" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement.
- (b) Whenever the words "includes" or "including" (or any variation thereof) are used in this Agreement, they shall be deemed to be followed by the words "without limitation."
 - (c) References herein to "days," unless indicated otherwise, are to consecutive calendar days.
- (d) References to specific Articles and Sections are to the Articles and Sections of this Agreement, unless specifically stated otherwise.
- (e) The terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa, and words denoting any gender shall include all genders.
 - (f) All references to "dollars" or "\$" shall mean "U.S. dollars," unless specifically stated otherwise.
- (g) All references herein to a particular "Company Disclosure Schedule" shall mean such schedule as it is included in the Company Disclosure Schedule attached hereto. All references herein to a particular "Parent Disclosure Schedule" shall mean such schedule as it is included in the Parent Disclosure Schedule attached hereto.
 - (h) All references herein to a particular "Exhibit" shall mean such exhibit as it is attached hereto.
- (i) References herein to documents (other than the Company Disclosure Schedule, Parent Disclosure Schedule, and this Agreement) having been "provided," "made available" or "delivered" means that such information was either (i) provided directly to the Parent or the Company, as applicable, by the other Party, (ii) included in the virtual data rooms established by the Parent and the Company created for the purposes of providing information to the other Party in connection with this Agreement at least two (2) Business Days prior to the execution and delivery of this Agreement or (iii) solely with respect to information made available by the Parent, filed with and publicly available on the SEC's EDGAR system in the Parent SEC Reports filed prior to the date that is two (2) Business Days prior to the date of this Agreement.
 - (j) All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP.

(k) No summary of this Agreement prepared by any Party shall affect the meaning or interpretation of this Agreement.

ARTICLE II THE MERGER

- 2.1. The Merger. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with the applicable provisions of the DGCL and the MBCA, at the Effective Time, (a) the Merger Sub shall be merged with and into the Company, whereupon the separate corporate existence of the Merger Sub shall cease, and (b) the Company shall continue as the surviving company in the Merger (the "Surviving Company") as a wholly owned Subsidiary of Parent.
- 2.2. Effective Time. On the Closing Date, the Merger Sub and the Company shall execute and file with the Secretary of State of the State of Delaware a certificate of merger (the "Certificate of Merger"), executed in accordance with, and containing such information as is required by, the applicable provisions of the DGCL in order to effect the Merger. The Merger shall become effective on such date and at such time as the Certificate of Merger is filed with the Secretary of State of the State of Delaware, or at such other time as is agreed upon in writing by Parent and the Company and specified in the Certificate of Merger in accordance with the relevant provisions of the DGCL (such time is hereinafter referred to as the "Effective Time").
- 2.3. Effects of the Merger. The Merger will have the effects set forth in this Agreement and in the applicable provisions of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all of the assets, properties, rights, privileges, immunities, powers and franchises of the Company and the Merger Sub shall vest in the Surviving Company, and all debts, liabilities, duties and obligations of the Company and the Merger Sub shall become the debts, liabilities, duties and obligations of the Surviving Company, all as provided under the DGCL.
 - 2.4. Organizational Documents of the Surviving Company.
- (a) At the Effective Time, the Certificate of Incorporation of the Company as in effect immediately prior to the Effective Time shall be the Certificate of Incorporation of the Surviving Company until thereafter amended in accordance with the provisions thereof and applicable Law, except that the name of the Surviving Company shall be "Mobile X Services, Inc."
- (b) At the Effective Time, the Bylaws of the Company as in effect immediately prior to the Effective Time shall be the Bylaws of the Surviving Company until thereafter amended in accordance with the provisions thereof and applicable Law.
 - 2.5. Management and Officers of the Surviving Company and Parent.
- (a) At the Effective Time, the directors of the Company immediately prior to the Effective Time shall be the initial directors of the Surviving Company and shall hold office

until their respective successors are duly appointed or until their earlier death, resignation or removal.

- (b) At the Effective Time, the officers of the Company immediately prior to the Effective Time shall be the initial officers of the Surviving Company and shall hold office until their respective successors are duly appointed or until their earlier death, resignation or removal.
- (c) At the Effective Time, the persons whose names are provided to the Parent in writing at least five (5) Business Days prior to the Closing shall be the officers of the Parent and shall hold office until their respective successors are duly appointed or until their earlier death, resignation or removal.
- (d) At the Effective Time, the persons designated in accordance with Section 7.9(b)(ii) of this Agreement shall be the directors of the Parent and shall hold office until their respective successors are duly elected and qualified or until their earlier death, resignation or removal.
- 2.6. Effect on Securities. On the terms and subject to the conditions set forth herein, as of the Effective Time, by virtue of the Merger and without any action on the part of the Parent, the Merger Sub, the Company or any of their respective equity holders, the following will occur:
- (a) <u>Conversion of Merger Sub's Shares</u>. Each share of common stock of the Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted automatically into and become one validly issued, fully paid and non-assessable share of common stock of the Surviving Company. Each stock certificate of Merger Sub evidencing ownership of any such shares shall, as of the Effective Time, evidence ownership of such shares of common stock of the Surviving Company.
- (b) <u>Cancellation of Excluded Company Shares</u>. Each Company Share held as treasury stock or issued and outstanding immediately prior to the Effective Time that is owned directly or indirectly by the Company, Parent or Merger Sub, if any, shall no longer be outstanding, shall be cancelled automatically and shall cease to exist (the "<u>Excluded Company Shares</u>"), and no consideration shall be paid or delivered in exchange therefor.
- (c) <u>Conversion of Company Shares</u>. Subject to <u>Section 2.6(d)</u>, each Company Share (other than the Excluded Company Shares and any Dissenting Company Shares) outstanding immediately prior to the Effective Time shall be automatically converted solely into the right to receive a number of shares of Parent Common Stock equal to the Exchange Ratio (the "<u>Merger Consideration</u>").
- (d) <u>Fractional Shares</u>. No certificates or scrip representing fractional shares of Parent Common Stock shall be issued upon the conversion of Company Shares pursuant to <u>Section 2.6(c)</u> and such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a holder of shares of Parent Common Stock. Notwithstanding any other provision of this Agreement, each holder of Company Shares converted pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Parent Common Stock (after taking into account all Company Shares exchanged by such holder) shall in lieu thereof,

upon cancellation and conversion of such holder's Company Shares, receive in cash (rounded down to the nearest whole cent), without interest, an amount equal to such fractional amount multiplied by the last reported sale price of Parent Common Stock on Nasdaq on the last official trading day prior to the date of the Effective Time.

- (e) <u>Parent Stock Options</u>. All Parent Stock Options outstanding immediately prior to the Effective Time under the Parent Stock Plans shall be treated in accordance with <u>Section 2.16</u>.
- Company Stock Options. As of the Effective Time, each option to acquire Company Shares (each, a "Company Stock Option") that is outstanding under any Company Stock Plan immediately prior to the Effective Time, whether or not then vested or exercisable, shall be, by virtue of the Merger and without any action on the part of the holder thereof, or any other Person, be assumed by Parent and shall be converted into a Parent Stock Option in accordance with this Section 2.6(f). Each such Parent Stock Option as so assumed and converted shall continue to have, and shall be subject to, the same terms and conditions as applied to the Company Stock Option immediately prior to the Effective Time. As of the Effective Time, each such Parent Stock Option as so assumed and converted shall be an option to acquire that number of whole shares of Parent Common Stock (rounded down to the nearest whole share) equal to the product of: (i) the number of shares of Company Common Stock subject to such Company Stock Option; and (ii) the Exchange Ratio, at an exercise price per share of Parent Common Stock (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (A) the exercise price per share of Company Common Stock of such Company Stock Option by (B) the Exchange Ratio; provided, that the exercise price and the number of shares of Parent Common Stock subject to the Parent Stock Option shall be determined in a manner consistent with the requirements of Section 409A of the Code, and, in the case of Company Stock Options that are intended to qualify as incentive stock options within the meaning of Section 422 of the Code, consistent with the requirements of Section 424(a) of the Code. At or prior to the Effective Time, the Company, the Company Board, and the compensation committee of such board, as applicable, shall adopt any resolutions and take any actions (including obtaining any employee consents) that may be necessary to effectuate the provisions of this Section 2.6(f). All Company Stock Options converted pursuant to this Section 2.6(f) will be deemed granted under a new Parent Stock Option Plan that will be adopted by Parent and approved by Parent shareholders in accordance with applicable rules of Nasdaq at the Parent Shareholders Meeting. At or prior to the Effective Time, Parent shall reserve for future issuance a number of shares of Parent Common Stock at least equal to the number of shares of Parent Common Stock that will be subject to Parent Stock Options as a result of the actions contemplated by this Section 2.6(f). As soon as practicable after the Effective Time, if and to the extent necessary to cause a sufficient number of shares of Parent Common Stock to be registered and issuable with respect to the Parent Stock Options, Parent shall prepare and file with the SEC a registration statement on Form S-8 (or any successor or other appropriate form) with respect to the shares of Parent Common Stock subject to the Parent Stock Options.
 - 2.7. Anti-Dilution Adjustments; Exchange Ratio Calculation; Freeze Date.
- (a) If, between the date hereof and the Effective Time, the issued and outstanding Company Shares shall be changed into a different number of Company Shares or a

different class of Equity Interests by reason of any reorganization, reclassification, recapitalization, combination, stock split, reverse stock split, dividend or exchange, or any similar event shall occur, the Exchange Ratio shall be equitably adjusted, without duplication, to proportionally reflect such change and provide to the Company Stockholders the same economic effect as contemplated by this Agreement prior to such event; *provided*, *however*, that nothing in this Section 2.7(a) shall be construed to permit the Company to take any action with respect to its securities that is prohibited by the terms of this Agreement.

- (b) If, between the date hereof and the Effective Time, the issued and outstanding shares of Parent capital stock shall be changed into a different number of shares of Parent capital stock or a different class of Equity Interests by reason of any reorganization, reclassification, recapitalization, combination, stock split, reverse stock split (including the Reverse Split), dividend or exchange, or any similar event shall occur, the Merger Consideration shall be equitably adjusted, without duplication, to proportionally reflect such change and provide to the Company Stockholders the same economic effect as contemplated by this Agreement prior to such event; *provided*, *however*, that nothing in this Section 2.7(b) shall be construed to permit Parent to take any action with respect to its securities that is prohibited by the terms of this Agreement.
- (c) The Parent and the Company shall reasonably cooperate to calculate the Exchange Ratio in accordance with this Agreement, including by giving effect to the Reverse Split. Parent shall provide the Company with a statement setting forth in reasonable detail its good faith estimate of the Exchange Ratio at least Seven (7) Business Days prior to the expected Closing Date, for the Company's review and comment, which estimate shall then be updated as of at least Three (3) Business Days before the Closing Date, which shall again be subject to the Company's review and comment. Such updated estimate shall, after taking into account the Company's review and comment, be used as the Exchange Ratio at the Closing and neither the Parent nor the Company shall take or permit any action that would reasonably be expected to change the Exchange Ratio from the date that is Three (3) Business Days before the Closing Date until the Closing (other than actions, if any, expressly contemplated by Section 2.16).
- (d) Except as otherwise provided in this Agreement, to facilitate the calculation of the Exchange Ratio, each Party will use its respective reasonable efforts to take all appropriate action to freeze option exercises or other stock issuance events at least Four (4) Business Days prior to the Closing Date, except as may be provided pursuant to other arrangements that are made pursuant to <u>Section 2.16(c)</u>.
- 2.8. No Further Ownership Rights in Company Shares or Transfers. The Merger Consideration, when issued in accordance with the terms of this Agreement, shall be deemed to have been delivered, issued and paid in full satisfaction of all Company Stockholders' rights relating to the Company Shares. From and after the Effective Time, (a) all holders of Company Shares shall cease to have any rights as equity holders of the Company other than the right to receive the Merger Consideration issuable to such holders under this Agreement, (b) the Company Shares transfer books of the Surviving Company shall be closed, and there shall be no further registration of transfers on the transfer books of the Surviving Company of the Equity Interests in the Company that were outstanding immediately prior to the Effective Time.

- 2.9. Closing. The closing of the Merger (the "Closing") shall take place at, or be directed from, the offices of Ballard Spahr LLP in Minneapolis, Minnesota, at 10:00 a.m., Minneapolis, Minnesota time, as soon as practicable (and, in any event, within three (3) Business Days of the satisfaction or (to the extent permitted by applicable Law) waiver of all of the conditions precedent set forth in Article III (other than those conditions that by their nature or terms are to be satisfied at the Closing, but subject to the satisfaction or (to the extent permitted by applicable Law) waiver of those conditions)), or at such other place or time or on such other date as Parent and the Company may mutually agree in writing. The date on which the Closing occurs is referred to in this Agreement as the "Closing Date." The Closing shall be deemed to be effective as of the Effective Time, and all documents delivered and actions taken at the Closing shall be deemed to have been delivered or taken simultaneously at such time.
- 2.10. Payment of Merger Consideration. Immediately after the Effective Time, the Parent shall pay and deliver the Merger Consideration as follows:
- (a) The Parent shall direct its transfer agent to issue, in the name of each Company Stockholder, the number of shares of Parent Common Stock to be issued to each such Company Stockholder pursuant to the terms of this Agreement (as such number of shares are reflected on a certificate signed by a duly authorized officer of the Company and delivered by the Company to the Parent at least three (3) Business Days before the Closing Date, which will set forth and be certified therein as true and accurate as of the Effective Time: (i) each Company Stockholder as of the Effective Time; (ii) the number of Company Shares held by each such Company Stockholder; and (iii) the number of shares of Parent Common Stock to be issued on conversion of the Company Shares held by each such Company Stockholder at Closing in accordance with the Company's Organizational Documents and this Agreement (such certificate, the "Allocation Certificate"));
- (b) The Parent shall deliver to its transfer agent for delivery to the Company Stockholders the aggregate amount of cash sufficient to pay for fractional shares of Parent Common Stock calculated pursuant to Section 2.6(d); and
- (c) Promptly after the Effective Time, Parent and the Surviving Corporation shall cause to be mailed, or if appropriate transmitted electronically, by the Parent's transfer agent, acting as exchange agent, to each Person who was, at the Effective Time, a holder of record of Company Shares entitled to receive the Merger Consideration pursuant to this Agreement a customary and reasonable form of letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to any certificates evidencing such Company Shares (the "Certificates") shall pass, only upon proper delivery of the Certificates to the exchange agent (or effective affidavits of loss in lieu of such Certificates as set forth in below) and instructions for use in effecting the surrender of the Certificates and of any uncertificated Company Shares entitled to receive Merger Consideration ("Uncertificated Shares") pursuant to such letter of transmittal). Upon surrender to the exchange agent of a Certificate, or upon receipt of by the exchange agent of such evidence, if any, of transfer as the exchange agent may reasonably request in the case of a book-entry transfer of Uncertificated Shares, together with such letter of transmittal, duly completed and validly executed in accordance with the instructions thereto, and such other documents as may be reasonably required pursuant to such instructions, the holder of such Certificate (or effective affidavits of loss in lieu of such

Certificates as set forth below) and the holder of such Uncertificated Shares shall be entitled to promptly receive in exchange therefor the Merger Consideration for each Company Share formerly evidenced by such Certificate or Uncertificated Shares, and such Certificate and transferred Uncertificated Shares so surrendered shall then be cancelled. The exchange agent will accept such Certificates and Uncertificated Shares upon compliance with such reasonable and customary terms and conditions as the exchange agent may impose to cause an orderly exchange thereof in accordance with normal exchange practices. Until so surrendered, outstanding Certificates and Uncertificated Shares will be deemed from and after the Effective Time to evidence only the right to receive the applicable Merger Consideration, without interest thereon, payable in respect thereof pursuant to this Agreement. If payment of the Merger Consideration is to be made to a person other than the Person in whose name the surrendered Certificate or Uncertificated Share is registered on the stock transfer books of the Company, it shall be a condition of payment that the Certificate or Uncertificated Share so surrendered shall be endorsed properly or otherwise be in proper form for transfer and that the person requesting such payment shall have paid all transfer and other similar Taxes required by reason of the payment of the Merger Consideration to a person other than the registered holder of the Certificate or Uncertificated Share surrendered or shall have established to the reasonable satisfaction of the Surviving Corporation that such Taxes either have been paid or are not applicable. The Merger Consideration paid in accordance with this Agreement upon the surrender for exchange of Certificates and Uncertificated Shares shall be deemed to have been paid in full satisfaction of all rights pertaining to the Company Shares theretofore represented by such Certificates and Uncertificated Shares. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by such person of a bond in such reasonable and customary amount as the Surviving Corporation may require as indemnity against claims that may be made against it with respect to such Certificate, the exchange agent will issue in exchange for such lost, stolen or destroyed Certificate the Merger Consideration payable in respect thereof, pursuant to this Agreement.

2.11. Net Working Capital; Adjustments to Cash Dividend; Payment of Cash Dividend.

(a) The Parent and the Company shall reasonably cooperate to calculate the adjustments required to determine the amount of the Cash Dividend in accordance with this Agreement. Parent shall use its reasonable efforts to procure, at least eight (8) Business Days prior to the expected Closing Date, invoices from all service providers to Parent related to this Agreement and the Transactions for all services expected to be performed through the Closing Date, as well as invoices needed to determine the amount of Parent Transaction Expenses that are unpaid as of the Closing Date. Parent shall provide the Company with a statement setting forth in reasonable detail its good faith estimates of the Parent's Net Working Capital and Net Working Capital (Adjusted), as of the Closing Date and of the aggregate amount of Parent Transaction Expenses and Indebtedness as required by Section 7.17 as of the Closing, at least Seven (7) Business Days prior to the expected Closing Date, together with all supporting documentation as the Company may reasonably request, for the Company's review and comment, which estimates shall then be updated as of at least Four (4) Business Days before the Closing Date, which shall again be subject to the Company's review and comment. Such updated estimates shall, after taking into account the Company's review and comment, absent manifest error or fraud, be used as the definitive, final and binding Net Working Capital, Net Working

Capital (Adjusted) and the Parent Transaction Expenses and Indebtedness to calculate the amount of the Cash Dividend at the Closing, and thereafter neither the Parent nor the Company shall take or permit any action that would reasonably be expected to change the Cash Dividend from the date that is Three (3) Business Days before the Closing Date until the Closing; *provided, however*, if the Company reasonably determines that the Cash Dividend should be more than \$250,000 less than the amount determined by the Parent, the Company shall have the right by written notice to Parent to delay the Closing by up to Ten (10) Business Days in which to resolve the disagreement and in the absence of a resolution in such time period, the Company shall also have the right to require the Parent to negotiate in good faith an amendment to this Agreement to provide for a further delay in the Closing and the bifurcation of the Cash Dividend into an undisputed and a disputed portion, the undisputed portion of which would be determined and promptly paid as originally contemplated with respect to the Cash Dividend, and the latter of which would be determined and paid only after the disputes have been resolved after the Closing.

- (b) Upon such determination of the Net Working Capital, Net Working Capital (Adjusted), and the Parent Transaction Expenses and Indebtedness, the Provisional Cash Dividend will *first* be decreased by an amount equal to Fifty Percent (50%) of the aggregate Parent Transaction Expenses (whether previously paid or still payable), and *second*, such resulting amount will be either be (x) increased by the amount, if any, that the Net Working Capital (Adjusted) of the Parent at Closing is greater than the Target Working Capital, or (y) decreased by the amount, if any, that the Net Working Capital (Adjusted) of the Parent at Closing is less than the Target Working Capital or (z) left unchanged, if the Net Working Capital (Adjusted) of the Parent is equal to the Target Working Capital; <u>provided, however</u>, that the amount of the Cash Dividend that would result from taking into account only the foregoing adjustments shall, notwithstanding the actual amount of such adjustments, not be less than \$18.0 million nor more than \$18.5 million; and finally, *third*, the resulting amount after the foregoing adjustments shall be decreased by One Hundred Percent (100%) of any Indebtedness of Parent incurred after the date hereof or Transaction Bonus Amounts (whether incurred before or after the date hereof), and such final resulting amount after all of the foregoing adjustments shall be the Cash Dividend to be paid pursuant to Section 2.11(c).
- (c) Subject to Nasdaq rules, as soon as practicable after the amount of the Cash Dividend has been determined in accordance with Section 2.10(b), Parent shall set a record date for such dividend to be the Business Day immediately prior to the expected Closing Date (or such other date as the Parties may otherwise agree) (the "Dividend Record Date") and a payment date for such dividend to be the earliest practicable date after the Closing Date. The shareholders of the Parent of record as of the close of business on the Dividend Record Date shall receive, on a pro rata basis per share of Parent Common Stock then held by them, their share of the Cash Dividend. The Parent shall promptly deliver to its transfer agent (acting as a paying agent), and direct its transfer agent to pay (including through the facilities of the Depository Trust Company), in accordance with the rules of Nasdaq, in the name of each such holder of Parent Common Stock, a check (or other appropriate form of payment) in the amount of such shareholder's pro rata share of the Cash Dividend on the payment date thereof (based on the number of shares of Parent Common Stock that are outstanding as of the Dividend Record Date, such number of shares to be certified by the Parent's transfer agent, a copy of which shall be provided to the Company prior to the Closing, together with such other information reasonably requested by the

Company to confirm the due payment of the Cash Dividend to the appropriate shareholders of Parent).

- 2.12. No Liability. None of the Parent, the Merger Sub or the Company shall be liable to any Person in respect of any portion of the Merger Consideration or Cash Dividend delivered to a public official in compliance with any applicable abandoned property, escheat or similar Law.
- 2.13. Withholding. Notwithstanding any other provision of this Agreement to the contrary, each of the Parent and the Surviving Company shall be entitled to deduct and withhold from any Parent Stock Options and from any consideration otherwise payable pursuant to this Agreement such amounts, if any, as are required to be deducted and withheld under applicable Law; provided, however, prior to the Closing, to the extent Parent believes withholding is required with respect to the Company or any shareholders thereof, the Parent shall provide the Company with reasonable notice of its intention to deduct or withhold and Parent will reasonably cooperate with the Company in reducing or eliminating such deduction or withholding and the Company will reasonably cooperate with Parent in implementing any such deduction or withholding that is required. Any amounts so deducted and withheld and timely paid over to or deposited with the appropriate taxing authority shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding are made.
- 2.14. Acknowledgement of Company Stockholders Regarding the Merger Consideration. The Company acknowledges and agrees that the Parent is entitled to rely conclusively on any Allocation Certificate delivered by the Company hereunder to the Parent, and upon delivery of the Merger Consideration in accordance with this Agreement and the Allocation Certificate, neither the Parent nor the Merger Sub shall have any further responsibility or Liability hereunder with respect to the payment of the Merger Consideration.
- 2.15. Tax Consequences. For federal income tax purposes, the Merger is intended to constitute a reorganization within the meaning of Section 368(a) of the Code in a transaction intended to qualify as a statutory merger under Section 368(a)(1)(A) of the Code by reason of Section 368(a) (2)(E) of the Code. The Parties to this Agreement hereby adopt this Agreement as a "plan of reorganization" within the meaning of Treasury Regulations Sections 1.368-2(g) and 1.368-3(a).
 - 2.16. Parent Equity Awards.
- (a) The Parent has granted equity and equity-based awards in the form of stock options ("Parent Stock Options") under (i) the Parent's 1997 Stock Option Plan ("Parent 1997 Plan"), and (ii) the Parent's 2013 Equity Incentive Plan ("Parent 2013 Plan" and, together with the Parent 1997 Plan, the "Parent Stock Plans"). Between the date of this Agreement and the Effective Time, if requested in writing by the Company, the Parent shall take all necessary action (which action shall be effective as of the Effective Time) to terminate the Parent Stock Plans (without the creation of additional Liability to the Parent, or the Company or the Surviving Company or impairing the ability of any option holder to exercise any outstanding options). As soon as reasonably practicable following the date of this Agreement and in any event prior to the Effective Time, the Parent Board (or, if appropriate, any committee administering the Parent

Stock Plans) shall adopt such resolutions and take all actions that are reasonably necessary for the treatment of the Parent Stock Options pursuant to this Section 2.16. For purposes of any Parent Stock Plan, no Parent Stock Options shall be assumed or substituted and all such options shall be treated as "non-assumed incentive awards" as otherwise defined in such Parent Stock Plans. The Parent shall take all actions that are necessary to ensure that from and after the Effective Time neither Parent nor the Surviving Corporation will be required to deliver shares of Parent Common Stock or other capital stock of the Parent to any Person pursuant to or in settlement of Parent Stock Options other than as set forth in this Section 2.16.

- (b) As of or prior to the Dividend Record Date, each outstanding Parent Stock Option granted under the Parent Stock Plans to each holder thereof shall become vested pursuant to the terms of the Parent Stock Plans or actions taken by the Board in order to effectuate such vesting. At the Effective Time, each outstanding Parent Stock Option shall be cancelled and automatically converted into that number of whole shares of Parent Common Stock (and cash in lieu of any fractional shares) equal in value to the excess ("Excess") of the Fair Market Value as of the Business Day immediately prior to the Closing Date of the number of outstanding shares under the Parent Stock Options over the sum of (i) the aggregate exercise price plus (ii) the amount of any Taxes, if any, required to be withheld with respect to such Excess in accordance with Section 2.13. For the avoidance of doubt, in order for holders of Parent Stock Options to be entitled to their pro rata share of the Cash Dividend, they must have duly exercised their stock options a sufficient time before the Dividend Record Date so that they are holders of outstanding shares of Parent Common Stock as of the Dividend Record Date received upon exercise of their stock options, or the Parent must have made such other arrangements (subject to the Company's approval, not to be unreasonably withheld) to ensure that the holders of Parent Stock Options will receive their pro rata share of the Cash Dividend upon exercise of their stock options on or prior to the Effective Time, which may include an exercise of stock options conditional on the Closing or an exercise of stock options on the Dividend Record Date, in each case if the procedures of the Parent and its transfer agent and registrar so allow. Any Parent Stock Option in which the aggregate exercise price exceeds the Fair Market Value of the outstanding shares under the Parent Stock Option as of the Effective Time shall be cancelled without any action on the part of any holder thereof and without any payment or consider
- Unless otherwise agreed by Parent and Company, prior to seeking shareholder approval of the Merger, Parent will engage an independent fiduciary reasonably acceptable to Company (the "ESOP Fiduciary") for the Parent's existing Employee Stock Ownership Plan ("ESOP") to take any and all reasonably appropriate actions with respect to the ESOP in connection with the transactions contemplated by this Agreement in order to satisfy its fiduciary duty to the ESOP in connection with the Parent Shareholder Meeting, the Merger and this Agreement in accordance with this Agreement and the ESOP governing documents. Parent will cooperate in promptly providing all material information requested by the ESOP Fiduciary in connection with its services as described above. Parent will, prior to Closing, pay all reasonable fees and costs incurred by, or reasonably estimated to be due from, Parent for or in connection with the services of the ESOP Fiduciary and any advisors thereto, with all such amounts to be Parent Transaction Expenses and Shared Expenses. Parent will cause to be distributed to each ESOP participant or beneficiary the same notices and information statement

relating to the Merger as are distributed to each non-ESOP holder of Parent Common Stock and will hold the vote of the ESOP participants in strict confidence. At the Company's written request, the Parent and the Company will in good faith cooperate to announce and effectuate a termination of the ESOP effective at or after the Closing. For the avoidance of doubt, this <u>Section 2.16(c)</u> is for the sole benefit of the Parties and neither the ESOP Fiduciary nor any participant in the ESOP shall have any rights under this <u>Section 2.16(c)</u>.

2.17. Further Assurances. If at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Surviving Company, the Parent and their respective shareholders following the Merger with full right, title and possession to all assets, property, rights, privileges, powers and franchises, the applicable directors, officers, members and managers of the Company, the Parent and Merger Sub (or their successors, assigns and designees) are fully authorized in the name of their respective corporations or otherwise to take, and will take, all such lawful and necessary action so long as such action is not inconsistent with this Agreement.

2.18. Dissenting Shares.

- (a) Notwithstanding anything to the contrary set forth in this Agreement, all Company Shares that are issued and outstanding immediately prior to the Effective Time and held by the stockholders of the Company who shall have neither voted in favor of the Merger nor consented thereto in writing and who are entitled to demand and shall have properly and validly perfected their statutory rights of appraisal in respect of such Company Shares in accordance with Section 262 of the DGCL (collectively, "Dissenting Company Shares") shall not be converted into, or represent the right to receive, the Merger Consideration. Such Company Stockholders shall be entitled to receive payment of the appraised value of such Dissenting Company Shares in accordance with the provisions of Section 262 of the DGCL, except that all Dissenting Company Shares held by the stockholders of the Company who shall have failed to perfect or who shall have effectively withdrawn or waived or lost their rights to appraisal of such Dissenting Company Shares under such Section 262 of the DGCL shall no longer be considered to be Dissenting Company Shares and shall thereupon be deemed to have been converted into, and to have become exchangeable for, as of the Effective Time, the right to receive the Merger Consideration, without interest thereon, upon surrender of the certificate or certificates (or other appropriate evidence of transfer in the case of Uncertificated Shares) that formerly evidenced such Company Shares in the manner provided in Section 2.10(c).
- (b) Prior to the Closing, the Company shall give Parent (i) prompt notice of any demands for appraisal received by the Company and any withdrawals of such demands, and (ii) the opportunity to participate in all negotiations and proceedings with respect to demands for appraisal under the DGCL. The Company shall not, except with the prior written consent of Parent (which consent shall not be unreasonably withheld, conditioned or delayed), make any payment with respect to any demands for appraisal or offer to settle or settle any such demands.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to the Parent and the Merger Sub that the statements contained in this <u>Article III</u> are true and correct, except as set forth herein or in the Company Disclosure Schedule (with specific reference to the section of this Agreement to which the information stated in such disclosure relates; <u>provided that</u> information contained in any section of the Company Disclosure Schedule shall be deemed to be disclosed with respect to any other section of this Agreement to the extent that it is readily apparent from the face of such disclosure that such information is applicable to such other section of this Agreement) delivered and/or otherwise made available by the Company to the Parent and the Merger Sub as of the date of this Agreement (the "Company Disclosure Schedule").

3.1. Organization; Power and Enforceability. The Company is a corporation duly organized, validly existing and in good standing under the Laws of the state of Delaware and has full corporate power and authority to own, operate or lease the properties and assets now owned, operated or leased by it and to carry on its business as it has been and is currently conducted. Section 3.1 of the Company Disclosure Schedule sets forth each jurisdiction in which the Company is licensed or qualified to do business, and the Company is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the properties owned or leased by it or the operation of its business as currently conducted makes such licensing or qualification necessary, except where the failure to be so duly qualified or licensed and in good standing, individually or in the aggregate, would not have a Company Material Adverse Effect. True and complete copies of the Organizational Documents of the Company as of the Effective Time have been provided to the Parent. This Agreement has been duly executed and delivered by the Company, and, assuming due authorization, execution, and delivery by the Parent and the Merger Sub, this Agreement constitutes a legal, valid, and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by the General Enforceability Exceptions.

3.2. Authority.

(a) The Company has all requisite corporate power and authority to enter into and to perform its obligations under this Agreement. The execution and delivery of this Agreement by the Company and the consummation by the Company of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of the Company and no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement or to consummate the Merger and the other Transactions. The Written Consent, if executed and delivered by all of the Company Stockholders that have executed Company Voting Agreements, would satisfy the Company Stockholder Approval, which is the only vote or consent of the holders of any class or series of the Company's Equity Interests necessary to approve and adopt this Agreement, approve the Merger, and consummate the Merger and the other Transactions pursuant to applicable Law or the Organizational Documents of the Company.

(b) The Company Board by resolutions duly adopted by either unanimous vote at a meeting of all members of the Company Board duly called and held or unanimous written consent and, not subsequently rescinded or modified in any way, has: (i) determined that this Agreement and the Transactions are fair to and in the best interests of the Company and its stockholders, (ii) approved, adopted and declared advisable this Agreement and the Transactions, (iii) directed that the adoption of this Agreement be submitted to a vote at a meeting of the Company's shareholders or action by written consent in lieu of a meeting and (iv) recommended the adoption of this Agreement and approval of the Transactions by the Company's shareholders.

3.3. Capitalization.

- (a) Section 3.3(a) of the Company Disclosure Schedule lists each record and beneficial owner of the Company Shares and the number of Company Shares held by such owner, and the number of Company Shares reserved and available for issuance pursuant to any Company stock option or equity incentive plans ("Company Stock Plans") and the number of Company Shares that are subject to outstanding options under Company Stock Plans, as of the date hereof. Each Company Stockholder has good and valid title to the Company Shares, free and clear of all Encumbrances. The Company Shares constitute 100% of the total issued and outstanding Equity Interests in the Company as of the date hereof. The Company Shares have been duly authorized and are validly issued.
- (b) The Company Shares were issued in compliance with applicable Laws. The Company Shares were not issued in violation of the Organizational Documents of the Company or any other agreement, arrangement, or commitment to which any Company Stockholder or the Company is a party and are not subject to or in violation of any preemptive or similar rights of any Person.
- (c) Except as set forth on Section 3.3(c) of the Company Disclosure Schedule, there are no outstanding or authorized options, warrants, convertible securities or other rights, agreements, arrangements or commitments of any character relating to any Equity Interests in the Company or obligating any Company Stockholder or the Company to issue or sell any Equity Interests (including the Company Shares), or any other interest, in the Company. Other than the Organizational Documents, there are no voting trusts, proxies or other agreements or understandings in effect with respect to the voting or transfer of any of the Company Shares.
 - 3.4. No Subsidiaries. The Company does not own, or have any interest in any shares or have an ownership interest in any other Person.
- 3.5. No Conflicts. The execution, delivery and performance by the Company of this Agreement and the consummation of the Transactions, do not and will not: (a) conflict with or result in a violation or breach of, or default under, any provision of the Organizational Documents of the Company; (b) conflict with or result in a violation or breach of any provision of any Law or Governmental Order applicable to the Company; (c) except as set forth on Section 3.5 of the Company Disclosure Schedule, require the consent, notice or other action by any Person under, conflict with, result in a violation or breach of, constitute a default or an event that, with or without notice or lapse of time or both, would constitute a default under, result in the acceleration of or create in any party the right to accelerate, terminate, modify or cancel any

22 of 96

Company Material Contract; or (d) result in the creation or imposition of any Encumbrance other than Company Permitted Encumbrances on any properties or assets of the Company except, with respect to clauses (b), (c) and (d), for any such conflicts, violations, breaches, defaults or other occurrences that, individually or in the aggregate, would not have a Company Material Adverse Effect.

3.6. No Consents. No consent, waiver, approval, license, Permit, Governmental Order, declaration or filing with, or notice to, any Governmental Authority is required by or with respect to the Company in connection with the execution, delivery and performance of this Agreement and the consummation of the Transactions, except (i) the filing of the Certificate of Merger and the Reincorporation and Charter Amendment with the Secretary of State of the State of Delaware and such filings with Governmental Authorities to satisfy the applicable laws of states in which the Company is qualified to do business; (ii) such filings and applications as are required by Nasdaq; (iii) such filings and approvals as may be required by any federal or state securities laws, including compliance with any applicable requirements of the Exchange Act; (iv) compliance with any applicable requirements of the HSR Act; and (v) such other Consents the failure of which to obtain would not, individually or in the aggregate, prevent or materially delay, impair or hinder the consummation of the Merger or the ability of the Company to perform its covenants and obligations pursuant to this Agreement.

3.7. Financial Statements.

electro220769 ex2-1.htm

- (a) Section 3.7(a) of the Company Disclosure Schedule contains complete copies of the Company's unaudited financial statements for the period from January 1, 2021 through December 31, 2021 (the "Financial Statements").
- (b) The Financial Statements have been prepared in accordance with GAAP applied on a consistent basis throughout the period involved, subject to normal and recurring year-end adjustments (the effect of which will not be materially adverse) and the absence of notes. The Financial Statements are based on the books and records of the Company, and fairly present in all material respects the financial condition of the Company as of the respective dates they were prepared and the results of the operations of the Company for the periods indicated.

3.8. Information Provided.

- (a) None of the information supplied or to be supplied by or on behalf of the Company for inclusion or incorporation by reference in the registration statement on Form S-4 (or such other available form agreed to by the Company) to be filed with the SEC by Parent in connection with the Parent Stock Issuance (as amended or supplemented from time to time, and including the Proxy Statement/Consent/Prospectus contained therein, the "Registration Statement") shall, at the time the Registration Statement is filed with the SEC, and at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading.
- (b) None of the information supplied or to be supplied by or on behalf of the Company for inclusion or incorporation by reference in any filing pursuant to Rule 165 and Rule

425 under the Securities Act or Rule 14a-12 under the Exchange Act (each a "Regulation M-A Filing"), shall, at the time any such Regulation M-A Filing is filed with the SEC, at any time it is amended or supplemented, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading.

- (c) None of the information supplied or to be supplied by or on behalf of the Company for inclusion in the proxy statement and prospectus of the Parent and the Consent Solicitation Statement of the Company to be filed with the SEC and sent to the Parent's shareholders and the Company's stockholders in connection with the Parent Stock Issuance, the Company Stockholder Approval and the other matters to be approved as required by this Agreement (including any amendments or supplements thereto, the "Proxy Statement/Consent /Prospectus") shall, at the date it is first mailed to the Parent's shareholders, at the time of the Parent Shareholders Meeting, at the time of the Written Consent is first delivered to the Company's stockholders or at the time of any amendment or supplement thereof, contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading.
- (d) None of the information supplied or to be supplied by or on behalf of the Company for inclusion or incorporation by reference in any SEC filing relating to the PIPE Investment or in any documents or other materials provided by the Company in connection therewith shall, at the time any such filing is made with the SEC or provided to potential investors or third parties in connection therewith, or at any time it is amended or supplemented or, if it is a registration statement, at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 3.9. Brokers. Except as set forth on Section 3.9 of the Company Disclosure Schedule, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions based upon arrangements made by or on behalf of the Company or any Company Stockholder.
- 3.10. Access to Information; Disclaimer. The Company acknowledges and agrees that it (a) has had an opportunity to discuss the business of Parent with the management of Parent, (b) has been afforded the opportunity to ask questions of and receive answers from officers of Parent and (c) has conducted its own independent investigation of Parent, its business and the transactions contemplated hereby, and has not relied on any representation, warranty or other statement by any Person on behalf of Parent or otherwise, other than the representations and warranties of Parent and Merger Sub expressly contained in Article V of this Agreement and that all other representations and warranties are specifically disclaimed. Without limiting the foregoing, the Company further acknowledges and agrees that none of Parent or any of its shareholders, directors, officers, employees, Affiliates, advisors, agents or other representatives has made any representation or warranty concerning any estimates, projections, forecasts, business plans or other forward-looking information regarding Parent or its business and operations. The Company hereby acknowledges that there are uncertainties inherent in

attempting to develop such estimates, projections, forecasts, business plans and other forward-looking information with which the Company is familiar, that the Company is taking full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections, forecasts, business plans and other forward-looking information furnished to them (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, business plans and other forward-looking information), and that the Company will have no claim, and hereby releases all claims, against Parent or any of its shareholders, directors, officers, employees, Affiliates, advisors, agents or other representatives with respect thereto.

3.11. Ownership of Parent Common Stock. None of the Company or its Subsidiaries directly or indirectly owns, beneficially or otherwise, any of the outstanding Parent Common Stock. None of the Company or its Affiliates or associates is, or has in the last four (4) years been, an "interested shareholder" under Section 302A.011, Subd. 49, of the MBCA.

ARTICLE IV ADDITIONAL REPRESENTATIONS AND WARRANTIES OF COMPANY

The Company represents and warrants to the Parent and the Merger Sub that the statements contained in this Article IV are true and correct, except as set forth herein or in the Company Disclosure Schedule (with specific reference to the section of this Agreement to which the information stated in such disclosure relates; provided that information contained in any section of the Company Disclosure Schedule shall be deemed to be disclosed with respect to any other section of this Agreement to the extent that it is readily apparent from the face of such disclosure that such information is applicable to such other section of this Agreement), with respect to any business operated by the Company before the Effective Time and solely to the extent, and for the period, the Company has operated such business.

4.1. Absence of Certain Changes or Events. Since January 1, 2022, and other than as set forth on Section 4.1 of the Company Disclosure Schedule, there has not been, with respect to the Company, any (a) Company Material Adverse Effect; (b) amendment of the Organizational Documents of the Company; (c) split, combination, reclassification, redemption, purchase or acquisition of any Equity Interests in the Company; (d) issuance, sale or other disposition of, or creation of any Encumbrance on, any Equity Interests in the Company, or grant of any options, warrants or other rights to purchase or obtain (including upon conversion, exchange or exercise) any Equity Interests in the Company; or (e) action or omission outside of the ordinary course of business of the Company.

4.2. Material Contracts.

(a) Except as set forth in Section 4.2(a) of the Company Disclosure Schedule, the Company is not a party or subject to (i) any Contract that would be considered a material contract (as defined in Item 601(b)(10) of Regulation S-K) with respect to the Company (assuming the Company was subject to the requirements of the Exchange Act); (ii) any Contract under which the consequences of a default or termination would reasonably be likely to have a Company Material Adverse Effect; (iii) any Contract that limits or purports to limit the ability of the Company to compete in any line of business or with any Person or in any geographic area or

during any period of time; (iv) any Contract related to the acquisition or disposition of any business, a material amount of equity or assets of any other Person or any real property (whether by merger, sale of stock or other Equity Interests, sale of assets or otherwise); (v) any Contract related to any Indebtedness of the Company; or (vi) any other Contract that is material to the Company and not previously disclosed pursuant to this Section 4.2 (each, a "Company Material Contract").

(b) Each Company Material Contract is valid and binding on the Company in accordance with its terms and is in full force and effect. None of the Company or, to the Company's Knowledge, any other party thereto is in breach of or default under (or is alleged to be in breach of or default under) in any material respect, or has provided or received any notice of any intention to terminate, any Company Material Contract (other than that the Company has cured or that would not constitute a Company Material Adverse Effect). No event or circumstance has occurred that, with notice or lapse of time or both, would constitute an event of default under any Company Material Contract or result in a termination thereof or would cause or permit the acceleration or other changes of any right or obligation or the loss of any benefit thereunder, except as would not have a Company Material Adverse Effect. Complete and correct copies of each Company Material Contract (including all modifications, amendments, and supplements thereto and waivers thereunder) have been made available to the Parent.

4.3. Assets and Properties.

- (a) Except as set forth on Section 4.3 of the Company Disclosure Schedule, the Company has good and valid title to, or a valid leasehold interest in, all Real Property and personal property and other assets reflected in the Financial Statements or acquired after January 1, 2022, other than properties and assets sold or otherwise disposed of in the ordinary course of business consistent with past practice since January 1, 2022. All such properties and assets (including leasehold interests) are free and clear of Encumbrances except for the following (collectively referred to as "Company Permitted Encumbrances"): (i) liens for Taxes not yet delinquent or which are being contested in good faith through appropriate proceedings; or (ii) mechanics, carriers', workmen's, repairmen's or other like liens arising or incurred in the ordinary course of business consistent with past practice or amounts that are not delinquent and which are not, individually or in the aggregate, material to the business of the Company.
- (b) The Company does not own any real property. With respect to the Real Property, the Company has delivered or made available to Parent true, complete and correct copies of any leases or easements affecting the Real Property. The Company is not a sublessor or grantor under any sublease or other instrument granting to any other Person any right to the possession, lease, occupancy, or enjoyment of any leased Real Property. The use and operation of the Real Property in the conduct of the Company's business do not violate in any material respect any Law, covenant, condition, restriction, easement, license, permit or agreement applicable to the Real Property.
- 4.4. Intellectual Property. Except as set forth in <u>Section 4.4</u> of the Company Disclosure Schedule, the Company owns, or has the licenses or rights to use, all Intellectual Property used by the Company to conduct its business. Except as set forth in <u>Section 4.4</u> of the Company Disclosure Schedule, the Company (a) has not and does not, to the Company's

Knowledge, interfere with, infringe upon, dilute, misappropriate, or violate the Intellectual Property of any Person, (b) has not received any written charge, complaint, claim, demand, or notice alleging interference, infringement, dilution, misappropriation, or violation of the Intellectual Property of any Person, and (c) other than as may be expressly set forth in any Company Material Contract, has no enforceable agreement that is primarily for the purpose of indemnifying any Person for or against any interference, infringement, dilution, misappropriation, or violation with respect to any Intellectual Property.

- 4.5. Indebtedness. Section 4.5 of the Company Disclosure Schedule sets forth a true, correct and complete list of all the Indebtedness outstanding as of the date hereof with respect to the Company, in an amount in excess of \$25,000, individually, other than Indebtedness reflected in the Financial Statements of the Company.
 - 4.6. Legal Proceedings; Governmental Orders.
- (a) Except as set forth in Section 4.6(a) of the Company Disclosure Schedule, there are no Actions pending or, to the Company's Knowledge, threatened (i) against or by the Company affecting any of its properties or assets (or by or against any Affiliate of the Company and relating to the Company); or (ii) against or by the Company or any Affiliate of the Company that challenges or seeks to prevent, enjoin or otherwise delay the transactions contemplated by this Agreement.
- (b) There are no outstanding Governmental Orders and no unsatisfied judgments, penalties or awards against or affecting the Company or any of its properties or assets.
- 4.7. Compliance With Laws. The Company is currently and has been in compliance in all material respects with all Laws applicable to it or its business, properties or assets.
- 4.8. Permits. Except as set forth on Section 4.8 of the Company Disclosure Schedule, all Permits required for the Company to conduct its business have been obtained by it and are valid and in full force and effect, except as would not have a Company Material Adverse Effect. All fees and charges with respect to such Permits as of the date hereof have been paid in full. Section 4.8 of the Company Disclosure Schedule lists all current Permits issued to the Company. No event has occurred that, with or without notice or lapse of time or both, would reasonably be expected to result in the revocation, suspension, lapse or limitation of any Permit set forth in Section 4.8 of the Company Disclosure Schedule.
- 4.9. Affiliate Interests. Neither the Company, nor any of its employees, directors, officers or equityholders, nor any of their respective Affiliates has any interest in any asset owned or used by the Company in the operation of its business. No director, officer or equityholder or, to the Knowledge of the Company, any employee of the Company, or any Affiliate of any of the foregoing is an Affiliate of or the owner (of record or beneficially) of any Equity Interests or any other financial or profit interest in, a Person that has (a) except as set forth in Section 4.9 of the Company Disclosure Schedule, had business dealings or a material financial interest in any transaction with the Company, or (b) engaged in competition with the Company, in either case other than: (x) ownership of less than two (2) percent of the outstanding capital stock of a Person that is listed on any national or regional securities exchange, and (y) for

compensation for services actually performed as an employee or independent contractor of the Company.

- 4.10. Taxes. Except as set forth on <u>Section 4.10</u> of the Company Disclosure Schedule:
- (a) The Company has timely filed all material Tax Returns which are required to be filed by it, and all such Tax Returns are true, complete and correct in all material respects. All material Taxes due and owing by the Company (whether or not shown on any Tax Return) have been fully and timely paid.
- (b) No deficiency or proposed adjustment which has not been paid or resolved for any amount of income or other material Tax has been asserted or assessed by any Governmental Authority against the Company. The Company has not consented to extend the time in which any income or other material Tax may be assessed or collected by any Governmental Authority, which extension is still in effect. There are no ongoing or pending income or material Tax audits by any Governmental Authority against the Company. The Company is not a party to or bound by, or has any obligation under any income or material Tax allocation, sharing, indemnity or similar agreement or arrangement (other than any agreement not primarily concerning Taxes).
- (c) The Company is not a party to a transaction that is a "reportable transaction," as such term is defined in Treasury Regulations Section 1.6011-4(b) (or any similar provision of state, local or non-U.S. Law).
- (d) The Company is (and has since formation been treated as) a corporation for U.S. federal income Tax purposes. The Company has not taken or agreed to take action that would prevent the Merger from constituting a tax-free reorganization under Section 368(a)(1)(A) of the Code by reason of Section 368(a)(2)(E) of the Code.
- 4.11. Books and Records. The minute books of the Company have been made available to Parent and have been maintained in all material respects in accordance with sound business practices. The minute books of the Company contain accurate and complete, in all material respects, records of all meetings, and actions taken by written consent of, the Company Stockholders, and no meeting, or action taken by written consent, of any such Company Stockholders has been held for which minutes have not been prepared and are not contained in such minute books. At the Closing, all of those books and records will be in the possession of the Company.
- 4.12. Undisclosed Liabilities. Except as set forth on Section 4.12 of the Company Disclosure Schedule, the Company has no liabilities, obligations or commitments of any nature whatsoever, asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured, or otherwise ("Liabilities"), in each case required to be included on any financial statements prepared in accordance with GAAP, except (a) those which are adequately reflected or reserved against in the Financial Statements, (b) those which have been incurred in the ordinary course of business since January 1, 2022; (c) liabilities for performance of obligations of the Company under Contracts binding upon the Company (other than resulting from any breach, termination or acceleration of such Contracts) either delivered or

made available to Parent or Parent's Representatives prior to the date hereof or entered into in the ordinary course of business following the date hereof; and (d) as of the date hereof, those which are not, individually or in the aggregate, in excess of \$25,000.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

The Parent and the Merger Sub jointly and severally represent and warrant to the Company that the statements contained in this Article V are true and correct, except as expressly set forth herein or in the disclosure schedule (with specific reference to the section of this Agreement to which the information stated in such disclosure relates; provided that information contained in any section of the disclosure schedule shall be deemed to be disclosed with respect to any other section of this Agreement to the extent that it is readily apparent from the face of such disclosure that such information is applicable to such other section of this Agreement) delivered by the Parent and the Merger Sub to the Company on the date of this Agreement (the "Parent Disclosure Schedule").

5.1. Organization, Power and Enforceability.

- (a) The Parent is a corporation duly organized, validly existing and in good standing under the Laws of the state of Minnesota. At the Effective Time and pursuant to the Reincorporation and Charter Amendment, Parent will be a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. The Parent has full corporate power and authority to enter into this Agreement, to carry out its obligations hereunder, to consummate the Transactions, and to enable it to own, lease or otherwise hold its properties and assets and to conduct its business as presently conducted. Section 5.1 of the Parent Disclosure Schedule sets forth each jurisdiction in which the Parent is licensed or qualified to do business, and the Parent is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the properties owned or leased by it or the operation of its business as currently conducted makes such licensing or qualification necessary except where the failure to be so duly qualified or licensed and in good standing, individually or in the aggregate, would not have a Parent Material Adverse Effect. True and complete copies of the Organizational Documents of the Parent as of the date hereof have been provided to the Company. This Agreement has been duly executed and delivered by the Parent, and (assuming due authorization, execution, and delivery by the Company) this Agreement constitutes a legal, valid, and binding obligation of the Parent enforceable against the Parent in accordance with its terms except as such enforceability may be limited by the General Enforceability Exceptions.
- (b) The Merger Sub is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. The Merger Sub has full corporate power and authority to enter into this Agreement, to carry out its obligations hereunder and to consummate the Transactions. This Agreement has been duly executed and delivered by the Merger Sub, and (assuming due authorization, execution, and delivery by the Company) this Agreement constitutes a legal, valid, and binding obligation of the Merger Sub enforceable against the Merger Sub in accordance with its terms except as such enforceability may be limited

by the General Enforceability Exceptions. True and complete copies of the Organizational Documents of the Merger Sub as of the date hereof have been provided to the Company. The Merger Sub was formed solely for the purpose of entering into the Transactions, and since the date of its incorporation, the Merger Sub has not carried on any business, conducted any operations or incurred any liabilities or obligations other than those incident to its formation and pursuant to this Agreement, the performance of its obligations hereunder and matters ancillary thereto.

5.2. Authority.

- Each of the Parent and the Merger Sub has all requisite corporate power and authority to enter into and to perform its obligations under this Agreement and, subject to obtaining the Parent Shareholder Approval to consummate the Transactions. The execution and delivery of this Agreement by the Parent and the Merger Sub and the consummation by the Parent and the Merger Sub of the transactions contemplated by this Agreement have been duly authorized by all necessary corporate action on the part of the Parent and the Merger Sub and no other corporate proceedings on the part of the Parent or the Merger Sub are necessary to authorize the execution and delivery of this Agreement or to consummate the Transactions, including the Merger and the Parent Stock Issuance, subject only to: (1) the adoption of this Agreement by the Parent as the sole shareholder of the Merger Sub; and (2) obtaining the Parent Shareholder Approval. The Parent Shareholder Approval is the only vote of the holder of any class or series of the Parent's Equity Interests necessary to adopt and approve this Agreement, approve the Merger, and consummate the Merger and the other Transactions pursuant to applicable Law or the Organizational Documents of the Parent. The Parent Shareholder Approval will have been validly and duly obtained if all of the holders of Parent Common Stock that have executed Parent Voting Agreements all vote in the manner required by such agreements.
- (b) The Parent Board, and any applicable committee thereof required pursuant to the provisions of the MBCA, by resolutions duly adopted by a unanimous vote at a meeting of all directors of the Parent or all members of any such committee duly called and held and not subsequently rescinded or modified in any way has (i) determined that this Agreement, the Transactions, and the Reincorporation and Charter Amendment, upon the terms and subject to the conditions set forth in this Agreement, are fair to, and in the best interests of, the Parent and the Parent's shareholders, (ii) approved and declared advisable this Agreement and the Parent Voting Agreement, including the execution, delivery, and performance thereof, the consummation of the Transactions and the Reincorporation and Charter Amendment, (iii) directed that this Agreement, the Transactions, and Reincorporation and Charter Amendment be submitted to a vote of the Parent's shareholders for adoption at the Parent Shareholders Meeting, and (iv) resolved to recommend that Parent's shareholders vote in favor of approval of this Agreement, the Transactions, and the Reincorporation and Charter Amendment (collectively, the "Parent Board Recommendation").
- (c) The Merger Sub Board by resolutions duly adopted by a unanimous vote at a meeting of all directors of Merger Sub duly called and held and not subsequently rescinded or modified in any way has (i) determined that this Agreement and the transactions contemplated hereby, including the Merger, upon the terms and subject to the conditions set forth herein, are

fair to, and in the best interests of, Merger Sub and the Parent, as the sole stockholder of Merger Sub, (ii) approved and declared advisable this Agreement, including the execution, delivery, and performance thereof, and the consummation of the transactions contemplated by this Agreement, including the Merger, upon the terms and subject to the conditions set forth herein, and (iii) resolved to recommend that the Parent, as the sole stockholder of Merger Sub, approve the adoption of this Agreement in accordance with the DGCL.

- (d) The Parent validly opted-out of Section 302A.671 by expressly providing in its articles of incorporation approved by its shareholders that such section does not apply to the Parent. In addition, on or prior to the date hereof, a committee consisting of only disinterested directors of the Parent Board, which complies with the requirements for such a committee set forth in Section 302A.673, Subd.1(d), of the MBCA, and acting in accordance with Section 302A.673 of the MBCA has duly and unanimously:
 - (i) approved this Agreement, the Parent Voting Agreement, the Certificate of Merger, the Parent Stock Issuance and the Transactions, which approval, to the extent applicable and assuming the accuracy of the representations and warranties of the Company set forth in Section 3.11, constituted approval for the purposes of Sections 302A.673 of the MBCA; and
 - (ii) recommended to the Parent Board that the Parent Board approve this Agreement, the Parent Stock Issuance and the Transactions.
- (e) As a result of the actions referred to in Section 5.2(d), this Agreement, the Parent Stock Issuance and the Transactions, including the Merger, the Parent Voting Agreement and the other Transactions, are not and will not be subject to the restrictions on "control share acquisitions" or "business combinations" with an "interested shareholder" under the provision of Sections 302A.671 and 302A.673, respectively, of the MBCA. No other state takeover statute or similar statute or regulation or similar charter provision applies to or purports to apply to the Merger or the other Transactions.

5.3. Capital Structure.

(a) As of the date of this Agreement, the authorized capital stock of the Parent consists of 10,000,000 shares of Parent Common Stock, par value \$.10 per share. At the close of business on the date of this Agreement (such date and time, the "Measurement Date"), (i) 3,395,521 shares of Parent Common Stock were issued and outstanding; (ii) 607,500 shares of Parent Common Stock were reserved and available for issuance pursuant to the Parent Stock Plans, of which 332,500 shares were subject to outstanding options and 275,000 shares are eligible for grant under future award. As of the Measurement Date, no (A) shares of capital stock or other voting securities of; (B) other equity or voting interests in; (C) securities convertible into or exchangeable for, or options, warrants or other rights to acquire or receive any, capital stock, voting securities or other equity interests in; or (D) stock appreciation rights, "phantom" stock rights, or other rights that give the holder thereof any economic or voting interest of a nature accruing to the holders of capital stock in (clauses (A), (B), (C) and (D), collectively, "Equity Interests") the Parent were issued, reserved for issuance or outstanding except as set forth in this Section 5.3(a) or otherwise contemplated for issuance pursuant to this Agreement. From and

after the Measurement Date through the date of this Agreement, the Parent has not (i) issued any Equity Interests or (ii) incurred any obligation to make any payments based on the price or value of any Equity Interests or dividends (or other distributions) paid thereon or revenues, earnings or financial performance or any other attribute of the Parent, in each case other than pursuant to Parent's equity-based awards, purchase rights and stock awards granted pursuant to the Parent Stock Plans, in each case that were outstanding as of the Measurement Date, and in accordance with their respective terms as in effect at such time.

- (b) All issued and outstanding Equity Interests in the Parent are, and at the time of issuance all Equity Interests in the Parent that may be issued prior to the Effective Time, including all shares that may be issued pursuant to the Parent Stock Plans and the Parent Stock Issuance, will be, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the laws of the jurisdiction in which the Parent is organized, the Organizational Documents of the Parent or any Contract to which the Parent is a party or otherwise bound.
- (c) There are no outstanding bonds, debentures, notes or other Indebtedness of the Parent that may have at any time (whether actual or contingent) the right to vote, or that are convertible into or exchangeable for securities having the right to vote, on any matters on which holders of shares of the Parent may vote ("Voting Parent Debt") or any securities that are convertible into or exchangeable for, or options, warrants or other rights to acquire or receive any, Voting Parent Debt.
- Except for awards made under the Parent Stock Plans in accordance with their respective terms as in effect as of the date hereof, the Parent Stock Issuance, or the PIPE Investment, there are not any outstanding obligations of the Parent to issue, deliver or sell, or cause to be issued, delivered or sold, any Equity Interests in the Parent. There are not any outstanding obligations of the Parent to directly or indirectly redeem, repurchase or otherwise acquire any Equity Interests in the Parent, except for (i) acquisitions of shares of Parent Common Stock in connection with the surrender of shares of Parent Common Stock by holders of stock options under the Parent Stock Plans in order to pay the exercise price of such stock options; (ii) the withholding of shares of Parent Common Stock to satisfy Tax obligations with respect to stock options or stock-based awards made pursuant to a Parent Stock Plan; or (iii) the acquisition by the Parent of stock options or stock-based awards in connection with the forfeiture of such awards, in each case in accordance with their respective terms under the Parent Stock Plans. The Parent is not a party to any agreement with respect to the voting, transfer or registration of any capital stock or voting securities of, or other Equity Interests in, the Parent. Except as contemplated by this Agreement, the Parent is not a party to any agreement pursuant to which any Person is entitled to elect, designate or nominate any director of the Parent. No Subsidiary of the Parent owns any shares of capital stock of the Parent.
- (e) Section 5.3(e) of the Parent Disclosure Schedule sets forth the name of each holder of an outstanding Parent Stock Option granted under the Parent Stock Plans, identifying the nature of the award, the number of shares of Parent Common Stock subject to each Parent Stock Option, the grant, vesting and expiration dates and the exercise price relating to the Parent Stock Options held. Each Parent Stock Option (i) was granted in compliance with

all applicable laws and all of the terms and conditions of the Parent Stock Plan pursuant to which it was issued, and (ii) has an exercise price per share of Company Common Stock equal to or greater than the fair market value of a share of Parent Common Stock on the date of such grant.

- 5.4. No Conflicts. The execution, delivery and performance by the Parent and the Merger Sub of this Agreement, and the consummation of the Transactions, do not and will not (a) conflict with or result in a violation or breach of, or default under, any provision of the Organizational Documents of the Parent or the Merger Sub, subject to receipt of the Parent Shareholder Approval; (b) conflict with or result in a violation or breach of any provision of any Law or Governmental Order applicable to the Parent or the Merger Sub; (c) require the consent, notice or other action by any Person under, conflict with, result in a violation or breach of, constitute a default or an event that, with or without notice or lapse of time or both, would constitute a default under, result in the acceleration of or create in any party the right to accelerate, terminate, modify or cancel any Parent Material Contract; or (d) result in the creation or imposition of any Encumbrance other than Parent Permitted Encumbrances on any properties or assets of the Parent or the Merger Sub, except, with respect to clauses (b), (c) and (d), for any such conflicts, violations, breaches, defaults or other occurrences that, individually or in the aggregate, would not have a Parent Material Adverse Effect.
- 5.5. Consents. Except as set forth on Section 5.5 of the Parent Disclosure Schedule and as otherwise set forth in and contemplated by this Agreement, no consent, waiver, approval, license, Permit, Governmental Order, declaration or filing with, or notice to, any Governmental Authority is required by or with respect to the Parent or the Merger Sub in connection with the execution, delivery and performance of this Agreement and the consummation of the Transactions, except (i) the filing of the Certificate of Merger and the Reincorporation and Charter Amendment with the Secretary of State of Delaware and such filings with Governmental Authorities to satisfy the applicable laws of states in which the Company is qualified to do business; (ii) such filings and approvals as may be required by any federal or state securities laws, including compliance with any applicable requirements of the Exchange Act; (iv) compliance with any applicable requirements of the HSR Act; and (v) such other Consents the failure of which to obtain would not, individually or in the aggregate, prevent or materially delay, impair or hinder the consummation of the Merger or the ability of Parent and Merger Sub to perform their respective covenants and obligations pursuant to this Agreement.
 - 5.6. SEC Filings; Financial Statements; Information Provided.
- (a) The Parent has filed all registration statements, forms, reports, certifications and other documents required to be filed by the Parent with the SEC since January 1, 2018. All such registration statements, forms, reports and other documents, as amended prior to the date hereof, and those that the Parent may file after the date hereof until the Closing, are referred to herein as the "Parent SEC Reports." All of the Parent SEC Reports (A) were or will be filed on a timely basis, (B) at the time filed (or if amended prior to the date hereof, when so amended), complied, or will comply when filed, as to form in all material respects with the requirements of the Securities Act and the Exchange Act applicable to such Parent SEC Reports and (C) did not or will not at the time they were filed (or if amended prior to the date hereof, when so amended) or are filed contain any untrue statement of a material fact or omit to state a

material fact required to be stated in such Parent SEC Reports or necessary in order to make the statements in such Parent SEC Reports, in the light of the circumstances under which they were made, not misleading, in any material respect.

- Each of the financial statements (including, in each case, any related notes and schedules) contained or to be contained in the Parent SEC Reports at the time filed (or if amended prior to the date hereof, when so amended) (i) complied or will comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, (ii) were or will be prepared in accordance with GAAP applied on a consistent basis throughout the periods involved and at the dates involved (except as may be indicated in the notes to such financial statements or, in the case of unaudited interim financial statements, as permitted by the SEC on Form 10-Q under the Exchange Act), and (iii) fairly presented or will fairly present in all material respects the financial position of Parent as of the dates indicated and the assets, liabilities, business, financial condition, results of its operations and cash flows for the periods indicated, consistent with the books and records of Parent, except that the unaudited interim financial statements were or are subject to normal and recurring year-end adjustments as permitted by GAAP and the applicable rules and regulations of the SEC (but only if the effect of such adjustments would not, individually or in the aggregate, be material). The balance sheet of Parent as of December 31, 2021 is referred to herein as the "Parent Balance Sheet."
- Boulay PLLP, Parent's current auditors, is and has been at all times since its engagement by Parent (i) "independent" with respect to Parent within the meaning of Regulation S-X and (ii) in compliance with subsections (g) through (1) of Section 10A of the Exchange Act (to the extent applicable) and the related rules of the SEC and the Public Company Accounting Oversight Board.
- None of the information supplied or to be supplied by or on behalf of the Parent or the Merger Sub for inclusion or incorporation by reference in the Registration Statement or any Regulation M-A Filing will, at the time the Registration Statement or any Regulation M-A Filing is filed with the SEC, and at any time it is amended or supplemented or at the time the Registration Statement becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading. None of the information supplied or to be supplied by or on behalf of the Parent or the Merger Sub for inclusion or incorporation by reference in the Proxy Statement/Consent/Prospectus will, at the date it is first mailed to the Parent's shareholders or at the time of the Parent Shareholders Meeting or at the time of any amendment or supplement thereof, contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading. The Proxy Statement/Consent/Prospectus will comply as to form in all material respects with the requirements of the Exchange Act. None of the information supplied or to be supplied by or on behalf of the Parent or the Merger Sub for inclusion or incorporation by reference in any SEC filing relating to the PIPE Investment or in any documents or other materials provided by the Parent in connection therewith shall, at the time any such filing is made with the SEC or provided to potential investors or third parties in connection therewith, or at any time it is amended or supplemented or, if it is a registration statement, at the time it becomes effective

under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

- The Parent has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 under the Exchange Act. Parent's disclosure controls and procedures are reasonably designed to ensure that all material information required to be disclosed by Parent in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to Parent's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. Since January 1, 2018, the Parent's principal executive officer and its principal financial officer have disclosed to the Parent's auditors and the audit committee of the Parent Board all known (i) significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting that are reasonably likely to adversely and materially affect the Company's ability to record, process, summarize and report financial information, (ii) material weaknesses in the design and operation of internal controls over financial reporting, and (iii) any fraud, whether or not material, that involves the management or other employees who have a significant role in the Parent's internal controls over financial reporting. The Parent has materially complied with or substantially addressed any such deficiencies, material weaknesses or fraud.
- (f) The Parent is in compliance in all material respects with the applicable provisions of the Sarbanes-Oxley Act and the Dodd-Frank Wall Street Reform and Consumer Protection Act. Each required form, report and document containing financial statements that has been filed with or submitted to the SEC was accompanied by any certifications required to be filed or submitted by Parent's principal executive officer and principal financial officer pursuant to the Sarbanes-Oxley Act and, at the time of filing or submission of each such certification, any such certification complied in all material respects with the applicable provisions of the Sarbanes-Oxley Act and the statements contained in such certifications were true and correct on the date such certifications were made. Neither Parent nor any of its executive officers has received written notice from any Governmental Authority challenging or questioning the accuracy, completeness, form or manner of filing of such certifications.
- (g) The issued and outstanding shares of Parent Common Stock are registered pursuant to Section 12(b) of the Exchange Act and are listed for trading on the Nasdaq under the symbol "ELSE." The Parent is in compliance in all material respects with the applicable listing and other rules and regulations of Nasdaq. The Parent has not received any notice of noncompliance or deficiencies with any applicable listing or other rules and regulations of Nasdaq, and no fact or circumstance exists that, to the Knowledge of the Parent, would reasonably be expected to result in such notice of noncompliance or deficiency as of the date hereof or within one year of the Closing Date (provided, if the sole noncompliance or deficiency is failure to comply with the \$1.00 minimum bid price, then such noncompliance or deficiency shall be disregarded provided it can reasonably be cured by a reverse stock split within such one year period). There is no Action pending or, to the knowledge of Parent, threatened against

Parent by the Nasdaq or the SEC with respect to any intention by such entity to deregister the Parent Common Stock or terminate the listing of Parent on the Nasdaq Capital Market. None of Parent or any of its Affiliates has taken any action in an attempt to terminate the registration of the Parent Common Stock under the Exchange Act.

- (h) The Parent is not an "investment company" within the meaning of the Investment Company Act of 1940, as amended.
- (i) As of the date of this Agreement, the Parent has timely responded to all comment letters of the staff of the SEC relating to the Parent SEC Reports, and the SEC has not advised the Parent that any final responses are inadequate, insufficient or otherwise non-responsive. The Parent has made available to the Company true, correct and complete copies of all comment letters, written inquiries and enforcement correspondence between the SEC, on the one hand, and the Parent, on the other hand, occurring since January 1, 2017 and will, reasonably promptly following the receipt thereof, make available to the Company any such correspondence sent or received after the date hereof. To the Knowledge of the Parent, as of the date of this Agreement, none of the Parent SEC Reports is the subject of ongoing SEC review or outstanding SEC comment.
- 5.7. Brokers. Except as set forth on Section 5.7 of the Parent Disclosure Schedule, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Parent. The Parent has made available to the Company prior to the execution of this Agreement a true and complete copy of all agreements between the Parent and its financial advisors pursuant to which such advisors would be entitled to any payment relating to the Transactions.
- 5.8. Absence of Certain Changes or Events. Except as set forth in the Parent SEC Reports, since the date of the Parent Balance Sheet, there has not been, with respect to the Parent, any (a) Parent Material Adverse Effect; (b) amendment of the Organizational Documents of the Parent (other than as required by this Agreement); (c) split, combination, reclassification, redemption, purchase or acquisition of any Equity Interests in the Parent (other than as required by this Agreement); (d) issuance, sale or other disposition of, or creation of any Encumbrance on, any Equity Interests in the Parent, or grant of any options, warrants or other rights to purchase or obtain (including upon conversion, exchange or exercise) any Equity Interests in the Parent; (e) any entry by Parent or any of its Affiliates into, any amendment of or modification to or agreement to amend or modify (or announcement of an intention to so amend or modify) or termination of any Parent Plan; (f) any other grant by Parent or any of its Affiliates of any awards or rights under any Parent Plan; or (g) action or omission outside of the ordinary course of business of the Parent consistent with past practice except as set forth in or contemplated by this Agreement or otherwise disclosed on the Parent Disclosure Schedules.
 - 5.9. Material Contracts.

(a) Except as set forth in <u>Section 5.9(a)</u> of the Parent Disclosure Schedule or as filed or furnished, as applicable, by the Parent with the SEC since January 1, 2018, the Parent is not a party or subject to (i) any Contract that would be considered a material contract (as

defined in Item 601(b)(10) of Regulation S-K) with respect to the Company (assuming the Company was subject to the requirements of the Exchange Act); (ii) any Contract under which the consequences of a default or termination would reasonably be likely to have a Parent Material Adverse Effect; (iii) any Contract that limits or purports to limit the ability of the Parent to compete in any line of business or with any Person or in any geographic area or during any period of time; (iv) any Contract related to the acquisition or disposition of any business, a material amount of equity or assets of any other Person or any real property (whether by merger, sale of stock or other Equity Interests, sale of assets or otherwise); (v) any Contract related to any Indebtedness of the Parent; or (vi) any other Contract that is material to the Company and not previously disclosed pursuant to this Section 5.9 (each, a "Parent Material Contract").

(b) Each Parent Material Contract is valid and binding on the Parent in accordance with its terms and is in full force and effect. The Parent is not in breach of or default under (or is alleged to be in breach of or default under) in any material respect, and as of the date hereof, the Parent has not provided or received any notice of any intention to terminate, any Parent Material Contract (other than that the Parent has cured or that would not constitute a Parent Material Adverse Effect). No event or circumstance has occurred that, with notice or lapse of time or both, would constitute an event of default under any Parent Material Contract or result in a termination thereof or would cause or permit the acceleration or other changes of any right or obligation or the loss of any benefit thereunder, except as would not have a Parent Material Adverse Effect. Complete and correct copies of each Parent Material Contract (including all modifications, amendments, and supplements thereto and waivers thereunder) have been made available to the Company; provided that Parent Material Contracts filed as part of the Parent SEC Reports shall be deemed provided to the Company for purposes of this sentence.

5.10. Taxes.

- (a) The Parent has timely filed all income Tax Returns and all other material Tax Returns that are required to be filed by it, and all such Tax Returns are true, complete and correct in all material respects. All Taxes due and owing by the Parent (whether or not shown on any Tax Return) have been fully and timely paid. All Taxes that the Parent is obligated to withhold from amounts owing to any employee, creditor or third party have been fully paid or properly withheld and, if required, timely paid to the applicable Governmental Authority.
- (b) No deficiency or proposed adjustment which has not been paid or resolved for any amount of income or other material Tax has been asserted or assessed by any Governmental Authority against the Parent. The Parent has not consented to extend the time in which any income or other material Tax may be assessed or collected by any Governmental Authority, which extension is still in effect. There are no ongoing or pending income or other material Tax audits by any Governmental Authority against the Parent. The Parent is not a party to or bound by, or has any obligation under, any income or other material Tax allocation, sharing, indemnity or similar agreement or arrangement (other than any agreement entered into in the ordinary course of business and not primarily concerning Taxes).
- (c) The Parent has not been a member of a group filing a consolidated, combined, unitary or affiliated Tax Return (other the group in which the Parent is the common parent). No claim has been made in writing by a Governmental Authority in a jurisdiction where

Parent files Tax Returns or pay Taxes that the Parent is required to file Tax Returns in or is subject to taxation by such jurisdiction.

- (d) The unpaid Taxes of the Parent (A) did not, as of the date of the most recent financial statements, exceed the reserve for Taxes (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the financial statements and (B) do not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Parent in filing its Tax Returns. There are no Encumbrances for Taxes upon any property or asset of the Parent (other than statutory liens for Taxes not yet due and payable).
- (e) The Parent will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any period (or any portion thereof) ending after the Closing Date as a result of (A) any installment sale or open transaction disposition made on or prior to the Closing Date, (B) any change in or use of an improper accounting method for a taxable period (or portion thereof) ending on or prior to the Closing Date, (C) any agreement with any Governmental Authority entered into prior to Closing, (D) any prepaid amount received or deferred revenue accrued on or prior to the Closing, (E) any intercompany transaction entered into prior to Closing or excess loss account as in existence immediately prior to Closing described in Section 1502 of the Code (or any similar provision of state, local or non-U.S. Law), (F) any election under Section 108(i) of the Code made prior to the Closing, or (G) the application of Section 965 of the Code. Parent has not made an election pursuant to Section 965(h) of the Code.
- (f) Parent is not a partner for Tax purposes with respect to any joint venture, partnership, or other arrangement which is or is properly treated as a partnership for U.S. federal income Tax purposes. Parent (A) is not and has never been a party to a transaction that is a "reportable transaction," as such term is defined in Treasury Regulations Section 1.6011-4(b)(2) (or any similar provision of state, local or non-U.S. Law), or (B) has participated in or plans to participate in any Tax amnesty program. Neither the Parent nor the Merger Sub has distributed the stock of another Person or had its stock distributed in a transaction purported or intended to be governed by Sections 355 or 361 of the Code (or any similar provision of state, local or non-U.S. Law).
- (g) Parent is (and has since formation been treated as) a corporation for U.S. federal income Tax purposes. The Parent has not taken or agreed to take action that would prevent the Merger from constituting a tax-free reorganization under Section 368(a)(1)(A) of the Code by reason of Section 368(a)(2)(E) of the Code.
- (h) Parent has properly collected and remitted any required material sales, use, value added and similar Taxes with respect to sales made or services provided to its customers and has properly received and retained any appropriate Tax exemption certificates or other documentation for all such sales made or services provided without charging or remitting sales, use, value added or similar Taxes that qualify as exempt from sales or similar Taxes.
- (i) There is no contract, agreement, plan or arrangement to which Parent is bound to provide a gross-up or otherwise reimburse any current or former employee, director,

service provider or other person for excise taxes paid pursuant to Section 409A of the Code or Section 4999 of the Code. Parent is a not a party to any agreement, contract, arrangement or plan that has resulted or would result, separately or in the aggregate, in the payment of any "excess parachute payment" within the meaning of Section 280G of the Code (or any corresponding provision of state, local or non-U.S. income Tax Law).

(j) The Parent has (i) not deferred the employer's share of any "applicable employment taxes" under Section 2302 of the CARES Act, (ii) not received or claimed any Tax credits under Sections 7001 through 7005 of the Families First Coronavirus Response Act or Section 2301 of the CARES Act, (iii) not deferred any payroll tax obligations (including those imposed by Section 3101(a) and 3201 of the Code) (for example, by failure to timely withhold, deposit or remit such amounts in accordance with the applicable provisions of the Code and the Treasury Regulations promulgated thereunder) pursuant to or in connection with the Payroll Tax Executive Order, and (iv) not sought and does not intend to seek (nor have any of their Affiliates sought or intends to seek) a covered loan under paragraph (36) of Section 7(a) of the Small Business Act (15 U.S.C. 636(a)), as added by Section 1102 of the CARES Act.

5.11. Assets and Properties.

- (a) The Parent has good and valid title to, or a valid leasehold interest in, all Real Property and personal property and other assets reflected in the Parent Balance Sheet or acquired on or after January 1, 2021, other than properties and assets sold or otherwise disposed of in the ordinary course of business consistent with past practice since January 1, 2021. All such properties and assets (including leasehold interests) are free and clear of Encumbrances except for the following (collectively referred to as "Parent Permitted Encumbrances"): (i) liens for Taxes not yet due and payable or which are being contested in good faith through appropriate proceedings; or (ii) mechanics, carriers', workmen's, repairmen's or other like liens arising or incurred in the ordinary course of business consistent with past practice or amounts that are not delinquent and which are not, individually or in the aggregate, material to the business of the Parent.
- (b) (i) The Parent has good and marketable title to any real property owned by the Parent (the "Parent Owned Real Property"), free and clear of any Liens except Permitted Liens; (ii) no Person holds any outstanding options or rights of first refusal to purchase the Parent Owned Real Property, or any portion of the Parent Owned Real Property or any interest therein; and (iii) there is no condemnation or other proceeding in eminent domain, pending (or, to the Knowledge of the Parent, threatened) or affecting any portion of the Parent Owned Real Property. Section 5.11 of the Parent Disclosure Schedule sets forth the correct street address of each parcel of Parent Owned Real Property and such other information as is reasonably necessary to identify each parcel of Parent Owned Real Property.
- 5.12. Intellectual Property. The Parent owns, or has the licenses or rights to use, all Intellectual Property used by the Parent to conduct its business. Except as set forth on <u>Schedule 5.12</u>, the Parent (a) has not and does not, to the Parent's Knowledge, interfere with, infringe upon, dilute, misappropriate, or violate the Intellectual Property of any Person, (b) has not received any written charge, complaint, claim, demand, or notice alleging interference, infringement, dilution, misappropriation, or violation of the Intellectual Property of any Person,

and (c) other than as may be expressly set forth in any Parent Material Contract, has no enforceable agreement that is primarily for the purpose of indemnifying any Person for or against any interference, infringement, dilution, misappropriation, or violation with respect to any Intellectual Property.

- 5.13. Legal Proceedings. Except as set forth in Section 5.13 of the Parent Disclosure Schedule, there are no Actions pending or, to the Parent's Knowledge, threatened (i) against or by the Parent affecting any of its properties or assets (or by or against any Affiliate of the Parent and relating to the Parent); or (ii) against or by the Parent or any Affiliate of the Parent that challenges or seeks to prevent, enjoin or otherwise delay the Transactions There are no outstanding Governmental Orders and no unsatisfied judgments, penalties or awards against or affecting the Parent or any of its properties or assets.
- 5.14. Indebtedness. Section 5.14 of the Parent Disclosure Schedule sets forth a true, correct and complete list of all the Indebtedness outstanding with respect to the Parent in an amount in excess of \$25,000, individually, other than Indebtedness reflected in the Parent Balance Sheet.
- 5.15. Compliance With Laws. The Parent since January 1, 2018, has been, in compliance in all material respects with all Laws applicable to the Parent or by which it or its business or properties is bound. Since January 1, 2018, no Governmental Authority has issued any written notice or written notification stating that the Parent is not in compliance with any Law, except where any such non-compliance has not had and would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- 5.16. Merger Sub. Merger Sub: (a) has engaged in no business activities other than those related to the transactions contemplated by this Agreement; and (b) is a direct, wholly-owned Subsidiary of Parent.
- 5.17. Fairness Opinion. The Parent Board has received from Lake Street Capital Advisors, LLC a written opinion to the effect that, as of the date of such opinion, and based upon and subject to the factors and assumptions set forth therein, the consideration to be received by the holders of Parent Common Stock pursuant to the Merger is fair, from a financial point of view, to such holders. A copy of such opinion will be made available to the Company as soon as practicable following the date hereof for informational purposes only.
 - 5.18. Employee Benefit Matters.
- (a) Section 5.18(a) of the Parent Disclosure Schedule contains an accurate and complete list of each "employee benefit plan" (as defined in Section 3(3) of ERISA) and each retirement, pension, welfare, bonus, incentive, commission, stock option or other equity or equity-based, deferred compensation, employment severance, retention, change of control, paid time off, fringe or other benefit or compensation plan, program, policy, agreement, arrangement or contract (including employment, consulting and collective bargaining agreements providing such benefits) that is maintained, sponsored, contributed or required to be contributed to by the Parent or with respect to which the Parent has or could have any Liability (collectively, the "Parent Plans" and each, a "Parent Plan"). Section 5.18(a) of the Parent Disclosure Schedule

also contains an accurate and complete schedule of all Transaction Bonus Amounts, including the amounts due and the payees thereof, and any material conditions thereto.

- Each Parent Plan has in the past six (6) years been maintained, funded and administered, in all material respects, in compliance with its terms and with the applicable requirements of ERISA, the Code and other applicable Law. With respect to each Parent Plan, all contributions (including all employer contributions and employee salary reduction contributions), distributions, reimbursements and premium payments that are due have been timely made and all contributions, distributions, reimbursements and premium payments for any period ending on or before the Closing Date that are not yet due have been made or properly accrued. With respect to each Parent Plan and to the extent applicable, the Parent has delivered to the Company true, correct and complete copies of the current plan documents and summary plan descriptions, the most recent determination letter received from the Internal Revenue Service, the most recent financial statements and actuarial or other valuation reports prepared with respect thereto, the two most recent IRS Form 5500 annual reports (with applicable attachments) as filed, all current related trust agreements, insurance contracts, and other funding arrangements, and any non-routine correspondence with any Governmental Authority with respect to any Parent Plan within the past six (6) years. No Parent Plan is, and neither the Parent nor any ERISA Affiliate maintains, sponsors, contributes to, has any obligation to contribute to, or has or could have any Liability under or with respect to any plan that is or was: (i) a "defined benefit plan" (as defined in Section 3(35) of ERISA); (ii) subject to Section 412 of the Code or Title IV of ERISA; (iii) a "multiemployer plan" (as defined in Section 3(37) of ERISA); or (iv) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA. Parent has no Liability by reason of at any time being treated as a single employer under Section 414 of the Code with any other Person. No Parent Plan provides for, and the Parent and its Subsidiaries do not have any Liability with respect to the provision of post-retirement or post-termination medical, health, or life insurance or other welfare-type benefits for any Person (other than in accordance with Part 6 of Subtitle B of Title I of ERISA or Section 4980B of the Code or comparable state Law ("COBRA") or benefits through the end of the month of termination of employment. No Action with respect to any Parent Plan (other than routine claims for benefits) is pending or threatened, and to the Knowledge of the Parent, there is no fact or circumstance that could give rise to any such Action. There have been no "prohibited transactions" within the meaning of Section 4975 of the Code or Sections 406 or 407 of ERISA and not otherwise exempt under Section 408 of ERISA, and, to the Knowledge of the Parent, no breaches of fiduciary duty (as determined under ERISA), in each case, with respect to any Parent Plan.
- (c) The Parent has complied and complies with the requirements of COBRA. The Parent has not incurred, nor is reasonably expected to incur or to be subject to, any Tax, penalty or other Liability that may be imposed under Sections 4980D or 4980H of the Code.
- (d) Except as set forth in Section 5.18(d) of the Parent Disclosure Schedule, the Transactions alone, or in a combination with a termination of any current or former directors, officers, employees or other service providers of the Parent, or any other event, will not (i) entitle any current or former director, officer employee or service provider of the Parent to severance pay, bonus or any other compensation, benefit or payment; (ii) accelerate the time of payment or vesting (except in connection with Parent Stock Options as set forth in and contemplated by this Agreement), or trigger any payment or funding, of any compensation or benefits for any current

or former director, officer, employee or service provider of the Parent or trigger any other obligation under any Parent Plan; (iii) result in any breach or violation of or default under or limit any right to amend, modify or terminate any Parent Plan; or (iv) result in the payment of any "excess parachute payment" (as defined in Section 280G(b)(1) of the Code).

- (e) No current or former director, officer, employee or service provider of the Parent is entitled to receive any gross-up or additional payment in connection with the Tax required by Section 409A or Section 4999 of the Code. No material adverse Tax penalties, additional taxes, excise taxes or any acceleration of taxes as a result of Sections 409A or 457A of the Code would be reasonably expected to occur for any Person (or to result in any claim against the Parent) with respect to any Parent Plans that are "nonqualified deferred compensation plans" within the meaning of either such Code Section (or that would be such "nonqualified deferred compensation plans" but for an exemption thereunder).
- (f) No Parent Plan is governed by the laws of any jurisdiction other than the United States or provides compensation or benefits to any employee or former employee of the Parent (or any dependent thereof) who resides outside of the United States.
- The Parent and its Affiliates are not party to any collective bargaining agreements with respect to their employees with any labor organization, group or association, nor has voluntarily recognized, is negotiating a collective bargaining agreement with or has agreed to negotiate a collective bargaining agreement with any labor organization, group or association with respect to such employees. There are no efforts by organized labor or its representatives pending or, to the Parent's Knowledge, threatened against the Parent or its Affiliates to unionize any of their employees. Since January 1, 2018, there have been and there are no (i) strikes, work stoppages, material work slowdowns or lockouts in effect or pending or, to the Parent's Knowledge, threatened against or involving the Parent or any of its Affiliates or (ii) material unfair labor practice charges pending before the National Labor Relations Board, grievances, complaints, claims or judicial or administrative proceedings (including any in which a former employee seeks or might be awarded reinstatement to employment), in each case, which are material and are pending or, to the Parent's Knowledge, threatened by or on behalf of any director, officer, employee or other service provider of the Parent.
- (h) The Parent is in compliance in all material respects with all applicable Laws with respect to labor relations, employment and employment practices, occupational safety and health standards, terms and conditions of employment, payment of wages, classification of employees, immigration, visa, work status, human rights, pay equity and workers' compensation, and is not engaged in any unfair labor practices. There is no unfair labor practice charge or complaint against Parent pending or, to the Knowledge of Parent, threatened before the National Labor Relations Board or any comparable Governmental Authority. The Parent has no liability or obligations, including under or on account of a Parent Plan, arising out of the hiring of Persons to provide services to the Company and treating such Persons as consultants or independent contractors and not as employees of the Parent. Except for the Persons listed in Section 5.18(h) of the Parent Disclosure Schedule, there are no independent contractors of the Parent, and all Persons so listed qualify as independent contractors under applicable Law. Except for the Persons listed in Section 5.18(h) of the Parent Disclosure Schedule, there are no "leased employees" (as such term is defined in Section 414(n) of the Code) of the Parent.

- (i) Except as set forth on the Parent Disclosure Schedule, no Parent Plan is intended to be qualified under Section 401(a) of the Code. Every such Parent Plan set forth on the Parent Disclosure Schedule and identified thereon as being intended to be qualified under Section 401(a) of the Code is so qualified under the Code, notwithstanding the lack of any determination, opinion or advisory letter as to its qualified status thereunder.
- (g) With respect to the ESOP and without limiting the other provisions of this Section 5.18(i): (i) all "employer securities" (as defined in Section 407(d)(1) of ERISA) at any time held by the ESOP have at all times been "employer securities" as defined in Section 409(l) of the Code and "qualifying employer securities" as defined in Section 4975(e)(8) of the Code and Section 407(d)(5) of ERISA; (ii) the terms, provisions, use of the proceeds and repayment of any loan to the ESOP (an "ESOP Loan") satisfied in all respects the applicable requirements for an "exempt loan" within the meaning of Section 4975(d) of the Code and the regulations thereunder or Prohibited Transaction Exemption 80-26; (iii) no event of default has occurred or presently exists with respect to any ESOP Loan; (iv) all such loans have been fully satisfied and there are no outstanding amounts due by the ESOP with respect to any ESOP Loan; (v) the ESOP has at all times been maintained in form and in operation in compliance in all material respects with Section 401(a) of the Code; and (vi) any transaction to which the ESOP was at any time a party involving the purchase, sale or exchange of any employer security complied in all material respects with the applicable requirements of ERISA and the Code.
- 5.19. Permits. All material Permits required for the Parent to conduct its business have been obtained by it and are valid and in full force and effect, except as would not have a Parent Material Adverse Effect. All fees and charges with respect to such Permits as of the date hereof have been paid in full. Section 5.19 of the Parent Disclosure Schedule lists all current material Permits issued to the Parent. No event has occurred that, with or without notice or lapse of time or both, would reasonably be expected to result in the revocation, suspension, lapse or limitation of any Permit set forth in Section 5.19 of the Parent Disclosure Schedule.
- 5.20. Environmental Matters. (a) The Parent is and, since January 1, 2019, has been in compliance in all material respects with all Environmental, Health and Safety Laws, which compliance includes obtaining, maintaining and complying with all Permits required or issued under Environmental, Health and Safety Laws to conduct its business and own or operate its assets; (b) the Parent has not received any notice, report, order, or other information, or been party to any Action, alleging that they are in material violation of, or have any material Liability under, any Environmental, Health and Safety Law; (c) the Parent has not generated, transported, treated, stored, disposed of, manufactured, distributed, used, handled, arranged for or permitted the disposal of, released, or exposed any Person to, any Hazardous Material, or owned or operated any property or facility which is or has been contaminated by any Hazardous Material, in each case so as to give rise to any material Liabilities under any Environmental, Health and Safety Laws; (d) the Parent has not assumed, undertaken, provided an indemnity with respect to, or otherwise become subject to, any material Liability of any other Person relating to Environmental, Health and Safety Laws or Hazardous Materials; (e) neither this Agreement nor the consummation of the Transactions will give rise to any obligations for site investigation or cleanup, or notification to or consent of government agencies or third parties, pursuant to any of the so-called "transaction-triggered" or "responsible property transfer" Environmental, Health and Safety Laws; and (f) the Parent has provided to the Company all material environmental

documents in the possession or control of the Parent relating to the Parent and the conduct of its business.

- 5.21. Insurance. Except for insurance policies related to the Parent Plans, Section 5.21 of the Parent Disclosure Schedule lists and briefly describes each insurance policy maintained by the Parent or to which the Parent is a party, an insured, or a beneficiary, or under which the Parent, or any director, officer, or manager of the Parent, in his or her capacity as such, is or has been covered at any time within the twelve months preceding this Agreement (the "Parent Insurance Policies"). Since January 1, 2021, (a) the Parent has had no claim which could be expected to cause a material increase in the rates of insurance for the Parent Insurance Policies and (b) the Parent has not received notice that any insurer under any Parent Insurance Policy is denying liability with respect to a claim thereunder or defending under a reservation of rights clause. The Parent has made available to the Company copies of all Parent Insurance Policies.
- 5.22. Affiliate Interests. Neither the Parent, nor any of its employees, directors, or officers, nor any of their respective Affiliates has any interest in any asset owned or used by the Parent in the operation of its business except as set forth Section 5.22 of the Parent Disclosure Schedule. Except as set forth on Section 5.22 of the Parent Disclosure Schedule, no director or officer or, to the Knowledge of the Parent, any employee of the Parent, or any Affiliate of any of the foregoing is an Affiliate of or the owner (of record or beneficially) of any Equity Interests or any other financial or profit interest in, a Person that has (a) had business dealings or a material financial interest in any transaction with the Parent, or (b) engaged in competition with the Parent, in either case other than: (x) ownership of less than two (2) percent of the outstanding capital stock of a Person that is listed on any national or regional securities exchange, and (y) for compensation for services actually performed as an employee or independent contractor of the Parent.
- 5.23. Access to Information; Disclaimer. The Parent acknowledges and agrees that it (a) has had an opportunity to discuss the business of Company and its Subsidiaries with the management of Company, (b) has been afforded the opportunity to ask questions of and receive answers from officers of Company and (c) has conducted its own independent investigation of Company and its Subsidiaries, their respective businesses and the transactions contemplated hereby, and has not relied on any representation, warranty or other statement by any Person on behalf of Company or any of its Subsidiaries or otherwise, other than the representations and warranties of Company expressly contained in Article III and Article IV of this Agreement and that all other representations and warranties are specifically disclaimed. Without limiting the foregoing, the Parent further acknowledges and agrees that none of Company or any of its shareholders, directors, officers, employees, Affiliates, advisors, agents or other representatives has made any representation or warranty concerning any estimates, projections, forecasts, business plans or other forward-looking information regarding Company, its Subsidiaries or their respective businesses and operations. The Parent hereby acknowledges that there are uncertainties inherent in attempting to develop such estimates, projections, forecasts, business plans and other forward-looking information with which the Parent is familiar, that the Parent is taking full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections, forecasts, business plans and other forward-looking information furnished to them (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, business plans and other forward-looking information, and that the Parent will have

no claim, and hereby releases all claims, against the Company or any of its current or former shareholders, directors, officers, employees, Affiliates, advisors, agents or other representatives with respect thereto.

ARTICLE VI CONDUCT OF BUSINESS

- 6.1. Pre-Closing Covenants of Company. Except (x) in connection with (i) the PIPE Investment, (ii) sales or issuances by the Company of up to \$10.0 million of equity capital, (iii) the incurrence of up to \$10.0 million of indebtedness with CELLCO PARTNERSHIP (d/b/a Verizon Wireless) or (iv) purchases or contracts to purchase other companies or assets in the telecommunications, MVNO or mobile phone or network business, (y) as otherwise provided in this Agreement or consented to in writing by the Parent (such consent not to be unreasonably withheld, conditioned or delayed) and (z) as set forth in Section 6.1 of the Company Disclosure Schedule, from the date hereof until the Closing, the Company shall:
 - (a) comply in all material respects with all applicable Laws;
- (b) not take, or agree to commit to take, any action that would or is reasonably likely to result in any of the conditions to the Merger not being satisfied, or that would impair the ability of any of the Parties to consummate the Merger in accordance with the terms hereof or delay such consummation; and
 - (c) not take or permit any action that would cause any of the changes, events, or conditions described in Section 4.1 to occur.
- 6.2. Pre-Closing Covenants of Parent. Except (x) in connection with the PIPE Investment and the Reincorporation and Charter Amendment, (y) as otherwise provided in this Agreement or consented to in writing by the Company (such consent not to be unreasonably withheld, conditioned or delayed) and (z) as set forth in Section 6.2 of the Parent Disclosure Schedule, from the date hereof until the Closing, the Parent (i) shall conduct the business of the Parent only in the ordinary course of business consistent with past practice, and (ii) shall:
- (a) not (i) declare, set aside or pay any dividends on, or make any other distributions in respect of, any of its Parent Common Stock or any other Equity Interests to any shareholder of the Parent, (ii) split, combine or reclassify any of its Parent Common Stock or any other Equity Interests or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its Parent Common Stock or any other Equity Interests, except for issuances of capital stock or other Equity Interests upon the exercise of options outstanding as of the date hereof in accordance with their present terms, (iii) except as expressly contemplated by this Agreement, purchase, redeem or otherwise acquire any shares of Parent Common Stock or any other securities thereof or any rights, warrants or options to acquire any such shares or other securities, or (iv) make any other actual, constructive or deemed distribution in respect of any shares of Parent Common Stock or other Equity Interests or otherwise make any payments to shareholders of the Parent in their capacity as such;

- (b) not issue, deliver, sell, pledge or otherwise encumber or subject to any Lien any Parent Common Stock, any other voting securities or other Equity Interests or any securities convertible into, or any rights, warrants or options to acquire, any such Parent Common Stock, voting securities or other Equity Interests or convertible securities;
 - (c) not amend its Organizational Documents;
- (d) not acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the stock or assets of any business or any Person;
- (e) not incur any Indebtedness for borrowed money, enter any capital leases, or issue any debt securities, make any loans, advances (other than routine travel and business expense advances) or capital contributions to, or investments in, any Person;
- (f) not assume, guarantee or endorse the Indebtedness of any other Person, or take any action that would reasonably be expected to cause the Parent to incur any liability to, or assume any liability of the ESOP or its participants;
 - (g) preserve and maintain all of its Permits;
 - (h) pay its debts, Taxes and other obligations when due;
- (i) maintain the properties and assets owned, operated or used by the Parent in the same condition as they were on the date of this Agreement, subject to reasonable wear and tear;
 - (j) continue in full force and effect without modification all insurance policies;
- (k) perform all of its material obligations under all Parent Material Contracts relating to or affecting its properties, assets or business:
 - (1) maintain its books and records in accordance with past practice;
 - (m) comply with all applicable Laws;
- (n) not make or rescind any material election relating to Taxes, settle any claim, action, suit, litigation, proceeding, arbitration, investigation, audit or controversy relating to Taxes, file any amended Tax Return or claim for refund, and not make any material change in its accounting or Tax policies or procedures, in each case except as required by applicable Law or in compliance with GAAP;
- (o) except as required to comply with applicable Law or any Parent Plan as in effect on the date of this Agreement or as expressly required pursuant to this Agreement and except as set forth on the Parent Disclosure Schedule, not (i) grant any loan or any increase in the compensation or benefits of, or pay any bonus to, any director, officer, employee or service provider of the Parent other than such increases that may be made in the ordinary course of business consistent with past practice; (ii) pay any compensation or benefit not provided for under any Parent Plan, other than the payment of cash compensation in the ordinary course of

business consistent with past practice; (iii) grant any increase in severance, change of control, retention, termination or similar compensation or benefits to any director, officer, employee or service provider of the Parent, except for Parent Transaction Bonuses; (iv) adopt, establish, enter into, amend or modify, or agree to establish, amend or modify (or announce an intention to establish, amend or modify), or terminate any Parent Plan, or collective bargaining agreement or other labor union contract; (v) grant any awards under any Parent Plan (including the grant of stock options, restricted stock units, stock appreciation rights, performance units, restricted stock or other stock based or stock related awards), or remove existing restrictions in any Parent Plan or awards made thereunder, or add new rights thereto (including the right to net withholding of applicable Taxes); (vi) enter into any trust, annuity or insurance Contract or similar agreement or take any other action to fund or otherwise secure the payment of any compensation or benefit; or (vii) take any action to accelerate the time of vesting or payment of any compensation or benefit;

- (p) not take, or agree to commit to take, any action that would or is reasonably likely to result in any of the conditions to the Merger not being satisfied, or that would impair the ability of any of the Parties to consummate the Merger in accordance with the terms hereof or delay such consummation, or that would reasonably be expected to be inconsistent with the tax treatment contemplated by Section 7.14(a); and
 - (q) not take or permit any action that would cause any of the changes, events, or conditions described in Section 5.8 to occur.
- 6.3. Confidentiality. The Parties acknowledge that the Parent and the Company have previously executed a confidentiality agreement, effective as of June 18, 2021 (the "Confidentiality Agreement"), which Confidentiality Agreement shall continue in full force and effect in accordance with its terms, except as expressly modified by this Agreement.
- 6.4. No Control. Nothing contained in this Agreement shall give the Parent or the Company, directly or indirectly, the right to control or direct the other Party's operations prior to the Effective Time. Prior to the Effective Time, each Party will exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and, if applicable, its Subsidiaries' respective operations.

ARTICLE VII ADDITIONAL COVENANTS

7.1. No Solicitation.

- (a) No Solicitation or Negotiation. Except as set forth in this Section 7.1, from the date hereof until the Effective Time, the Parent shall not, the Parent shall cause its directors, officers, members, employees, agents, attorneys, consultants, contractors, accountants, financial advisors and other authorized representatives ("Representatives") not to, directly or indirectly:
 - (i) solicit, seek or initiate or knowingly take any action to facilitate or encourage any offers, inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal, or

engage, participate in, or knowingly facilitate, any discussions or negotiations regarding, or furnish any nonpublic information to any person in connection with any inquiries, proposals or offers that constitute or could reasonably be expected to lead to, an Acquisition Proposal or the making thereof to the Parent or its shareholders;

- (ii) solicit, initiate, encourage, enter into, continue or otherwise participate or engage in any discussions or negotiations regarding any Acquisition Proposal, or furnish to any person any non-public information or afford any person other than the Parent or the Company, as applicable, access to such party's property, books or records (except pursuant to a request by a Governmental Authority) in connection with any offers, inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal or the making thereof to the Parent or its shareholders;
- (iii) take any action to render any provision of any "fair price," "moratorium," "control share acquisition," "business combination" or other similar and applicable anti-takeover statute or any restrictive provision of any applicable anti-takeover provision in the Parent's organizational documents, in each case inapplicable to any person (other than Company or any of their affiliates) or any Acquisition Proposal (and to the extent permitted thereunder, the Parent shall promptly take all steps necessary to terminate any waiver that may have been heretofore granted to any such person or Acquisition Proposal under any such provisions); or
 - (iv) resolve or publicly propose to do any of the foregoing described in clauses (i) through (iii).

Notwithstanding the foregoing or anything to the contrary set forth in this Agreement, subject to compliance with Section 7.1(c), prior to receipt of the Parent Shareholder Approval, the Parent may (A) furnish non-public information with respect to the Parent to any Qualified Person (and the Representatives of such Qualified Person), or (B) engage in discussions or negotiations (including solicitation of revised Acquisition Proposals) with any Qualified Person (and the Representatives of such Qualified Person) regarding any such bona fide, written Acquisition Proposal that the Parent Board has concluded in good faith, after consultation with outside legal counsel and its financial advisors, constitutes or would reasonably be expected to lead to a Superior Proposal from the Person or group submitting such bona fide, written Acquisition Proposal; but only if and to the extent that in connection with the foregoing clauses (A) and (B). (w) the Parent receives from the Qualified Person an executed confidentiality agreement on terms not less favorable to the Parent than the Confidentiality Agreement and containing additional provisions that expressly permit such party to comply with the terms of this Section 7.1 (a copy of which shall be provided to the Company within 24 hours), (x) the Parent has not otherwise breached this Section 7.1 with respect to such Acquisition Proposal or the person making such Acquisition Proposal, (y) prior to furnishing any such nonpublic information to, or entering into discussions with, such Person, and (z) the Parent Board has determined that taking such actions would be required to prevent a breach of its fiduciary duties under applicable Law. The Parent shall notify the Company prior to furnishing any information or entering into any

discussions or negotiations as provided in this Section 7.1(a). It is understood and agreed that any violation of the restrictions in this Section 7.1 (or action that, if taken by the Company or the Parent, as applicable, would constitute such a violation) by any Representatives of the Company or the Parent shall be deemed to be a breach of this Section 7.1 by the Company or the Parent.

- (b) No Change in Recommendation or Alternative Acquisition Agreement.
- (i) Prior to the Effective Time, the Parent Board shall not, except as set forth in this Section 7.1, (A) withhold, withdraw, materially qualify or modify, or publicly propose to withhold, withdraw, materially qualify or modify, the approval or recommendation by the Parent Board or committee thereof with respect to the Merger or the Parent Stock Issuance pursuant to this Agreement, (B) fail to include the Parent Board Recommendation in the Proxy Statement/Consent/Prospectus; (C) adopt, endorse, approve or recommend, or propose publicly to adopt, endorse, approve or recommend, or submit to the vote of any securityholders of Parent, any Acquisition Proposal, or approve any transaction under, or any transaction resulting in any third party becoming an "interested stockholder" under, Section 302A.673 of the MBCA, or authorize, resolve or agree to take any such action; (D) fail to recommend against acceptance of any Acquisition Proposal that is a tender offer or exchange offer, and to expressly reaffirm the Parent Board Recommendation, within five (5) Business Days after commencement (or if the Parent Shareholder Meeting is to occur prior to the fifth Business Day, such period ending on the later of (x) the Business Day prior to the Parent Shareholder Meeting and (y) the second Business Day after such request); or (E) following the public announcement of an Acquisition Proposal (or the public announcement of an intention to make an Acquisition Proposal) other than the commencement of a tender or exchange offer contemplated by clause (D), fail to issue a press release expressly reaffirming the Parent Board Recommendation within five (5) Business Days after the Company requests a reaffirmation thereof (or if the Parent Shareholder Meeting is to occur prior to the fifth Business Day, such period ending on the later of (x) the Business Day prior to the Parent Shareholder Meeting and (y) the second Business Day after such request) (any of the foregoing in clauses (A) –(E), a "Parent Board Recommendation Change");
- (ii) Prior to receipt of the Parent Shareholder Vote the Parent shall not enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement, option agreement, joint venture agreement or similar agreement (an "Alternative Acquisition Agreement") providing for the consummation of a transaction contemplated by any Acquisition Proposal (other than a confidentiality agreement referred to in Section 7.1(a) entered into in the circumstances referred to in Section 7.1(a); and
- (iii) The Parent Board, and each committee thereof, shall not, except as set forth in this <u>Section 7.1</u>, adopt, approve or recommend, or publicly propose to adopt, approve or recommend, any Acquisition Proposal or Alternative Acquisition Agreement.

Notwithstanding the foregoing or anything to the contrary set forth in this Agreement (including the provisions of this Section 7.1), if at any time after the date of this Agreement and prior to receipt of the Parent Shareholder Vote, Parent receives an unsolicited, bona fide, written Acquisition Proposal that the Parent Board has determined in good faith, after consultation with outside legal counsel and its financial advisors, is a Superior Proposal (after giving effect to all of the revisions to the terms of this Agreement which may be offered by the Company, including pursuant to this paragraph), then the Parent Board prior to the time of, but not after, receipt of the Parent Shareholder Vote (x) may make a Parent Board Recommendation Change or (y) may terminate this Agreement in accordance with Section 9.1(h) and concurrently with such termination enter into a definitive agreement providing for the implementation of such Superior Proposal, in each case if at any time prior to the time, but not after, receipt of the Parent Shareholder Vote is obtained the Parent Board has determined in good faith, after consultation with outside legal counsel and its financial advisors, that the failure to make such Parent Board Recommendation Change or the failure to terminate this Agreement would be inconsistent with its fiduciary obligations under applicable Law; provided, however, that Parent shall not terminate this Agreement pursuant to the foregoing clause (y), and any purported termination pursuant to the foregoing clause (y) shall be void and of no force or effect, unless concurrently with such termination the Parent pays the Parent Termination Fee and otherwise complies with the provisions of Section 9.3(b) and Section 9.3(c); and provided, further, that (1) the Parent Board may not effect a Parent Board Recommendation Change pursuant to the foregoing clause (x) or terminate this Agreement pursuant to the foregoing clause (y) unless such Acquisition Proposal (and any precursor thereto) did not result from or arise in connection with a breach by Parent, or any of its Representatives (as if they were bound hereby) of its obligations under this Section 7.1; (2) the Parent has provided at least four (4) Business Days' prior written notice to the Company that it intends to effect a Parent Board Recommendation Change, including a description in reasonable detail of the reasons for such recommendation change, and written, true, complete and unredacted copies of any relevant proposed transactions agreements with any party making a potential Superior Proposal (including the identity of the person making such Superior Proposal) (a "Recommendation Change Notice") (it being understood that the Recommendation Change Notice shall not constitute a Parent Board Recommendation Change for purposes of this Agreement); (3) such party has complied in all material respects with the requirements of this Section 7.1 in connection with any potential Superior Proposal; (4) during each such four Business Day period, if requested by Company, engaged in good faith negotiations with Company to amend this Agreement in such a manner that the Acquisition Proposal that was determined to constitute a Superior Proposal no longer is a Superior Proposal; and (5) at the end of such four Business Day period, if such Acquisition Proposal has not been withdrawn, the Parent Board shall have determined in good faith (after consultation with outside legal counsel and its financial advisors), that the Acquisition Proposal is still a Superior Proposal and a Parent Board Recommendation Change would still be required to comply with its fiduciary obligations under applicable Law (after taking into account any changes to the terms of this Agreement agreed to in writing by Company in response to any notice required by clause (2), as a result of the negotiations required by clause (4) or otherwise). In the event of any material amendment to any Superior Proposal (including any revision in the amount, form or mix of consideration the Parent's shareholders would receive as a result of such potential Superior Proposal), the Parent shall be required to provide the Company with notice of such material amendment and there shall be a new four Business Day period following such notification during which the Parent

shall comply again with the requirements of this Section 7.1(b) and the Parent Board shall not make a Parent Board Recommendation Change prior to the end of any such period as so extended.

- Notices of Proposals. Each Party will as promptly as reasonably practicable (and in any event within twenty-four (24) hours after receipt) (i) notify the other Party of its receipt of any Acquisition Proposal and (ii) provide to the other Party a copy of such Acquisition Proposal (if written), or a summary of the material terms and conditions of such Acquisition Proposal (if oral), including the identity of the person making such Acquisition Proposal, and copies of all written communications with such person with respect to such actual or potential Acquisition Proposal. The Parent shall notify the Company, in writing, of any decision of the Parent Board as to whether to consider any Acquisition Proposal from a Qualified Person or to enter into discussions or negotiations concerning any Acquisition Proposal from a Qualified Person or to provide non-public information with respect to such to any such Qualified Person, which notice shall be given as promptly as practicable after such determination was reached (and in any event no later than twenty-four (24) hours after such determination was reached). The Parent will (A) provide the Company with written notice setting forth such information as is reasonably necessary to keep the Company informed of the material terms of any such Acquisition Proposal from a Qualified Person and of any material amendments or modifications thereto, (B) keep the Company fully informed as promptly as practicable with respect to any changes to the material terms of an Acquisition Proposal by a Qualified Person submitted to such party (and in any event within twentyfour (24) hours following any such changes), including by providing a copy of all written proposals and a summary of all oral proposals or material oral modifications to an earlier written proposal, in each case relating to any Acquisition Proposal by a Qualified Person, (C) prior to, or substantially concurrently with, the provision of any non-public information of the Parent to any such Qualified Person, provide such information to the Company (including by posting such information to an electronic data room), to the extent such information has not previously been made available to the Company, and (D) promptly (and in any event within twenty-four (24) hours of such determination) notify the Company of any determination by the Parent Board that such Acquisition Proposal constitutes a Superior Proposal.
- (d) Certain Permitted Disclosure. Nothing contained in this Agreement shall prohibit the Parent or the Parent Board from complying with Rules 14d-9 and 14e-2(a) promulgated under the Exchange Act; provided, however, that any disclosure made by the Parent or the Parent Board pursuant to Rules 14d-9 and 14e-2(a) shall be limited to a statement that Parent is unable to take a position with respect to the bidder's tender offer unless the Parent Board determines in good faith, after consultation with its outside legal counsel, that such statement would be a breach of its fiduciary duties under applicable Law; provided, further, that any such disclosures (other than a "stop, look and listen" communication or similar communication of the type contemplated by Section 14d-9(f) under the Exchange Act) shall be deemed to be a Parent Board Recommendation Change, unless the Parent Board expressly publicly reaffirms its recommendation for the Merger and the other Transactions in such communication.
- (e) <u>Cessation of Ongoing Discussions.</u> Each of the Company and the Parent shall, and shall cause its Representatives to, cease immediately all solicitation, encouragement,

discussions and negotiations that commenced prior to the date of this Agreement regarding any proposal that constitutes, or could reasonably be expected to lead to, an Acquisition Proposal; provided, however, that the foregoing shall not in any way limit or modify the rights of a Party hereto under the other provisions of this Section 7.1. Each of the Company and the Parent will each immediately revoke or withdraw access of any Person (other than a Party and their respective Representatives and other than any Person who has not made, and would not be reasonably expected to make, any Acquisition Proposal) to any data room (virtual or actual) containing any non-public information with respect to such Party and request from each third party (other than the Company, Parent and their Representatives and other than any Person who has not made, and would not be reasonably expected to make, any Acquisition Proposal) the prompt return or destruction of all non-public information with respect to the Company or the Parent, as applicable, previously provided to such person. The Parent shall not, and shall cause its affiliates and all of its and their respective Representatives not to, release any person from, or waive, amend or modify any provision of, or grant any permission under any "standstill" provision or similar provision with respect to any capital stock of the Parent in any agreement to which the Parent or any of its affiliates is a party; provided that the Parent Board shall (after consultation with its outside counsel) that the failure to take such action would violate the directors' fiduciary duties under applicable Law and (y) any such action by the Parent Board does not violate any other provision of this Section 7.1. Except to the extent otherwise permitted by the foregoing sentence, the Parent shall, and shall cause its affiliates to, promptly take all reasonable steps within their power to terminate any waiver under any such provisions that may have been heretofore granted to any person other than the Comp

7.2. Access to Information; Cooperation; Financial Statements.

- Agreement, each of the Parent and the Company shall (and shall cause each of its Subsidiaries. if any to) afford to the other Party's officers, employees, accountants, counsel and other representatives, reasonable access, during normal business hours during the period prior to the Effective Time, to all its properties, books, contracts, commitments, personnel and records and, during such period, each of the Parent and the Company shall (and shall cause each of its Subsidiaries, if any to) furnish promptly to the other Party all information concerning its business, properties, assets and personnel as the other Party may reasonably request. Each of the Parent and the Company will hold any such information which is nonpublic in confidence in accordance with the Confidentiality Agreement. No information or knowledge obtained in any investigation pursuant to this Section 7.2 or otherwise shall affect or be deemed to modify any representation or warranty contained in this Agreement or the conditions to the obligations of the Parties to consummate the Merger.
- (b) As promptly as reasonably practicable, the Company shall deliver to the Parent (i) the audited balance sheets of the Company as of December 31, 2021 and the related audited statements of operations, changes in shareholders' equity and cash flows of the Company for each of the periods then ended, audited in accordance with the standards of the Public Company Accounting Oversight Board ("PCAOB") and containing an unqualified report of the Company's auditors (the "Closing Company Audited Financial Statements") and (ii) the

unaudited balance sheet and the related statements of operations, changes in shareholders' equity and cash flows of the Company as of and for a year-to-date period ended as of the end of such fiscal quarter that is required to be included in the Registration Statement or Proxy
Statement/Consent/Prospectus and any other filings to be made by the Parent with the SEC in connection with the Merger. All such financial statements, together with any unaudited consolidated balance sheet and the related statements of operations, changes in shareholders' equity and cash flows as of and for a year-to-date period ended as of the end of such fiscal quarter that is required to be included in the Registration Statement or Proxy
Statement/Consent/Prospectus and any other filings to be made by the Parent with the SEC in connection with the Merger, (A) will be prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except as may be indicated in the notes thereto), (B) will fairly present, in all material respects, the financial position, results of operations and cash flows of the Company as of the date thereof and for the period indicated therein, except as otherwise specifically noted therein, and (C) will, in the case of the Closing Company Audited Financial Statements, have been audited in accordance with the standards of the PCAOB.

- (c) The auditor engaged to audit the Closing Company Audited Financial Statements and to review the unaudited financial statements will be an independent registered public accounting firm with respect to the Company within the meaning of the Exchange Act and the applicable rules and regulations thereunder adopted by the SEC and the PCAOB.
- (d) The Company shall use commercially reasonable efforts (i) to assist, upon advance written notice, the Parent in causing to be prepared in a timely manner any other financial information or statements (including customary pro forma financial statements) that are required to be included in the Registration Statement or Proxy Statement/Consent/Prospectus and any other filings to be made by the Parent with the SEC in connection with the Merger and (ii) to obtain the consents of its auditors with respect thereto as may be required by applicable Law or requested by the SEC.
 - 7.3. Proxy Statement/Consent/Prospectus; Registration Statement; Written Consent.
- (a) In connection with the Parent Shareholders Meeting, as soon as practicable following the date of this Agreement, the Parent and the Company shall jointly prepare, and the Parent shall file with the SEC the Registration Statement (which shall include the Proxy Statement/Consent/Prospectus, unless either the Parent or the Company reasonably requests separately filing the proxy statement and consent solicitation components outside of the Registration Statement in response to SEC comments or guidance), and each of Parent and the Company shall promptly furnish all information concerning itself and its Affiliates as may reasonably be requested by the other Party and shall otherwise reasonably assist and cooperate with the other in connection with the preparation, filing and distribution of the Registration Statement and Proxy Statement/Consent/Prospectus. The Registration Statement shall include for registration all of the shares of Parent Common Stock to be issued under this Agreement as Merger Consideration and, if reasonably requested by either the Parent or the Company in response to SEC comments or guidance, will be a resale registration statement on Form S-3 in lieu of Form S-4. The Company and the Parent shall each use their respective reasonable best efforts to: (i) cause the Registration Statement to be declared effective under the Securities Act

as promptly as practicable after its filing; (ii) ensure that the Registration Statement complies in all material respects with the applicable provisions of the Securities Act and the Exchange Act; and (iii) keep the Registration Statement effective for so long as necessary to complete the Merger and Parent Stock Issuance. The Parent shall notify the Company promptly upon receipt of any comments from the SEC or any request from the SEC for amendments or supplements to the Registration Statement and will, as promptly as practicable after receipt thereof, provide the Company with copies of all material correspondence between it and its Representatives, on the one hand, and the SEC, on the other hand, and all written comments with respect to the Registration Statement or the Proxy Statement/Consent/Prospectus received from the SEC and advise the Company on any oral comments with respect to the Registration Statement received from the SEC. The Parent will notify the Company, promptly after the Parent receives notice thereof, of the time when the Registration Statement has become effective or any supplement or amendment to the Registration Statement has been filed, and of the issuance of any stop order or suspension of the qualification of the shares of Parent Common Stock issuable in connection with the Merger for offering or sale in any jurisdiction. The Parent shall: (A) after good faith consultation with the Company, set a record date (the "Record Date") for determining the Parent stockholders entitled to attend the Parent Shareholders Meeting, (B) cause the Proxy Statement/Consent/Prospectus to be mailed to the Parent's shareholders as promptly as practicable after the Registration Statement is declared effective under the Securities Act, and (C) ensure that the Proxy Statement/Consent/Prospectus complies in all material respects with the applicable provisions of the Securities Act and Exchange Act.

- (b) Notwithstanding anything to the contrary stated above, prior to filing and mailing, as applicable, the Registration Statement or Proxy Statement/Consent/Prospectus (or any amendment or supplement thereto) or responding to any comments of the SEC with respect thereto, the Parent shall provide the Company a reasonable opportunity to review and comment on such document or response and shall consider in good faith any such comments proposed by the Company.
- The Parent and the Company shall promptly make all necessary filings with respect to the Merger and the issuance of the Parent Common Stock under the Securities Act, the Exchange Act, applicable state blue sky laws and the rules and regulations thereunder, and the Company shall furnish to the Parent all information concerning the Company as may be reasonably requested in connection with any such actions. The Parent and the Company shall ensure that none of the information supplied by or on its behalf for inclusion or incorporation by reference in (i) the Registration Statement will, at the time the Registration Statement is filed with the SEC, at each time at which it is amended and at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they are made, not misleading or (ii) the Proxy Statement/Consent/Prospectus will, at the date it is first mailed to holders of Parent Common Stock and at the time of the Parent Shareholders Meeting contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. If at any time prior to the Closing any information relating to the Parent and the Company, or any of their respective Affiliates, officers or directors, is discovered by the Parent or the Company which should be set forth in an amendment or supplement to the Registration

Statement, so that any of such documents would not include a misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the Party that discovers such information shall promptly notify the other parties hereto and each of the Parent and the Company shall use its reasonable best efforts to cause an appropriate amendment or supplement describing such information to be promptly filed with the SEC and, to the extent required by law, disseminated to the holders of Parent Common Stock.

7.4. Nasdaq Listing. The Parent shall use its commercially reasonable efforts, (a) to maintain its existing listing on Nasdaq until the Effective Time and to obtain approval of the listing of the combined corporation on Nasdaq, (b) to the extent required by the rules and regulations of Nasdaq, to prepare and submit to Nasdaq a notification form for the listing of the shares of Parent Common Stock to be issued in connection with the transactions contemplated hereby, and to cause such shares to be approved for listing (subject to official notice of issuance) on Nasdaq under a ticker symbol to be selected by the Company; and (c) to the extent required by Nasdaq Marketplace Rule 5110, to file an initial listing application for the Parent Common Stock on Nasdaq (the "Nasdaq Listing Application") and to request that such Nasdaq Listing Application be conditionally approved prior to the Effective Time. The Parties will use commercially reasonable efforts to coordinate with respect to compliance with Nasdaq rules and regulations. Each Party will promptly inform the other Party of all verbal or written communications between Nasdaq and such Party or its representatives. Each of Parent and Merger Sub will use their respective commercially reasonable efforts to (i) cause the Nasdaq Listing Application, when filed, to comply in all material respects with all legal requirements applicable thereto, (ii) respond as promptly as reasonably practicable to and resolve all comments received from the Nasdaq or its staff concerning the Nasdaq Listing Application and (iii) have the Nasdaq Listing Application approved by the Nasdaq as promptly as practicable after such filing. No submission of, or amendment or supplement to, the Nasdaq Listing Application, or response to Nasdaq comments with respect thereto, will be made by Parent, Merger Sub or the Company, as applicable, without the other parties' prior consent (which shall not be unreasonably withheld, conditioned or delayed) and without providing such other parties hereto a reasonable opportunity to review and comment thereon. Parent and the Company agree to evenly split all Nasdaq fees associated with the Nasdaq Listing Application incurred prior to the Closing, if any, as well as any fees, costs and expenses associated with any third-party consultants that may be retained to assist with Nasdaq matters relating to the Transactions (the "Nasdaq Fees"). The Company will cooperate with Parent as reasonably requested by Parent with respect to the Nasdaq Listing Application and promptly furnish to Parent all information concerning the Company and its stockholders that may be required or reasonably requested in connection with any action contemplated by this Section 7.4.

7.5. Shareholder Approval.

(a) Upon the terms set forth in this Agreement, the Company shall seek the written consent, in form and substance reasonably acceptable to Parent, of (i) holders of a majority of the outstanding common stock of the Company in favor of the adoption of this Agreement and the Merger and all other Transactions contemplated by this Agreement and (ii) holders of at least 75% of the outstanding common stock pursuant to Section 2.02 of the Company Stockholders Agreement (the "Company Stockholder Approval") via written consent

Section 9.1.

(the "Written Consent") as soon as reasonably practicable after the Registration Statement becomes effective. In connection therewith, the Company, as promptly as practicable (A) shall establish the record date (which record date shall be mutually agreed with Parent) for determining the Company Stockholders entitled to provide such written consent, (B) shall cause the Consent Solicitation Statement to be disseminated to the Company Stockholders in compliance with applicable Law and (C) shall use commercially reasonable efforts to solicit written consents from the Company Stockholders to give the Company Stockholder Approval. The Company Board shall recommend to the Company Stockholders that they approve and adopt this Agreement and approve the Merger and all other Transactions.

Subject to Section 7.1, the Parent, acting through the Parent Board, shall include all appropriate provisions in the Proxy Statement/Consent/Prospectus and take all action reasonably necessary related thereto (i) in order to obtain the affirmative vote of the holders of a majority of the voting power of all shares of Parent Common Stock in favor of this Agreement, the Merger and the Transactions (the "Parent Shareholder Approval"), (ii) approval of a new equity incentive plan having the terms, conditions and share fund requested by the Company, subject to the approval of Parent (such consent not to be unreasonably withheld, conditioned or delayed) (the "New Equity Plan"), (iii) the adoption and approval of the Reincorporation and Charter Amendment, (iv) to the extent required by the Nasdaq listing rules, the approval of the Parent Share Issuance and the change of control of Parent, (vi) the adjournment of the Parent Shareholder Meeting, if necessary, to permit further solicitation of proxies because there are not sufficient votes to approve and adopt any of the foregoing proposals and (vii) the approval of any other proposals reasonably agreed by Parent and the Company to be necessary or appropriate in connection with the Transactions (but no other proposals other than as provided in this Section 7.5 (b)), at the Parent Shareholder Meeting as soon as practicable after the Registration Statement is declared effective, including giving notice of and convening and holding the Parent Shareholders Meeting, and, in connection therewith, the Parent shall mail the Proxy Statement/Consent/Prospectus to the holders of Parent Common Stock in advance of the Parent Shareholders Meeting. The Parent may only postpone or adjourn the Parent Shareholders Meeting (x) to solicit additional proxies for the purpose of obtaining the Parent Shareholder Approval, (y) for the absence of a quorum and (z) to allow reasonable additional time for the filing or mailing of any supplemental or amended disclosure that the Parent has determined after consultation with outside legal counsel is reasonably likely to be required under applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by holders of Parent Common Stock prior to the Parent Shareholder Meeting. Subject to the rights of the Parent and the Parent Board under Section 7.1(b)(i), the Parent shall use its reasonable best efforts to solicit from its shareholders proxies in favor of approval of this Agreement and the Transactions, and to secure approval of the other proposals described in this Section 7.5(b) and to secure any other approval of shareholders of the Parent that is required by applicable Law to effect the Transactions (including authority to adjourn the meeting to solicit additional votes and to obtain a "say-on-pay" or "say-on-parachute" vote). For the avoidance of doubt, the New Equity Plan shall have no impact on the Exchange Ratio but any future grants thereunder after the Closing will dilute all shareholders of the Parent on a pro-rata basis.

(c) Notwithstanding the foregoing, nothing herein shall limit a Party's right to terminate this Agreement pursuant to

7.6. Fulfillment of Conditions.

- (a) Subject to the terms hereof, including Section 7.6, the Company and the Parent shall each use commercially reasonable best efforts (but which shall not require a Party to commit or invest its own capital) to (i) take, or cause to be taken, all actions, and do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper or advisable to procure, negotiate and enter into the Subscription Agreements and consummate and make effective, and to satisfy all conditions to, in the most expeditious manner practicable, the Transactions as promptly as practicable, (ii) as promptly as practicable, obtain from any Governmental Authority or any other third party any consents, licenses, permits, waivers, approvals, authorizations, or orders required to be obtained or made by the Company or the Parent or any of their Subsidiaries, if any, in connection with the authorization, execution and delivery of this Agreement and the consummation of the Transactions, (iii) as promptly as practicable, make all necessary filings, and thereafter make any other required submissions, with respect to this Agreement and the Merger required under (A) the Securities Act and the Exchange Act, and any other applicable federal or state securities Laws, (B) the HSR Act and any other foreign or other antitrust Law, and (C) any other applicable Law, and (iv) execute or deliver any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.
- (b) The Company and the Parent shall reasonably cooperate with each other in connection with the making of all such filings. The Company and the Parent shall use their respective commercially reasonable efforts to furnish to each other all information required for any application or other filing to be made pursuant to the rules and regulations of any applicable Law (including all information required to be included in the Proxy Statement/Consent/Prospectus and the Registration Statement) in connection with the Transactions.
- (c) Each of the Company and the Parent shall give (or shall cause their respective Subsidiaries, if applicable, to give) any notices to third parties, and use, and cause their respective Subsidiaries, if applicable, to use, their reasonable best efforts to obtain any third party consents related to or required in connection with the Merger that are (i) necessary to consummate the Transactions, (ii) disclosed or required to be disclosed in the Company Disclosure Schedule or the Parent Disclosure Schedule, as the case may be, or (iii) required to prevent the occurrence of an event that may have a Company Material Adverse Effect or a Parent Material Adverse Effect from occurring prior to or after the Effective Time.
- 7.7. Public Disclosure. Except as may be required by applicable Law or stock market regulations, (a) the press release announcing the execution of this Agreement shall be issued only in such form as shall be mutually agreed upon by the Parent and the Company and (b) both the Company and the Parent shall use reasonable best efforts to consult with one another before issuing any press release or otherwise making any public statement with respect to the Transactions or this Agreement and shall not issue any such press release or make any such public statement prior to using such efforts (provided, however, that these restrictions shall not apply to any communications by the Parent with respect to any Acquisition Proposal, Superior Proposal, Recommendation Change Notice or Parent Board Recommendation Change).

- 7.8. Notification of Breaches. Each Party agrees that it shall promptly notify in writing the other Party if such notifying Party becomes aware of any breach of, or inaccuracy in, or of any facts or circumstances constituting or resulting in the breach of, or inaccuracy in, any representation, warranty or covenant of such notifying Party or of such other Party that would cause a failure of the conditions to Closing set forth in Article VIII with respect to such Party.
 - 7.9. Succession of Officers and Directors.
- (a) The Parent shall take all action necessary to cause the persons identified on <u>Section 7.9(a)</u> of the Parent Disclosure Schedule to be appointed as executive officers of the Parent as of the Effective Time.
- (b) The Parent shall take all action necessary to (i) obtain the resignations of the directors identified on Section 7.9(i) of the Parent Disclosure Schedule effective as of the Effective Time; and (ii) cause to be appointed to the Parent Board as directors, or continue to serve on the Parent Board as directors, as applicable, the persons whose names are provided to the Parent in writing at least five (5) Business Days prior to the Closing. If any person identified in accordance with this Section 7.9(b) is unable or unwilling to serve in such capacity, the Company shall designate a successor.
- 7.10. State Takeover Laws. If any "fair price," "business combination" or "control share acquisition" statute or other similar statute or regulation is or may become applicable to any of the Transactions, the Parties hereto shall use their respective reasonable best efforts to (a) take such actions as are reasonably necessary so that the transactions contemplated hereunder may be consummated as promptly as practicable on the terms contemplated hereby and (b) otherwise take all such actions as are reasonably necessary to eliminate or minimize the effects of any such statute or regulation on such transactions.
- 7.11. Section 16 Matters. Prior to the Effective Time, the Parent shall take all such steps as may be required to cause to be exempt under Rule 16b-3 promulgated under the Exchange Act any acquisitions of Parent Common Stock (including derivative securities with respect to such shares) that are treated as acquisitions under such rule and result from the transactions contemplated by this Agreement by each individual who may become or is reasonably expected to become subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Parent immediately after the Effective Time.
- 7.12. Cash Dividend. At a time sufficiently in advance of the Effective Time to comply with the requirements of this Agreement and Nasdaq, in consultation with the Company, the Parent Board shall declare and set aside the Cash Dividend, as adjusted pursuant to Section 2.11 of this Agreement, to its shareholders who are holders of record as of the Dividend Record Date, and then duly pay and issue the Cash Dividend to such holders of record as promptly after the Closing as reasonably practicable.
- 7.13. Termination of Certain Agreements and Rights. The Company shall cause the agreements set forth in <u>Section 7.13</u> of the Company Disclosure Schedule, including the Company Stockholders Agreement and any such agreement granting any person investor rights,

rights of first refusal, registration rights or director election rights, to be terminated immediately prior to the Effective Time.

7.14. Certain Tax Matters.

- (a) Tax Treatment. The Parties shall not file any U.S. federal, state or local Tax Return in a manner that is inconsistent with the treatment of the Merger as a "reorganization" within the meaning of Section 368(a) of the Code for U.S. federal, state and other relevant Tax purposes, unless otherwise required by a taxing authority pursuant to an audit defended in good faith. The Parties shall use their respective reasonable best efforts to cause the Merger to qualify, and will not take any action or cause any action to be taken which action would reasonably be expected to prevent the Merger from qualifying, as a reorganization within the meaning of Section 368(a) of the Code.
- (b) Tax Opinions. Each Party shall promptly notify the other Party in writing if, before the Closing Date, such Party knows or has reason to believe that the Merger may not qualify as a "reorganization" within the meaning of Section 368(a) of the Code. In the event the SEC requests or requires a tax opinion with respect to the Merger qualifying as a "reorganization" within the meaning of Section 368(a) of the Code, each Party shall use commercially reasonable efforts to execute and deliver customary tax representation letters to the applicable tax advisors in form and substance reasonably satisfactory to the advisors delivering such opinion and the Party delivering such tax representation letter, and shall use commercially reasonable efforts to cause their respective counsel to deliver such tax opinions as the SEC may requires or requests, in connection with the Proxy Statement/Consent/Prospectus, consistent with customary practice.

7.15. PIPE Investment.

(a) The Company will not terminate or amend the Commitment Letter without Parent's consent, not to be unreasonably withheld, and the Parent shall not terminate or amend its March 11, 2022 engagement letter with B. Riley & Co without the Company's consent, not to be unreasonably withheld. The Company and Parent will each use their commercially reasonable efforts (but which shall not require a Party to commit or invest its own capital), in cooperation with each other and B. Riley & Co., to procure, negotiate and enter into as soon as possible Subscription Agreements from the applicable accredited investors named therein (the "PIPE Investors"), whether pursuant to the Commitment Letter or such other commitments as the Parties may mutually agree upon, reasonably acceptable to each Party, to provide equity financing to the Parent solely for purposes of consummating the Transactions in the aggregate amount of no less than the Minimum PIPE Investment Amount. Parent shall keep the Company promptly advised of all negotiations, drafts and material updates regarding the PIPE Investment and the Subscription Agreement and will permit the Company and its advisors to assist with all such negotiations. When received by Parent, Parent shall provide to Company true, correct and complete copies of each of such Subscription Agreement. Parent shall not enter with any PIPE Investor any other agreements, side letters, or arrangements between the Parent and any PIPE Investor relating to any Subscription Agreement that could affect the obligation of such PIPE Investor to contribute to the Parent the applicable portion of the Minimum PIPE Investment Amount set forth in the Subscription Agreement of such PIPE Investor.

- When the Subscription Agreements have been entered into, unless otherwise approved in writing by the Company, Parent shall not permit any amendment or modification to be made to, any waiver (in whole or in part) or provide consent to (including consent to termination), of any provision or remedy under, or any replacements of, any of the Subscription Agreements (which consent or approval shall not be unreasonably withheld, conditioned or delayed). Parent and Company shall take, or cause to be taken, all actions and do, or cause to be done, all things necessary, proper or advisable to consummate the transactions contemplated by the Subscription Agreements on the terms and conditions described therein, including maintaining in effect the Subscription Agreements and to: (a) satisfy in all material respects on a timely basis all conditions and covenants applicable to the Parent in the Subscription Agreements and otherwise comply with its obligations thereunder; (b) in the event that all conditions in the Subscription Agreements (other than conditions that the Parent or any of its Affiliates control the satisfaction of and other than those conditions that by their nature are to be satisfied at the Closing) have been satisfied, consummate transactions contemplated by the Subscription Agreements at or prior to Closing; (c) confer with the Company regarding timing of the expected Closing Date (as defined in the Subscription Agreements); (d) deliver notices to counterparties to the Subscription Agreements sufficiently in advance of the Closing to cause them to fund their obligations as far in advance of the Closing as permitted by the Subscription Agreements; and (e) without limiting the Company's rights to enforce certain of such Subscription Agreements in the event that all conditions in the Subscription Agreements (other than conditions that the Parent or any of its Affiliates control the satisfaction of and other than those conditions that by their nature are to be satisfied at the Closing) have been satisfied, to cause (including by bringing appropriate and timely proceedings in a court of law to enforce the Parent's rights under the Subscription Agreement) the applicable PIPE Investors to pay to (or as directed by) the Parent the applicable portion of the Minimum PIPE Investment Amount, as applicable, set forth in the Subscription Agreements in accordance with their terms. Without limiting the generality of the foregoing, the Parent shall give the Company, prompt written notice: (a) of any breach or default (or any event or circumstance that, with or without notice, lapse of time or both, could give rise to any breach or default) by any party to any Subscription Agreement known to the Parent; (b) of the receipt of any written notice or other written communication from any party to any Subscription Agreement with respect to any actual, threatened or claimed expiration, lapse, withdrawal, breach, default, termination or repudiation by any party to any Subscription Agreement or any provisions of any Subscription Agreement; and (c) if the Parent does not expect to receive all or any portion of the Minimum PIPE Investment Amount on the terms, in the manner or from the PIPE Investors as contemplated by the Subscription Agreements. The Parent shall deliver all notices it is required to deliver under the Subscription Agreements on a timely basis in order to cause the PIPE Investors to consummate the PIPE Investment concurrently with the Closing and shall take all actions required under any Subscription Agreements with respect to the timely issuance and delivery of any physical certificates evidencing the shares of the Parent Common Stock as and when required under any such Subscription Agreements. The Company shall cooperate with the Parent in the fulfillment of all obligations of the Parent under this Section or under any Subscription Agreement and otherwise necessary or desirable to consummate the PIPE Investment.
- (c) In connection with the PIPE Investment, the Company and its Affiliates shall not make any untrue statement of material fact or any omit to state a fact necessary to

ensure that any statements so made are not misleading to any investor or prospective investor in the PIPE Investment and shall indemnify and hold harmless the Parent and its Affiliates from any loss, liability or expense arising therefrom or otherwise arising from any similar indemnification obligation owed (i) by the Company to any advisor retained by the Company in connection with the PIPE Investment under an engagement agreement or otherwise or (ii) by the Parent under the March 11, 2022 engagement letter with B. Riley & Co. in respect of statements made or information provided by the Company; provided, this indemnity shall not apply to any statements or information (i) the accuracy or non-misleading nature of which the Parent has represented and warranted in this Agreement or (ii) which is provided by Parent in writing specifically to be provided to investors or prospective investors in connection with the PIPE Investment.

7.16. D&O Insurance and Indemnification.

- All rights to indemnification or exculpation now existing in favor of the directors and officers of the Parent or the Company, as provided in the Parent Organizational Documents, the Company Organizational Documents or otherwise in effect as of immediately prior to the Effective Time, in either case, solely with respect to any matters occurring on or prior to the Effective Time shall survive the Transactions and shall continue in full force and effect from and after the Effective Time and the Parent or the Company, as applicable, will perform and discharge, or cause to be performed and discharged, all obligations to provide such indemnity and exculpation. To the maximum extent permitted by applicable Law, the Parent or the Company, as applicable, shall advance, or cause to be advanced, expenses in connection with such indemnification as provided in the Parent Organizational Documents, the Company Organizational Documents or other applicable agreements as in effect immediately prior to the Effective Time. The indemnification and liability limitation or exculpation provisions of the Parent Organizational Documents and the Company Organizational Documents shall not be amended, repealed or otherwise modified after the Effective Time in any manner that would materially and adversely affect the rights thereunder of individuals who, as of immediately prior to the Effective Time, or at any time prior to such time, were directors or officers of the Parent or the Company (the "D&O Persons") entitled to be so indemnified, their liability limited or be exculpated with respect to any matters occurring on or prior to the Effective Time and relating to the fact that such D&O Person was a director or officer of the Parent or the Company immediately prior to the Effective Time, unless such amendment, repeal or other modification is required by applicable Law.
- (b) From and following the Effective Time, the Parent shall maintain, without any lapses in coverage and with an effective date as of the Closing Date, its directors' and officers' liability insurance for the benefit of those Persons who are currently covered by any comparable insurance policies of Parent as of the date of this Agreement with respect to matters occurring on or prior to the Effective Time. In addition, Parent shall purchase, prior to the Effective Time, a six-year prepaid "tail policy" for the non-cancellable extension of the directors' and officers' liability coverage of Parent's existing directors' and officers' insurance policies for a claims reporting or discovery period of at least six years from and after the Effective Time with respect to any claim related to any period of time at or prior to the Effective Time (the "D&O Tail Policy"). In satisfying its obligations pursuant to this Section 7.16(b), Parent will not be obligated to pay annual premiums in excess of Two Hundred and Fifty percent (250%) of the

amount paid by Parent for coverage for its last full fiscal year (such Two Hundred and Fifty percent (250%) amount, the "Maximum Annual Premium"). If the annual premiums of such insurance coverage exceed the Maximum Annual Premium, then the Parent will be obligated to obtain a directors' and officers' liability insurance policy with the greatest coverage available for a cost not exceeding the Maximum Annual Premium from an insurance carrier with the same or better credit rating as the Parent's current directors' and officers' liability insurance carrier.

- (c) If the Parent, the Company or any of their respective successors or assigns (i) shall merge or consolidate with or merge into any other corporation or entity and shall not be the surviving or continuing corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of their respective properties and assets as an entity in one or a series of related transactions to any Person, then in each such case, proper provisions shall be made so that the successors or assigns of the Parent or the Company, as applicable, shall assume all of the obligations set forth in this Section 7.16.
- (d) The D&O Persons entitled to the indemnification, liability limitation, exculpation and insurance set forth in this $\underline{\text{Section 7.16}}$ are intended to be third-party beneficiaries of this $\underline{\text{Section 7.16}}$. This $\underline{\text{Section 7.16}}$ shall survive the consummation of the transactions contemplated by this Agreement and shall be binding on all successors and assigns of the Parent or the Company, and such obligations shall be in addition to any and all indemnification, liability limitation and exculpation afforded to such D&O Persons under the Parent Organizational Documents or applicable law.
- 7.17. Parent Transaction Expenses Statement; Indebtedness. No later than Eight (8) Business Days prior to the expected Closing Date, the Parent shall deliver to the Company a statement setting forth its reasonable good faith estimate of the amount of Parent Transaction Expenses and Indebtedness, with elements constituting Parent Transaction Expenses and Indebtedness separately categorized, and indicating if they have been accrued, should be accrued under GAAP, or that are payable at or as of the Effective Time.
- 7.18. Reverse Split. The Parent and the Company shall use their reasonable efforts to agree upon a reverse split ratio applicable to a reverse split of the Parent's common stock (with a concurrent reduction of the authorized capital stock by the same ratio) pursuant to the Reincorporation and Charter Amendment such that the Parent's Common Stock would be reasonably expected to remain in full compliance with Nasdaq's minimum bid price rule even after the issuance of the New Share Amount (the "Reverse Split"). If Parent and the Company are unable to agree to the reverse split ratio or other terms of the Reverse Split at least twenty (20) Business Days prior to the Closing, then the reverse split ratio shall be such that every 4 shares of Parent's Common Stock shall become 1 share of Parent Common Stock, and with cash paid in lieu of fractional shares according to customary market practice. Parent and the Company shall reasonably cooperate with each other to determine the timing for filing of the Reincorporation and Charter Amendment and, pursuant to such agreed time, the Parent shall file the Reincorporation and Charter Amendment with the appropriate Government Authorities. Promptly after the Reincorporation and Charter Amendment is effective in the State of Delaware, the Parent Board shall adopt, prior to Closing, resolutions substantially identical to those adopted in connection with its approval of this Agreement and the Merger, and such other resolutions as the Company may reasonably request.

- 7.19. Litigation, At all times until the Effective Time, the Parent shall control the defense of any Transaction Litigation, but the Parent will provide Company with prompt notice of all Transaction Litigation (including by providing copies of all pleadings with respect thereto) and keep Company promptly and reasonably informed with respect to the status thereof. The Parent will (a) give Company the opportunity to participate (but not control, and at Company's sole cost and expense) in the defense, settlement or prosecution of any Transaction Litigation; and (b) consult with Company with respect to the defense, settlement and prosecution of any Transaction Litigation. The Parent may not compromise, settle or come to an arrangement regarding, or agree to compromise, settle or come to an arrangement regarding, any Transaction Litigation unless Company has consented thereto in writing (which consent shall not be unreasonably withheld, delayed or conditioned). For purposes of this Section 7.19, "participate" means that Company will be kept apprised of proposed strategy and other significant decisions with respect to the Transaction Litigation by the Parent (to the extent that the attorney-client privilege between the Parent and its counsel is not undermined or otherwise adversely affected or jeopardized in any respect), and Company may offer comments or suggestions with respect to such Transaction Litigation but will not be afforded any decision-making power or other authority over such Transaction Litigation except for the settlement or compromise consent set forth above. Any amounts paid or incurred in connection with any such Transaction Litigation (including the reasonable costs of attorney's fees), if any, will not operate to reduce the Cash Dividend or Net Working Capital of the Parent. For the avoidance of doubt, any Action against the Company shall be controlled by the Company and any Action against the Company and Parent shall be jointly controlled, with each Party cooperating in good faith, applying the principles of this Section 7.19 to the maximum extent possible.
- 7.20. Continued Employment. Surviving Company and Parent will take reasonable steps to continue the employment on and after the Closing Date of each employee ("Employee") who is employed by Parent on the Closing Date. For a period of twelve (12) months following the Closing Date, Surviving Company will cause Parent to provide, and Parent will provide, each Employee who continues employment with Parent with an annual base salary or wage rate that is at least equal to the Employee's annual base salary or wage rate as in effect immediately prior to the Closing Date. In addition, for a period of twelve (12) months following the Closing Date, Surviving Company and Parent agree, subject to eligibility conditions and other terms of the applicable employee benefit plans and programs, to provide Employees who continue employment Parent with health and welfare benefits that are, in the aggregate, substantially comparable to the health and welfare benefits provided to Employees immediately prior to the Closing Date, it being understood that the parties agree that there will no obligation to make any contribution to or, or maintain the ESOP after the Closing Date. Notwithstanding anything in this Section 7.20 to the contrary, nothing contained herein, whether express or implied, shall be treated as an establishment, amendment or other modification of any Parent Plan or any employee benefit plan of the Company, the Parent or the Surviving Company, or shall limit the right of the Surviving Company or Parent to amend, terminate or otherwise modify any Parent Plan or other employee benefit plan of the Surviving Company following the Closing Date. The Parent and the Company acknowledge and agree that all provisions contained in this Section 7.20 are included for their sole benefit, and that nothing in this Section 7.20, whether express or implied, shall create any third party beneficiary or other rights:

 (i) in any other Person, including any Employee, any participant in any Parent Plan or other employee benefit plan, or any d

the Surviving Company or to any particular term or condition of employment. For the avoidance of doubt, all Employees are and will be, at all times, deemed "employees at will," under Minnesota law and subject to termination with or without cause and with or without notice, subject to Minnesota law.

ARTICLE VIII CONDITIONS TO MERGER

- 8.1. Conditions to Each Party's Obligation to Effect the Merger. The respective obligations of each Party to this Agreement to consummate the Transactions shall be subject to the satisfaction or waiver (where permissible pursuant to applicable Law) on or prior to the Closing Date of each of the following conditions:
- (a) <u>Company Stockholder Approval</u>. The Company Stockholder Approval shall have been obtained, and such approval shall not have been rescinded, withdrawn or modified in any way and shall remain in full force and effect.
- (b) <u>Shareholder Approval</u>. The Parent Shareholder Approval shall have been obtained, and such approval shall not have been rescinded, withdrawn or modified in any way and shall remain in full force and effect.
- (c) <u>Registration Statement; Proxy Statement/Consent/Prospectus</u>. The Registration Statement shall have become effective under the Securities Act and no stop order suspending the effectiveness of the Registration Statement shall have been issued and no proceeding for that purpose, and no similar proceeding with respect to the Proxy Statement/Consent/Prospectus, shall have been initiated or threatened in writing by the SEC or its staff.
- (d) No Injunctions. No Governmental Authority of competent jurisdiction shall have enacted, issued, promulgated, enforced, or entered any Laws or Governmental Orders, whether temporary, preliminary, or permanent, that make illegal, enjoin, or otherwise prohibit consummation of the Transactions.
- (e) Reincorporation and Charter Amendment; Reverse Split. The Reincorporation and Charter Amendment shall have been filed, accepted and evidence of filing received by the Parent from the Secretary of State of the State of Delaware and any related filings required by the Secretary of State of the State of the State of Minnesota shall have been filed, accepted and evidence of the filing received by the Parent, and the Reverse Split shall have occurred and be effective in the marketplace.
- (f) <u>PIPE Investment</u>. The PIPE Investment shall have been consummated (or shall be consummated concurrently with the Closing), generating gross proceeds of no less than \$20.0 million.
- (g) <u>Regulatory Approvals</u>. All applicable waiting periods (and any extension thereof) applicable to the Merger under the HSR Act shall have expired or early termination of

such waiting periods shall have been granted and all applicable foreign antitrust approvals shall have been obtained.

- (h) <u>Nasdaq Listing Application</u>. The Nasdaq Listing Application shall have been approved by Nasdaq such that the Parent Common Stock to be issued as Merger Consideration will be listed on Nasdaq as of the Effective Time, subject only to official notice of issuance thereof, and there shall be no further approvals required by, or deficiencies threatened or pending from, Nasdaq in connection with the foregoing.
- 8.2. Conditions to the Obligations of the Parent and the Merger Sub. The obligations of the Parent and the Merger Sub to consummate the Transactions shall be subject to the satisfaction or waiver (where permissible pursuant to applicable Law) by the Parent and the Merger Sub on or prior to the Closing Date of each of the following conditions:
 - (a) Representations and Warranties.
 - (i) Other than the Company Fundamental Representations and the representations set forth in Section 4.1(a), the representations and warranties of the Company contained in Articles III and IV shall be true and correct both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct as of such specified date), and shall be interpreted without giving effect to the words "materially" or "material" or to any qualifications based on such terms or based on the term "Company Material Adverse Effect", except where the failure of such representations and warranties to be true and correct, in the aggregate, does not constitute a Company Material Adverse Effect.
 - (ii) The Company Fundamental Representations that (A) are not qualified by the words "materially" or "material" or any qualifications based on such terms or based on the term "Company Material Adverse Effect" shall be true and correct in all material respects both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct in all material respects as of such specified date) and (B) that are qualified by the words "materially" or "material" or any qualifications based on such terms or based on the term "Company Material Adverse Effect" shall be true and correct in all respects (and, for the avoidance of doubt, taking into account such qualifications) both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct as of such specified date), except, in the case of the foregoing clauses (A) and (B), for such inaccuracies as are in the aggregate *de minimis*.
 - (iii) The representations and warranties set forth in Section 4.1(a) shall be true and correct in all respect.

- (b) <u>Performance of Obligations of Company</u>. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement on or prior to the Closing Date.
- (c) No Company Material Adverse Effect. No Company Material Adverse Effect shall have occurred since the date of this Agreement and be continuing.
- (d) Officers' Certificate. The Parent shall have received an officers' certificate duly executed by each of the Chief Executive Officer and Chief Financial Officer of Company to the effect that the conditions of Sections 8.2(a), (b) and (c), have been satisfied.
- 8.3. Conditions to the Obligations of Company. The obligation of the Company to consummate the Transactions shall be subject to the satisfaction or waiver (where permissible pursuant to applicable Law) by the Company on or prior to the Closing Date of each of the following conditions:
 - (a) Representations and Warranties.
 - (i) Other than the Parent Fundamental Representations and the representations set forth in Section 5.8(a), the representations and warranties of the Parent contained in Article V shall be true and correct both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct as of such specified date), and shall be interpreted without giving effect to the words "materially" or "material" or to any qualifications based on such terms or based on the term "Parent Material Adverse Effect", except where the failure of such representations and warranties to be true and correct, in the aggregate, does not constitute a Parent Material Adverse Effect.
 - (ii) The Parent Fundamental Representations that (A) are not qualified by the words "materially" or "material" or any qualifications based on such terms or based on the term "Parent Material Adverse Effect" shall be true and correct in all material respects both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct in all material respects as of such specified date) and (B) that are qualified by the words "materially" or "material" or any qualifications based on such terms or based on the term "Parent Material Adverse Effect" shall be true and correct in all respects (and, for the avoidance of doubt, taking into account such qualifications) both at and as of the date hereof and at and as of the Closing Date as though made at and as of the Closing Date (except to the extent such representations and warranties speak as of a specified date, in which case they need only be true and correct as of such specified date), except, in the case of the foregoing clauses (A) and (B), for such inaccuracies as are in the aggregate *de minimis*.

- (iii) The representations and warranties set forth in Section 5.8(a) shall be true and correct in all respect.
- (b) <u>Performance of Obligations of Parent and Merger Sub.</u> Parent and Merger Sub shall have performed in all material respects all obligations required to be performed by them under this Agreement on or prior to the Closing Date.
- (c) <u>No Parent Material Adverse Effect</u>. No Parent Material Adverse Effect shall have occurred since the date of this Agreement and be continuing.
- (d) Officers' Certificate. Company shall have received an officers' certificate duly executed by each of the Chief Executive Officer and Chief Financial Officer of Parent to the effect that the conditions of Sections 8.3(a), (i), and (c) have been satisfied.
 - (e) Resignations. All members of the Parent Board shall have executed written resignations effective as of the Effective Time.

ARTICLE IX TERMINATION AND AMENDMENT

- 9.1. Termination. This Agreement may be terminated at any time prior to the Effective Time (with respect to Sections 9.1(b) through 9.1 (h), by written notice by the terminating Party to the other Party), whether before or, subject to the terms hereof, after obtaining Parent Shareholder Approval, only as follows:
 - (a) by mutual written consent of the Parent and the Company;
- (b) by either the Parent or the Company if the Merger shall not have been consummated by January 31, 2023 (the "Outside Date") (provided that the right to terminate this Agreement under this Section 9.1) shall not be available to any Party (and in the case of the Parent, Merger Sub) whose failure to fulfill any obligation under this Agreement has been a principal cause of or resulted in the failure of the Merger to occur on or before the Outside Date); provided, however, that, in the event that the Registration Statement or the Proxy Statement/Consent/Prospectus is still being reviewed or commented on by the SEC on such date, then the Outside Date shall be automatically extended for an additional forty (40) days from such date;
- (c) by either the Parent or the Company if a Governmental Authority of competent jurisdiction shall have issued a nonappealable final order, decree or ruling or taken any other nonappealable final action, in each case having the effect of permanently restraining, enjoining or otherwise prohibiting the Merger; *provided*, *however*, that a Party hereto shall not be permitted to terminate this Agreement pursuant to this Section 9.1 (c) if the issuance of any such order, decree, ruling or other action is attributable to the failure of such Party (or any Affiliate of such Party) to perform in any material respect any covenant in this Agreement required to be performed by such Party (or any Affiliate of such Party) at or prior to the Effective Time;

- (d) by either the Parent or the Company if, at the Parent Shareholder Meeting (including any adjournment or postponement) at which a vote on the Parent Shareholder Approval is taken, the Parent Shareholder Approval shall not have been obtained; *provided*, that it being understood that the right to terminate this Agreement pursuant to this Section 9.1(d) will not be available to any Party whose action or failure to act (which action or failure to act constitutes a breach by such Party of this Agreement (and it being understood that a breach of this Agreement by Merger Sub shall be deemed to be a breach by Parent for purposes of this Agreement)) has been the cause of, or resulted in, the failure to obtain the Parent Shareholder Approval at the Parent Shareholder Meeting (or any adjournment or postponement thereof);
- (e) by the Company, at any time prior to the receipt of the Parent Shareholder Approval, if a Parent Board Recommendation Change has occurred (whether or not in compliance with <u>Section 7.1</u>) or the Parent shall have breached its obligations under <u>Section 7.1</u> or <u>Section 7.5</u> (b) of this Agreement;
- (f) by the Parent, if there has been a material breach of or material failure to perform any representation, warranty, covenant or agreement set forth in this Agreement on the part of the Company, which breach would cause the conditions set forth in Section 8.2(a) or (i) not to be satisfied; provided that neither the Parent nor the Merger Sub is then in material breach of any representation, warranty or covenant under this Agreement and provided, further, that if such breach or failure to perform is curable by the Company, as applicable, then this Agreement shall not terminate pursuant to this Section 9.1(f) as a result of such particular breach or failure until the earlier of (i) the expiration of a ten (10) day period commencing upon delivery of written notice from the Parent to the Company of such breach or failure and (ii) the Company ceasing to exercise commercially reasonable efforts to cure such breach or failure following delivery of written notice from the Parent of such breach or failure and its intention to terminate pursuant to this Section 9.1(f) (it being understood that, in each case, this Agreement shall not terminate pursuant to this Section 9.1(f) as a result of such particular breach or failure if such breach or failure is cured prior to such termination becoming effective);
- (g) by the Company, if there has been a material breach of or material failure to perform any representation, warranty, covenant or agreement set forth in this Agreement on the part of the Parent, which breach would cause the conditions set forth in Section 8.3(a) or 8.3(a)(i) not to be satisfied; provided that the Company is not then in material breach of any representation, warranty or covenant under this Agreement and provided, further, that if such breach or failure to perform is curable by the Parent, then this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or failure until the earlier of (i) the expiration of a ten (10) day period commencing upon delivery of written notice from the Company to the Parent of such breach or failure and (ii) the Parent or the Merger Sub (as applicable) ceasing to exercise commercially reasonable efforts to cure such breach or failure following delivery of written notice from Company of such breach or failure and its intention to terminate pursuant to this Section 9.1(g) (it being understood that, in each case, this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or failure if such breach or failure is cured prior to such termination becoming effective); or
- (h) by the Parent if, at any time prior to the receipt of the Parent Shareholder Approval with respect to the issuance of shares of Parent Common Stock pursuant to the Merger,

68 of 96

each of the following occur: (i) the Parent shall have received a Superior Proposal; (ii) the Parent shall have complied in all material respects with its obligations under Section 7.1, including with respect to making a Parent Board Recommendation Change with respect to such Superior Proposal; (iii) the Parent Board approves, and the Parent concurrently with the termination of this Agreement enters into, a definitive agreement with respect to such Superior Proposal (iv) as a condition to such termination the Parent pays the Termination Fee due to the Company in accordance with the applicable provision of Section 9.3(b).

- (i) The Party desiring to terminate this Agreement shall deliver prompt written notice of such termination to the other Party, setting forth in such notice the provision of this Section 9.1 pursuant to which such Party is terminating this Agreement and the facts and circumstances forming the basis for such termination pursuant to such provision.
- 9.2. Effect of Termination. In the event of termination of this Agreement as provided in Section 9.1, this Agreement shall immediately become void and there shall be no Liability or obligation on the part of the Parent, the Company, the Merger Sub or their respective officers, directors, shareholders or Affiliates; provided that (a) any such termination shall not relieve any Party from Liability for any knowing and intentional breach of this Agreement, fraud or intentional misconduct and (b) the provisions of Section 6.3 (Confidentiality), this Section 9.2 (Effect of Termination), Section 9.3 (Fees and Expenses) and Article X (MISCELLANEOUS) of this Agreement and the Confidentiality Agreement shall remain in full force and effect and survive any termination of this Agreement.

9.3. Fees and Expenses.

electro220769 ex2-1.htm

- (a) Except as set forth in this Section 9.3, (i) if this Agreement is terminated before the Merger is consummated, (w) the Parent shall pay the Parent Transaction Expenses, (other than the Shared Expenses owed by the Company), (x) the Company shall pay the Company Transaction Expenses, (y) the Company shall pay 100% of all fees and expenses payable to B. Riley in connection with the PIPE Investments (regardless of whether the Company or the Parent engaged B. Riley) or payable to the proposed investors therein in respect of their respective legal counsel's review of the PIPE Investment and (z) the Parent and the Company shall share equally and pay their respective shares of the Shared Expenses, and (ii) if the Merger is consummated, all fees and expenses incurred by either the Parent or the Company in connection with this Agreement, the Transactions (including all Parent Transaction Expenses, Company Transaction Expenses and expenses payable by either the Parent or the Company related to the PIPE Investment) shall be paid by the Parent promptly after the Effective Time, and the Parties acknowledge that certain of those expenses incurred by the Parent shall be shared through the adjustments to the Cash Dividend set forth in Section 2.11.
 - (b) If:
 - (i) Parent terminates this Agreement pursuant to Section 9.1(h);
 - (ii) the Company terminates this Agreement pursuant to <u>Section 9.1(e)</u> or would have been entitled to terminate this Agreement pursuant to Section 9.1(e) prior

to or at the time the Parent terminates this Agreement pursuant to Section 9.1(b) or 9.1(d); or

- (A) after the date of this Agreement, an Acquisition Proposal with respect to Parent is made, proposed or communicated to the Parent Board or management, or is publicly made, proposed or communicated or otherwise becomes publicly known; (B) thereafter this Agreement is terminated by either Parent or the Company pursuant to Section 9.1(b) or Section 9.1(d), or by the Company pursuant to Section 9.1(g); and (C) within twelve (12) months of such termination (1) any transaction included within the definition of an Acquisition Proposal with respect to the Parent is consummated or (2) the Parent enters into a definitive agreement providing for the consummation of any transaction within the definition of Acquisition Proposal, in each case whether or not involving the same Acquisition Proposal or the Person or group making the Acquisition Proposal referred to in clause (A); provided that for purposes of clause (C), the term "Acquisition Proposal" shall have the meaning assigned to such term in Exhibit B, except that all references to "15%" in such definition shall be deemed references to "50%", then the Parent shall pay to the Company (or its designee) a fee of \$1,100,000 (the "Parent Termination Fee") plus reimbursement of expenses pursuant to Section 9.3(c). Any fee due under this Section 9.3(b)(iii) shall be paid by wire transfer of same-day funds to an account designated by the Company (1) in the case of clause (i) above, prior to or simultaneously with such termination of this Agreement, (2) in the case of clause (ii) above, within five (5) Business Days after the date of such termination of this Agreement and (3) in the case of clause (iii) above, within five (5) Business Days after the earlier of (x) the date of the consummation of the transaction referred to in clause (C)(1) thereof and (y) the date of entry into the definitive agreement referred to in clause (C)(2) thereof. The parties hereto acknowledge and agree that in no event shall the Parent be required to pay the Parent Termination Fee on more than one occasion, whether or not the Parent Termination Fee may be payable under more than one provision of this Agreement at the same or at different times and the occurrence of different events. Notwithstanding anything herein to the contrary, the Company's right to receive the Parent Termination Fee pursuant to this Section 9.3(b) under circumstances under which the Company is entitled to receive the Parent Termination Fee shall be the sole and exclusive remedy of the Company or any of its Affiliates against the Parent or its Representatives for any and all losses that may be suffered based upon, resulting from or arising out of the circumstances giving rise to such termination and upon payment of the Parent Termination Fee to the Company, the Parent and its Representatives shall have no Liability or obligation relating to or arising out of the failure to consummate the transactions contemplated by this Agreement; provided that nothing herein shall relieve any Party from Liability for any fraud or willful breach of this Agreement.
- (c) In the event of a termination of this Agreement contemplated by Section 9.3(b), the Parent shall pay to the Company (or its designee) promptly upon delivery of written demand by the Company to the Parent (which the Company may deliver at any time after this Agreement is terminated), but in no event later than five (5) Business Days after the date

such demand is delivered, by wire transfer of same-day funds to an account designated by the Company, all reasonable, documented out-of-pocket Company Transaction Expenses incurred by the Company in connection with this Agreement; <u>provided</u> that the Parent shall not be obligated to pay for Company Expenses in excess of \$575,000.

- (d) The Company and the Parent acknowledge and agree that the agreements contained in Sections 9.3(b) and 9.3(c) are an integral part of the Transactions, and that, without these agreements, the Company would not enter into this Agreement. Accordingly, if the Parent fails to promptly pay any amount due pursuant to Section 9.3(b) or 9.3(c), and in order to obtain such payment, the Company commences an Action against Parent that results in a judgment in the Company's favor for such payment, the Parent shall pay to the Company its costs and expenses (including attorneys' fees and expenses) in connection with such Action, together with interest on the amount of such payment from the date such payment was required to be made until the date of payment at the prime rate as published in the Wall Street Journal in effect on the date such payment was required to be made.
- 9.4. Amendment. This Agreement may be amended by the Parties hereto, by action taken or authorized by their respective boards, at any time before or after approval of the matters presented in connection with the Merger by the shareholders or members of any of the Parties, but, after any such approval, no amendment shall be made which by Law requires further approval by such shareholders without such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the Parties hereto.
- 9.5. Extension; Waiver. At any time prior to the Effective Time, the Parties hereto, by action taken or authorized by their respective boards, may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions contained herein. Any agreement on the part of a Party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such Party. Such extension or waiver shall not be deemed to apply to any time for performance, inaccuracy in any representation or warranty, or noncompliance with any agreement or condition, as the case may be, other than that which is specified in the extension or waiver. The failure of any Party to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights.
- 9.6. Procedure for Termination, Amendment, Extension or Waiver. A termination of this Agreement pursuant to Section 9.1, an amendment, modification or supplement of this Agreement pursuant to Section 9.4 or an extension or waiver of this Agreement pursuant to Section 9.5 shall, in order to be effective, require action by the respective boards of the applicable Parties.

ARTICLE X MISCELLANEOUS

- 10.1. Non-Survival of Representations and Warranties. The (a) representations and warranties of the Company (other than the Company Fundamental Representations), Parent and Merger Sub contained in this Agreement or any certificate or instrument delivered pursuant to this Agreement shall terminate at the Effective Time, (b) Company Fundamental Representations shall survive until the one (1) year anniversary of the Effective Time, and (c) only the covenants that by their terms survive the Effective Time and this Article X shall survive the Effective Time.
- 10.2. Notices. All notices and other communications hereunder shall be in writing and shall be deemed duly delivered (a) three (3) Business Days after being sent by registered or certified mail, return receipt requested, postage prepaid, or (b) one (1) Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable overnight courier service, in each case to the intended recipient as set forth below:

If to Parent or Merger Sub, to:

Electro-Sensors, Inc. Attention: David Klenk 6111 Blue Circle Drive Minnetonka, Minnesota 55343 Email:

with a copy (which shall not constitute notice) to:

Ballard Spahr LLP Attention: George H. Singer; Thomas G. Lovett, IV 2000 IDS Center 80 South 8th Street Minneapolis, MN 55402 Email:

If to Company, to:

Mobile X Global, Inc. Attention: Peter Adderton, Chief Executive Officer; Justin O'Neill, General Counsel 12 Sail Vista Newport Coast, CA 92657 Email: with a copy (which shall not constitute notice) to:

Proskauer Rose LLP Attention: Ben D. Orlanski 2029 Century Park East Suite 2400 Los Angeles, CA 90067 Email:

Any Party to this Agreement may give any notice or other communication hereunder using any other means (including personal delivery, messenger service, telecopy, ordinary mail or electronic mail), but no such notice or other communication shall be deemed to have been duly given unless and until it actually is received by the Party for whom it is intended. Any Party to this Agreement may change the address to which notices and other communications hereunder are to be delivered by giving the other parties to this Agreement notice in the manner herein set forth.

- 10.3. Entire Agreement. This Agreement (including the Schedules and Exhibits hereto and the documents and instruments referred to herein that are to be delivered at the Closing), the Parent Voting Agreement, the Company Voting Agreement and the Confidentiality Agreement constitute the entire agreement among the Parties to this Agreement and supersedes any prior understandings, agreements or representations by or among the Parties hereto, or any of them, written or oral, with respect to the subject matter hereof and the Parties hereto expressly disclaim reliance on any such prior understandings, agreements or representations to the extent not embodied in this Agreement. Notwithstanding the foregoing, the Confidentiality Agreement shall remain in effect in accordance with its terms.
- 10.4. No Third-Party Beneficiaries. This Agreement is not intended to, and shall not, confer upon any other person any rights or remedies hereunder, except as expressly provided herein.
- 10.5. Assignment. No Party may assign any of its rights or delegate any of its performance obligations under this Agreement, in whole or in part, by operation of law or otherwise without the prior written consent of the other Parties, and any such assignment or attempted or purported assignment without such prior written consent shall be null and void. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of, and be enforceable by, the Parties hereto and their respective successors and permitted assigns. Any purported assignment of rights or delegation of performance obligations in violation of this Section 10.5 is void.
- 10.6. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the Parties hereto agree that the court making such determination shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable

term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the Parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term.

- 10.7. Counterparts and Signatures. This Agreement may be executed in two or more counterparts (including by facsimile or by an electronic scan delivered by electronic mail), each of which shall be deemed an original but all of which together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each of the Parties hereto and delivered to the other parties, it being understood that all parties need not sign the same counterpart. This Agreement may be executed and delivered by facsimile or by an electronic scan delivered by electronic mail.
- 10.8. Governing Law. All matters arising out of or relating to this Agreement and the Transactions (including its interpretation, construction, performance and enforcement) shall be governed by and construed in accordance with the internal laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of laws of any jurisdictions other than those of the State of Delaware, except that the matters contained in Article II with regards to the effects of the Merger between the Company and the Merger Sub shall be governed by the DGCL and MBCA, without giving effect to any choice or conflict of law provision or rule.
- 10.9. Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy. The Parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which they are entitled at law or in equity.
- 10.10. Submission to Jurisdiction. Each of the Parties to this Agreement (a) consents to submit itself to the exclusive personal jurisdiction of the Court of Chancery of the State of Delaware or, to the extent such court does not have subject matter jurisdiction, the United States District Court for the District of Delaware or, to the extent that neither of the foregoing courts has jurisdiction, the Superior Court of the State of Delaware (as applicable, the "Chosen Court") in any action or proceeding arising out of or relating to this Agreement or any of the Transactions, (b) agrees that all claims in respect of such action or proceeding shall be heard and determined in any such Chosen Court, (c) agrees that it shall not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such Chosen Court and (d) agrees not to bring any action or proceeding arising out of or relating to this Agreement or any of the transaction contemplated by this Agreement in any other court. Each of the Parties hereto waives any defense of inconvenient forum to the maintenance of any action or proceeding

so brought and waives any bond, surety or other security that might be required of any other Party with respect thereto. Any Party may make service on another Party by sending or delivering a copy of the process to the Party to be served at the address and in the manner provided for the giving of notices in <u>Section 10.1</u>. Nothing in this <u>Section 10.10</u>, however, shall affect the right of any Party to serve legal process in any other manner permitted by law.

- 10.11. WAIVER OF JURY TRIAL. EACH OF PARENT, THE MERGER SUB AND COMPANY HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS OR THE ACTIONS OF PARENT, THE MERGER SUB OR COMPANY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS AGREEMENT.
- 10.12. Disclosure Schedule. Each of the Company Disclosure Schedule and the Parent Disclosure Schedule shall be arranged in sections corresponding to the numbered sections contained in this Agreement, and the disclosure in any section shall qualify only (a) the corresponding section of this Agreement and (b) the other sections of this Agreement, to the extent that it is reasonably apparent from a reading of such disclosure that it also qualifies or applies to such other sections. The inclusion of any information in the Company Disclosure Schedule or the Parent Disclosure Schedule, as applicable, shall not be deemed to be an admission or acknowledgment, in and of itself, that such information is required by the terms hereof to be disclosed, is material, has resulted in or would result in a Company Material Adverse Effect or a Parent Material Adverse Effect, as applicable, or is outside the Ordinary Course of Business. The Parties acknowledge that Section 7.9 of the Parent Disclosure Schedule may be updated prior to the Effective Time to reflect any changes in the individuals to be appointed to the Parent Board and the class such individual is to serve or from whom resignations shall be obtained.
- 10.13. No Recourse. Except in the case of fraud, all actions, claims, obligations, liabilities or causes of actions (whether in contract or in tort, in law or in equity, or granted by statute whether by or through attempted piercing of the corporate, limited partnership or limited liability company veil) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to: (a) this Agreement, (b) the negotiation, execution, delivery or performance of this Agreement (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement), (c) any breach of this Agreement and (d) any failure of the Merger to be consummated, may be made only against (and, without prejudice to the rights of any express third-party beneficiary to whom rights under this Agreement inure pursuant to Section 10.4), solely those Persons that are expressly identified as parties to this Agreement and not against any Nonparty Affiliate (as defined below). Except in the case of fraud, on other Person, including any director, officer, employee, incorporator, member, partner, manager, stockholder, optionholder, Affiliate, agent, attorney or representative of, or any financial advisor or lender to (each of the foregoing, a "Nonparty Affiliate") any of the foregoing shall have any liabilities (whether in contract or in tort, in law or in equity, or granted by statute whether by or through attempted piercing of the corporate,

limited partnership or limited liability company veil) for any claims, causes of action, obligations or liabilities arising under, out of, in connection with or related in any manner to the items in the immediately preceding clauses (a) through (d) and each Party, on behalf of itself and its Affiliates, hereby irrevocably releases and forever discharges each of the Nonparty Affiliate from any such Liability or obligation. Nothing contained in this Section 10.13 shall relieve the Company or the Parent of any indemnification obligation set forth in this Agreement.

10.14. Legal Representation. The Parties agree that, notwithstanding the fact that Proskauer Rose LLP ("Proskauer") may have, prior to the Closing, represented the Company in connection with this Agreement, the documents and agreements referred to herein to be entered into by the Company and the transactions contemplated hereby and thereby, and has also represented the Company and/or its Affiliates in connection with matters other than the transaction that is the subject of this Agreement, Proskauer will be permitted in the future, after the Closing, to represent the Parent.

[Remainder of Page Intentionally Left Blank, Signature Page Follows]

electro220769_ex2-1.htm

electro220769_8k_FINAL 06/14/2022 04:19 PM

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

PARENT: ELECTRO-SENSORS, INC.

By: /s/ David L. Klenk

Name: David L. Klenk Title: Chief Executive Officer 76 of 96

[Signature Page to Agreement and Plan of Merger]

electro220769_ex2-1.htm electro220769_8k_FINAL 06/14/2022 04:19 PM

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

COMPANY: MOBILE X GLOBAL, INC.

By: /s/ Peter Adderton

Name: Peter Adderton
Title: Chief Executive Officer

77 of 96

[Signature Page to Agreement and Plan of Merger]

electro220769_ex2-1.htm

electro220769_8k_FINAL 06/14/2022 04:19 PM

78 of 96

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

MERGER SUB: MOBILE X NEWCO, INC.

By: /s/ David L. Klenk

Name: David L. Klenk Title: Chief Executive Officer

[Signature Page to Agreement and Plan of Merger]

EXHIBIT A-1

FORM OF PARENT VOTING AGREEMENT

The Form of Parent Voting Agreement is filed as Exhibit 10.1 to this Form 8-K

A-1-1

EXHIBIT A-2

FORM OF COMPANY VOTING AGREEMENT

The Form of Company Voting Agreement is filed as Exhibit 10.2 to this Form 8-K

EXHIBIT B

CERTAIN DEFINED TERMS

"Acquisition Proposal" means any proposal, offer or indication of interest from a third party (whether or not in writing) relating to, or that would reasonably be expected to lead to, in one transaction or a series of transactions, (i) any direct or indirect acquisition or purchase (including by any exclusive license, exclusive sublicense or lease) by any Person or group (as defined under Section 13(d) of the Exchange Act and the rules and regulations thereunder) of (A) assets (including equity securities of any Subsidiary of the Parent or any Intellectual Property Rights) or businesses that constitute 15% or more of the revenues, net income or assets of the Parent, taken as a whole, or (B) beneficial ownership of 15% or more of any class of equity securities of the Parent (ii) any purchase or sale of, or tender offer or exchange offer for, equity securities of the Parent that, if consummated, would result in any person or group (as defined under Section 13(d) of the Exchange Act and the rules and regulations thereunder) beneficially owning 15% or more of any class of equity securities of the Parent; or (iii) any merger, consolidation, business combination, recapitalization, reorganization, dual listed structure, joint venture, share exchange or similar transaction involving the Parent, as a result of which the owners of the equity securities of the Parent immediately prior to such event own less than 85% of the equity securities of the Parent immediately following such event, or any combination of the foregoing clauses (A), (B), and (C); in each case other than the Transactions; provided that the PIPE Investment shall not be considered an Acquisition Proposal.

"Action" means any claim, action, cause of action, demand, lawsuit, arbitration, inquiry, audit, proceeding, litigation, citation, summons, subpoena or investigation of a civil, criminal, administrative or regulatory nature, whether at law or in equity.

"Affiliate" of a Person means any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

"beneficial owner" means a person who shall be deemed to be the beneficial owner as determined by Rule 13d-3 of the Exchange Act.

"Business Day" means any day on which the principal offices of the SEC in Washington, D.C. are open to accept filings, or, in the case of determining a date when any payment is due, any day on which banks are not required or authorized to close in New York, NY; provided, that banks shall not be deemed to be authorized or obligated to be closed due to a "shelter in place" or similar closure of physical branch locations at the direction of any Governmental Authority if such banks' electronic funds transfer systems (including for wire transfers) are open for use by customers on such day..

"Code" means the Internal Revenue Code of 1986, as amended.

"Company Fundamental Representations" means the representations and warranties set forth in Section 3.1 (Organization; Power and Enforceability); Section 3.2 (Authority); Section 3.3 (Capitalization) and Section 3.9 (Brokers).

"Company Material Adverse Effect" means any change, effect, event, circumstance or development that is materially adverse to or has a material adverse effect on (i) the business, assets, liabilities, capitalization, financial condition, or results of operations of the Company, taken as a whole, or (ii) the ability of Parent or Merger Sub to consummate the Merger or any of the other Transactions or to perform any of their respective covenants or obligations under this Agreement; provided, however, that none of the following, to the extent arising after the date of this Agreement, either alone or in combination, shall be deemed to be a Company Material Adverse Effect, and none of the following shall be taken into account in determining whether there has been or will be a Company Material Adverse Effect: any change or event caused by or resulting from (A) changes in prevailing economic or market conditions in the United States or any other jurisdiction in which such entity has substantial business operations (except to the extent those changes have a disproportionate effect on the Company relative to the other participants of similar size operating in the industry or industries in which the Company operates), (B) changes or events affecting the industry or industries in which the Company operates generally or compete (except to the extent those changes or events have a disproportionate effect on the Company relative to the other participants of similar size operating in the industry or industries in which the Company operates), (C) changes in generally accepted accounting principles or requirements (except to the extent those changes have a disproportionate effect on the Company relative to the other participants of similar size operating in the industry or industries in which the Company operates), (D) changes in laws, rules or regulations of general applicability or interpretations thereof by any Governmental Authority (except to the extent those changes have a disproportionate effect on the Company relative to the other participants of similar size operating in the industry or industries in which the Company operates), (E) any natural disaster, pandemic or any outbreak of major hostilities in which the United States is involved or any act of terrorism within the United States or directed against its facilities or citizens wherever located or any governmental response to any of the foregoing (except to the extent those changes or events have a disproportionate effect on the Company relative to the other participants of similar size operating in the industry or industries in which the Company operates), or (F) any failure by Company to meet any internal guidance, budgets, plans, projections or forecasts of its revenues, earnings or other financial performance or results of operations (but not, in the case of Clause (F), the underlying cause of such changes or failures, unless such changes or failures would otherwise be excepted from this definition). For the avoidance of doubt, the Parties agree that the terms "material," "materially" and "materiality" as used in this Agreement with an initial lowercase "m" shall have their respective customary and ordinary meanings, without regard to the meanings ascribed to Company Material Adverse Effect or Parent Material Adverse Effect, in each case as defined in this Agreement. Notwithstanding anything to the contrary in this Agreement, any event, fact, development, circumstance, change, effect, or occurrence to the extent it arises out of compliance by the Company with a Law, directive, guideline or recommendation promulgated by any Governmental Entity to respond to the impact, presence, outbreak or spread of COVID-19 shall not be deemed, either alone or in combination, to constitute or contribute to a Company Material Adverse Effect (except to the extent those changes have a disproportionate effect on the

Company relative to the other participants of similar size operating in the industry or industries in which the Company operates).

"Company Shares" means the authorized, issued and outstanding shares of common stock of the Company.

"Company Share Amount" means the total number of all outstanding Company Shares (other than the Excluded Company Shares) assuming full conversion, exercise and exchange of all Equity Interests of Company, whether or not vested, in the money or subject to conditions to conversion, exercise and exchange.

"Company Stockholders" means the record and beneficial owners of Company Shares immediately prior to the Effective Time.

"Company Stockholder Agreement" means that certain Stockholders agreement, entered into as of April 12, 201, among the Company, Peter Adderton, Stuart Lachlan Bruce and Danielle Lauren Bird as trustees of the Bird Family Trust, Dan Cohrs, Peter Hill and Justin J. O'Neill, as trustee for the Justin J. O'Neill Trust, as it may be amended and supplemented from time to time.

"Company Transaction Expenses" means all fees and expenses incurred by Company at or prior to the Effective Time in connection with this Agreement and the transactions contemplated hereby, including any fees and expenses of legal counsel and accountants, the maximum amount of fees and expenses payable to financial advisors, investment bankers, brokers, consultants, and other advisors of Company.

"Consent Solicitation Statement" means the consent solicitation statement included (except as provided in Section 7.3(a)) as part of the Registration Statement with respect to the solicitation by the Company of the Company Stockholder Approval.

"Contracts" means all contracts, leases, deeds, mortgages, licenses, instruments, notes, commitments, undertakings, indentures, joint ventures and all other agreements, commitments and legally binding arrangements, whether written or oral.

"Dollars" or "\$" means the lawful currency of the United States.

"Effective Time" means the effective time of the Certificate of Merger.

"Encumbrance" means any charge, claim, community property interest, pledge, condition, equitable interest, lien (statutory or other), option, security interest, mortgage, easement, encroachment, right of way, right of first refusal, or restriction of any kind, including any restriction on use, voting, transfer, receipt of income or exercise of any other attribute of ownership.

"Environmental, Health and Safety Laws" means all foreign, federal, state, and local Laws and Contracts in effect as of or prior to the Closing Date relating to or imposing Liability

or standards of conduct concerning public health and safety, employee health and safety, pollution, or protection of the environment.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.

"ERISA Affiliate" means any Person that, together with the Company, at any relevant time would be treated as a single employer under Section 414 of the Code.

"Exchange Act" means Securities Exchange Act of 1934, as amended.

"Exchange Ratio" means a quantity equal to the New Share Amount <u>plus</u> the Parent Diluted Amount, <u>times</u> 0.7627, <u>divided by</u> the Company Share Amount, expressed as a decimal, as may be adjusted pursuant to <u>Section 2.7</u> of this Agreement. The Exchange Ratio shall be calculated on a post-Reverse Split basis, even if the Reverse Split has not become legally effective at the time that the Exchange Ratio is being calculated.

"Fair Market Value" means, with respect to Parent Stock Options, the definition contained in the Parent Stock Plans, as applicable.

"GAAP" means United States generally accepted accounting principles in effect from time to time.

"General Enforceability Exceptions" means bankruptcy, insolvency, reorganization, moratorium or other similar laws of general application relating to or affecting the enforcement of creditors' rights and by general equitable principles.

"Governmental Authority" means any federal, state, local or foreign government or political subdivision thereof, or any agency or instrumentality of such government or political subdivision, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of Law), or any arbitrator, court or tribunal of competent jurisdiction.

"Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.

"<u>Hazardous Materials</u>" means all hazardous, toxic, explosive or radioactive substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos, polychlorinated biphenyls, radon gas, noise, mold, odor and all other substances or wastes of any nature regulated, listed, defined, or for which Liability or standards of conduct may be imposed pursuant to, any Environmental, Health and Safety Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" means, without duplication and with respect to the Parent, all (a) indebtedness for borrowed money, whether current, short-term, secured or unsecured or indebtedness issued or incurred in substitution or exchange for indebtedness for borrowed money; (b) obligations for the deferred purchase price of capital equipment not in the ordinary course of business (but excluding any payables included in the determination of Net Working Capital); (c) long or short-term obligations evidenced by notes, bonds, debentures or other similar instruments; (d) obligations under any interest rate, currency swap or other hedging agreement or arrangement; (e) capital lease obligations; (f) reimbursement obligations under any letter of credit, banker's acceptance or similar credit transactions; (g) currency or interest rate swaps, collars, caps, hedges, derivatives or similar arrangements; (h) bank overdrafts and deferred liabilities; (i) guarantees made by the Parent on behalf of any third party in respect of obligations of the kind referred to in the foregoing clauses (a) through (h); (j) any unpaid interest, prepayment penalties, costs and fees that would arise or become due as a result of the prepayment of any of the obligations referred to in the foregoing clauses (a) through (i); and (k) all amounts (plus any associated withholding Taxes or any Taxes required to be paid by the Parent with respect thereto) payable by the Parent, whether immediately or in the future, under or with respect to any "change of control," retention, termination, compensation, severance or other similar arrangements by reason of (either alone or in conjunction with any other event, such as termination or continuation of employment) the consummation of the transactions contemplated by this Agreement (the amounts referred to in this clause (k), the "Transaction Bonus Amounts").

"Intellectual Property" means any and all rights in, arising out of, or associated with any of the following in any jurisdiction throughout the world: (a) issued patents and patent applications (whether provisional or non-provisional), including divisionals, continuations, continuations-in-part, substitutions, reissues, reexaminations, extensions, or restorations of any of the foregoing, and other Governmental Authority-issued indicia of invention ownership (including certificates of invention, petty patents, and patent utility models) ("Patents"); (b) trademarks, service marks, brands, certification marks, logos, trade dress, trade names, and other similar indicia of source or origin, together with the goodwill connected with the use of and symbolized by, and all registrations, applications for registration, and renewals of, any of the foregoing ("Trademarks"); (c) copyrights and works of authorship, whether or not copyrightable, and all registrations, applications for registration, and renewals of any of the foregoing ("Copyrights"); (d) internet domain names and social media account or user names (including "handles"), whether or not Trademarks, all associated web addresses, URLs, websites and web pages, social media sites and pages, and all content and data thereon or relating thereto, whether or not Copyrights; (e) mask works, and all registrations, applications for registration, and renewals thereof; (f) industrial designs, and all Patents, registrations, applications for registration, and renewals thereof; (g) trade secrets, know-how, inventions (whether or not patentable), discoveries, improvements, technology, business and technical information, databases, data compilations and collections, tools, methods, processes, techniques, and other confidential and proprietary information and all rights therein ("Trade Secrets"); (h) computer programs, operating systems, applications, firmware, and other code, including all source code, object code, application programming interfaces, data files, databases, prot

"Knowledge of the Company" and similar expressions mean the actual knowledge of the persons identified on Section K of the Company Disclosure Schedule, and such knowledge as such persons would reasonably be expected to have obtained, after reasonable inquiry, in the course of their performance of their positions at the Company.

"Knowledge of the Parent" and similar expressions mean the actual knowledge of the persons identified on Section K of the Parent Disclosure Schedule, and such knowledge as such persons would reasonably be expected to have obtained, after reasonable inquiry, in the course of their performance of their positions at the Parent.

"Law" means any statute, law, ordinance, regulation, rule, code, order, constitution, treaty, common law, judgment, decree, other requirement or rule of law of any Governmental Authority.

"Net Working Capital" means as of the Closing Date, (i) all current assets of the Parent (including cash, whether held in account, escrow or otherwise) but excluding fifty percent (50%) of any cash paid to Parent in connection with the exercise of stock options between the signing of this Agreement and Closing, minus (ii) all current liabilities of the Parent (for the avoidance of doubt, including any accrued but unpaid Parent Transaction Expenses).

"Net Working Capital (Adjusted)" means Net Working Capital plus Parent Transaction Expenses. For clarity, Net Working Capital (Adjusted) is intended by both Parties to represent Net Working Capital as of the Closing Date, calculated on a proforma basis as if the Parent Transaction Expenses had not been incurred.

"New Share Amount" means the Parent Diluted Amount <u>divided by One Thousand Sixty-Six Ten-Thousandths (0.1066)</u>, and then reduced by the Parent Diluted Amount, and then rounded up to the next whole share. For example, if the Parent Diluted Amount is Three Million Eight Hundred Thousand (3,800,000), the New Share Amount is equal to 31,847,280 (i.e., 3,800,000 <u>divided by 0.1066</u>, and then reduced by 3,800,000, and rounded up), and the total outstanding shares is equal to 35,647,280 (Parent Diluted Amount <u>plus</u> the New Share Amount).

"Organizational Documents" means (a) in the case of a Person that is a corporation, its articles or certificate of incorporation and its by-laws, regulations or similar governing instruments required by the laws of its jurisdiction of formation or organization; (b) in the case of a Person that is a partnership, its articles or certificate of partnership, formation or association, and its partnership agreement (in each case, limited, limited liability, general or otherwise); (c) in the case of a Person that is a limited liability company, its articles or certificate of formation or organization, and its limited liability company agreement or operating agreement; and (d) in the case of a Person that is none of a corporation, partnership (limited, limited liability, general or otherwise), limited liability company or natural person, its governing instruments as required or contemplated by the laws of its jurisdiction of organization.

"Parent Diluted Amount" means the total number of outstanding shares of common stock of Parent, assuming full conversion, exercise and exchange of all Equity Interests of Parent, whether or not vested, in the money or subject to conditions to conversion, exercise and exchange. For the avoidance of doubt, the following shares shall not be considered outstanding

for purposes of determining the Parent Diluted Amount: (i) all shares reserved or approved under the New Equity Plan and (ii) any shares available under the ESOP that have not been promised, granted, awarded or otherwise committed to any recipient or potential recipient.

"Parent Fundamental Representations" means the representations and warranties set forth in Section 5.1 (Organization, Power and Enforceability); Section 5.2 (Authority); Section 5.3 (Capital Structure); and Section 5.7 (Brokers).

'Parent Material Adverse Effect" means any change, effect, event, circumstance or development that is materially adverse to or has a material adverse effect on (i) the business, assets, liabilities, capitalization, financial condition, or results of operations of Parent or Merger Sub, taken as a whole, or (ii) the ability of Parent or Merger Sub to consummate the Merger or any of the other Transactions or to perform any of their respective covenants or obligations under this Agreement; provided, however, that none of the following, to the extent arising after the date of this Agreement, either alone or in combination, shall be deemed to be a Parent Material Adverse Effect, and none of the following shall be taken into account in determining whether there has been or will be a Parent Material Adverse Effect: any change or event caused by or resulting from (A) changes in prevailing economic or market conditions in the United States or any other jurisdiction in which such entities have substantial business operations (except to the extent those changes have a disproportionate effect on Parent or Merger Sub relative to the other participants of similar size operating in the industry or industries in which Parent or Merger Sub Subsidiaries operate), (B) changes or events affecting the industry or industries in which Parent operates generally (except to the extent those changes or events have a disproportionate effect on Parent relative to the other participants of similar size operating in the industry or industries in which Parent and Merger Sub operate), (C) changes in generally accepted accounting principles or requirements (except to the extent those changes have a disproportionate effect on Parent, or Merger Sub relative to the other participants of similar size operating in the industry or industries in which Parent operates), (D) changes in laws, rules or regulations of general applicability or interpretations thereof by any Governmental Authority (except to the extent those changes have a disproportionate effect on Parent or Merger Sub relative to the other participants of similar size operating in the industry or industries in which Parent operates), (E) any natural disaster, pandemic or any outbreak of major hostilities in which the United States is involved or any act of terrorism within the United States or directed against its facilities or citizens wherever located or any governmental response to any of the foregoing (except to the extent those changes or events have a disproportionate effect on Parent or Merger Sub relative to the other participants of similar size operating in the industry or industries in which Parent operates), (F) a change in the public trading price of Parent Common Stock or the implications thereof, (G) a change in the trading volume of Parent Common Stock, (H) any failure by Parent to meet any public estimates or expectations of Parent's revenue, earnings or other financial performance or results of operations for any period, or (I) any failure by Parent to meet any guidance, budgets, plans or forecasts of its revenues, earnings, or other financial performance or results of operations (but not, in the case of Clauses (F) through (I), the underlying cause of such changes or failures, unless such changes or failures would otherwise be excepted from this definition). For the avoidance of doubt, the Parties agree that the terms "material," "materially" and "materiality" as used in this Agreement with an initial lower case "m" shall have their respective customary and ordinary meanings, without regard to the meanings ascribed to Parent Material Adverse Effect or

Company Material Adverse Effect, in each case as defined in this Agreement. Notwithstanding anything to the contrary in this Agreement, any event, fact, development, circumstance, change, effect, or occurrence to the extent it arises out of compliance by the Company with a Law, directive, guideline or recommendation promulgated by any Governmental Entity to respond to the impact, presence, outbreak or spread of COVID-19 shall not be deemed, either alone or in combination, to constitute or contribute to a Parent Material Adverse Effect (except to the extent those changes have a disproportionate effect on the Parent relative to the other participants of similar size operating in the industry or industries in which the Parent operates).

"Parent Shareholders Meeting" means the special meeting of the shareholders of Parent to be held to consider the approval of the proposals required by this Agreement to be sought at such meeting.

"Parent Transaction Expenses" means (i) all fees and expenses incurred by Parent at or prior to the Effective Time in connection with this Agreement and the transactions contemplated hereby, including without limitation any fees and expenses of legal counsel and accountants, the maximum amount of fees and expenses payable to financial advisors, investment bankers, brokers, consultants, and other advisors of Parent; (ii) any Taxes required to be paid by the Parent with respect to the Parent Stock Options; (iii) any amounts paid or payable for the D&O Tail Policy; and (iv) the Shared Expenses. Transaction Bonus Amounts are Indebtedness and not Parent Transaction Expenses.

"Payroll Tax Executive Order" means the Presidential Memorandum issued on August 8, 2020 directing the Secretary of the Treasury to use his authority pursuant to Section 7508A of the Code to defer the withholding, deposit, and payment of certain payroll tax obligations, as supplemented by Internal Revenue Service Notice 2020-65.

"Permits" means all permits, licenses, franchises, approvals, authorizations, registrations, certificates, variances and similar rights obtained, or required to be obtained, from Governmental Authorities.

"Person" means an individual, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association, or other entity.

"Qualified Person" means any person making an unsolicited, bona fide, written Acquisition Proposal that the Parent Board determines in good faith (after consultation with outside counsel and its financial advisors) is, or would reasonably be expected to lead to, a Superior Proposal, and such Acquisition Proposal has not resulted from a breach by Parent, of its obligations under Section 7.1.

"Real Property" means the real property leased or subleased by the Company or for which there is an easement benefitting the Company, together with all buildings, structures and facilities located thereon.

"Reincorporation and Charter Amendment" means (i) the amendment to the Parent's Certificate of Incorporation containing the material terms set forth on Exhibit C attached hereto,

including terms to (A) authorize the Reverse Split, (B) increase the quantity of post-Reverse Split shares of Parent Common Stock it is authorized to issue to a number of shares of Parent Common Stock equal to 30,000,000 and preferred stock equal to 5,000,000, and (C) change the name of the Parent to "Mobile X Global, Inc." followed by (ii) the conversion of Parent's corporate domicile from Minnesota to Delaware by the filing of a Certificate of Conversion from a Non-Delaware Corporation with the Delaware Secretary of State.

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended.

"Shared Expenses" means (A) the fees paid or payable to the SEC in connection with filing the Registration Statement, the Proxy Statement/Consent/Prospectus, the Form S-3 for the PIPE Investment and any amendments and supplements thereto with the SEC, and the printing and mailing costs fees payable to financial printers in connection the printing and distribution thereof; (B) fees paid or payable to consultants in connection with SEC or Nasdaq matters; (C) the Nasdaq Fees paid or payable; (D) the costs of any filing fees pursuant to the HSR Act or similar antitrust Law paid or payable; (E) the fees and expenses of the ESOP Fiduciary; and (F) the fees and expenses of the transfer agent and amounts required to be paid by the Parent for any fractional shares. For clarity, Shared Expenses are a subset of Parent Transaction Expenses.

"Subsidiary" means, with respect to any Party, any corporation, partnership, trust, limited liability company or other non-corporate business enterprise in which such Party (or another Subsidiary of such Party) owns or controls, directly or indirectly, securities or other ownership interests representing (A) more that 50% of the voting power of all outstanding stock or ownership interests of such entity or (B) the right to receive more than 50% of the net assets of such entity available for distribution to the holders of outstanding stock or ownership interests upon a liquidation or dissolution of such entity.

"Superior Proposal" means, with respect to Parent, any bona fide, unsolicited written Acquisition Proposal which did not result from a breach of or arise in connection with a breach of Section 7.1, made by a third party that if consummated would result in such third party's (or its stockholders') owning, directly or indirectly, 50% or more of the Equity Interests or total assets of the Parent or of the shares of the surviving entity in a merger or the direct or indirect parent of the surviving entity in a merger), pursuant to a tender or exchange offer, a merger, a consolidation, business combination or recapitalization or a sale or exclusive license of its assets, (a) on terms which Parent Board determines in its good faith judgment to be more favorable to the holders of the Parent's capital stock from a financial point of view than the Transactions (after consultation with its financial and legal advisors), taking into account all the terms and conditions of such proposal and this Agreement (including any termination or break-up fees and conditions to consummation, as well as any written, binding offer by the Company to amend the terms of this Agreement,) that the Parent Board, (b) is not subject to any financing condition, (c) is reasonably capable of being completed on the terms proposed without unreasonable delay and (d) includes termination rights no less favorable than the terms set forth in this Agreement, and in all respects from a third party capable of performing such terms.

"Target Working Capital" means an amount equal to \$12,118,116.

"Tax Return" means any return, declaration, report, claim for refund, information return, or statement or other document relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Taxes" means (a) all federal, state, local, foreign and other income, gross receipts, sales, use, production, ad valorem, transfer, franchise, registration, profits, license, lease, service use, withholding, payroll, employment, unemployment, estimated, excise, severance, environmental, stamp, occupation, premium, property (real or personal), real property gains, windfall profits, customs, duties or other taxes, fees, levies, assessments or governmental charges of any kind, (b) all interest, penalties, fines, additions to tax, or additional amounts imposed by any Governmental Authority in connection with any item described in clause (a) or for failure to file any Tax Return, (c) any successor or transferee liability in respect of any items described in clauses (a) or (b) or both, including pursuant to Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or foreign law), and (d) any amounts payable under any tax sharing agreement or other contractual arrangement.

"Transaction Litigation" means any Action commenced or threatened in writing against a Party to this Agreement or any of its Subsidiaries, its board, any committee of the board or any of a Party's directors, officers, agents or affiliates or otherwise relating to, involving or affecting such Party or any of its Subsidiaries, directors, officers, agents or affiliates, in each case in connection with, arising from or otherwise relating to or regarding the Transactions, including any Action alleging or asserting any misrepresentation or omission in the Proxy Statement/Consent/Prospectus or any other communications to the Parent stockholders, other than any Action among the Parties to this Agreement; provided that, for the avoidance of doubt, any Action involving or arising under the HSR Act or any foreign or other antitrust Law shall not be considered Transaction Litigation.

Other Capitalized Terms. The following terms shall have the meanings specified in the indicated section of this Agreement:

Ferm Sectit Agreement Pream Allocation Certificate 2.106 Alternative Acquisition Agreement 7.1(b Cash Dividend Recid Certificate of Merger 2.2 Certificates 2.106 Chosen Court 10.10 Closing 2.9 Closing Oompany Audited Financial Statements 7.26 Closing Date 2.9 Collosing Date 2.9 Company Element Recit Company Board Recit Company Board Recit Company Material Contract 4.26 Company Material Contract 4.26 Company Permitted Encumbrances 4.36 Company Stock Option 2.60(f) Company Stock Plans 3.34 Company Stock Plans 3.34 Company Stock Noder Approval 7.54 Company Stock Plans 7.16 D&O Tail Policy 7.16 D&O Tail Policy 7.16 D&C Testons 2.18 Exercited	mble (a) (b)(ii) citals (c) (c) (b) (d) (d) (d) (d) (d) (d) (d) (d) (d) (d

Term	Section
Merger Sub	Preamble
Minimum PIPE Investment Amount	Recitals
Nasdaq	Recitals
Nasdaq Fees	7.4
Nasdaq Listing Application	7.4
New Equity Plan	7.5(b)
Nonparty Affiliate	10.13
Outside Date	9.1(b)
Parent	Preamble
Parent 1997 Plan	2.16(a)
Parent 2013 Plan	2.16(a)
Parent Balance Sheet	5.6(b)
Parent Board	Recitals
Parent Board Recommendation	5.2(b)
Parent Board Recommendation Change	7.1(b)(i)
Parent Common Stock	Recitals
Parent Disclosure Schedule	1.3(a)
Parent Insurance Policies	5.21
Parent Material Contract	5.9(a)
Parent Owned Real Property	5.11(b)
Parent Permitted Encumbrances	5.11(a)
Parent Plan	5.18(a)
Parent Plans	5.18(a)
Parent SEC Reports	5.6(a)
Parent Shareholder Approval	7.5(b)
Parent Stock Issuance	Recitals
Parent Stock Options	2.16(a)
Parent Stock Plans	2.16(a)
Parent Termination Fee	9.3(b)(iii)
Parent Voting Agreement	Recitals
Parties	Preamble
Party	Preamble
PCÃOB	7.2(b)
PIPE Investment	Recitals
PIPE Investors	7.15(a)
Proskauer	10.14
Provisional Cash Dividend	Recitals
Proxy Statement/Consent/Prospectus	3.8(c)
Recommendation Change Notice	7.1(b)
Record Date	7.3(a)
Registration Statement	3.8(a)
Regulation M-A Filing	3.8(b)
Representatives	7.1(a)
Reverse Split	7.18
Subscription Agreement	Recitals
1 0	

electro220769 ex2-1.htm	electro220769_8k_FINAL 06/14/2022 04:19 PM	93 of 96

<u>Term</u>	Section
Surviving Company	2.1
Transactions	Recitals
Uncertificated Shares	2.10(c)
Voting Parent Debt	5.3(c)
Written Consent	7.5(a)

EXHIBIT C

FORM OF PARENT ARTICLES OF INCORPORATION

MATERIAL TERMS OF REINCORPORATION AND CHARTER AMENDMENT

The Reincorporation and Charter Amendment shall consist of the terms set forth in the definition thereof, as supplemented by the terms set forth below and such additional terms and charter provisions as the Company may request, subject to the Parent's approval, not to be unreasonably withheld, conditioned or delayed ("Additional Approved Terms"). As the Company may request in order to more efficiently consummate the Merger and comply with applicable law and Nasdaq rules, subject to Parent's approval, not to be unreasonably withheld, conditioned or delayed, the Reincorporation and Charter Amendment and related filings may be (i) divided into separate filings to be filed sequentially with the appropriate secretary of state, (ii) filed as amendments and/or as amendment and restatements, as appropriate and (iii) adjusted so that the conversion into a Delaware corporation occurs prior to the other actions contemplated by the Reincorporation and Charter Amendment. The provisions below are written to comply with the Delaware General Corporation Law ("DGCL"), however to the extent such provisions are included in a filing with the Minnesota Secretary of State, such provisions shall be adjusted solely to the extent necessary to conform with the requirements of the Minnesota Business Corporation Act, as amended ("MBCA"), except as otherwise reasonably agreed to by the Parties. It is understood that in connection with the PIPE Investment a Certificate of Designations is anticipated to be required, which would be filed in conjunction with the Reincorporation and Charter Amendment.

**

State of Incorporation: Minnesota (until the conversion of Parent's domicile from Minnesota to Delaware pursuant to clause (ii) of the Reincorporation and Charter Amendment).

Reverse Split: To have substantially the following terms (with the blanks to be filled in pursuant to the requirements of the Merger Agreement):

Effective as of 12:01 a.m., Eastern Time on [•] each [•] shares of the issued and outstanding shares of Common Stock of this Corporation shall thereby and thereupon automatically be combined into one (1) validly issued, fully paid and non-assessable share of Common Stock of this Corporation (the "Reverse Stock Split"), with a reduction in the number of authorized shares of this Corporation's common stock by a corresponding ratio. No scrip or fractional shares will be issued by reason of the Reverse Stock Split. In lieu thereof, cash shall be distributed to each stockholder of record of the Company who would otherwise have been entitled to receipt of a fractional share and the amount of cash to be distributed shall be based upon the average closing price of a share of Common Stock on The NASDAQ Capital Market for the five trading days immediately preceding the effective date of the Reverse Stock Split.

Effective Time Certificate of Incorporation of Parent:

Name of the Corporation/Article I: Mobile X Global, Inc.

 $Registered\ Office\ and\ Agent/Article\ II;$

The address of the Corporation's registered office in the State of Delaware is The Corporation Service Company, 251 Little Falls Drive, City of Wilmington, County of New Castle, State of Delaware 19808, and the name of its registered agent at such address is Corporation Service Company.

Purpose/Article III:

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the DGCL as it now exists or may hereafter be amended and supplemented.

Post-Reverse Split Capitalization/Article IV:

The Corporation is authorized to issue two classes of stock to be designated, respectively, "Common Stock" and "Preferred Stock." The total number of shares of capital stock which the Corporation shall have authority to issue is 35,000,000. The total number of shares of Common Stock that the Corporation is authorized to issue is 30,000,000, having a par value of \$0.0001 per share, and the total number of shares of Preferred Stock that the Corporation is authorized to issue is 5,000,000, having a par value of \$0.0001 per share.

The designations and the powers, privileges and rights, and the qualifications, limitations or restrictions thereof in respect of each class of capital stock of the Corporation are as follows:

A. COMMON STOCK.

- 1. <u>General</u>. The voting, dividend, liquidation, and other rights and powers of the Common Stock are subject to and qualified by the rights, powers and preferences of any series of Preferred Stock as may be designated by the Board of Directors of the Corporation (the "<u>Board of Directors</u>") and outstanding from time to time.
- 2. <u>Voting</u>. Except as otherwise provided herein or expressly required by law, each holder of Common Stock, as such, shall be entitled to vote on each matter submitted to a vote of stockholders and shall be entitled to one (1) vote for each share of Common Stock held of record by such holder as of the record date for determining stockholders entitled to vote on such matter. Except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this [Amended and Restated] Certificate (including any Certificate of Designation (as defined below)) that relates solely to the rights, powers, preferences (or the qualifications, limitations or restrictions thereof) or other terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Amended and Restated Certificate (including any Certificate of Designation) or pursuant to the DGCL.

Subject to the rights of any holders of any outstanding series of Preferred Stock, the number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

- 3. <u>Dividends</u>. Subject to applicable law and the rights and preferences of any holders of any outstanding series of Preferred Stock, the holders of Common Stock, as such, shall be entitled to the payment of dividends on the Common Stock when, as and if declared by the Board of Directors in accordance with applicable law.
- 4. <u>Liquidation</u>. Subject to the rights and preferences of any holders of any shares of any outstanding series of Preferred Stock, in the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, the funds and assets of the Corporation that may be legally distributed to the Corporation's stockholders shall be distributed among the holders of the then outstanding Common Stock *pro rata* in accordance with the number of shares of Common Stock held by each such holder.

B. PREFERRED STOCK

Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated or expressed herein and in the resolution or resolutions providing for the creation and issuance of such series adopted by the Board of Directors as hereinafter provided.

Authority is hereby expressly granted to the Board of Directors from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by adopting a resolution or resolutions providing for the issuance of the shares thereof and by filing a certificate of designation relating thereto in accordance with the DGCL (a "Certificate of Designation"), to determine and fix the number of shares of such series and such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including

without limitation thereof, dividend rights, conversion rights, redemption privileges and liquidation preferences, and to increase or decrease (but not below the number of shares of such series then outstanding) the number of shares of any series as shall be stated and expressed in such resolutions, all to the fullest extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation and issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law and this [Amended and Restated] Certificate (including any Certificate of Designation). Except as otherwise required by law, holders of any series of Preferred Stock shall be entitled only to such voting rights, if any, as shall expressly be granted thereto by this Amended and Restated Certificate (including any Certificate of Designation).

The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

In addition, the Reincorporation and Charter Amendment will include the Additional Approved Terms pertaining to the Board of Directors, Stockholders, Liability, Indemnification, Forum Selection and Amendments.

 electro220769_ex3-2.htm
 EX-3.2
 1 of 1

 Section 3.12 of Electro-Sensors, Inc. Bylaws, effective June 10, 2022
 electro220769_8k_FINAL
 06/14/2022 04:19 PM

Exhibit 3.2

Article 3 of the Electro-Sensors, Inc., Bylaws, Directors, is hereby amended to add a new Section 3.12, as forth below, effective June 10, 2022

Section 3.12. Exclusive Forum for Internal Corporate Claims.

The sole and exclusive forum for (i) any claim that is based upon a violation of a duty under the laws of the State of Minnesota by a current or former director, officer or shareholder in such capacity; (ii) any derivative action or proceeding brought on behalf of the corporation; or (iii) any action asserting a claim arising under any provision of the Minnesota Business Corporation Act or the corporation's articles or bylaws will be the federal courts (where jurisdiction exists) and state courts located in Hennepin County, Minnesota. Any person or entity purchasing or otherwise acquiring an interest in shares of capital stock of the corporation is deemed to have notice of and consented to the provisions of this bylaw.

electro220769_ex10-1.htm EX-10.1 1 of 11

Form of Voting Agreement dated June 10, 2022 among Mobile X Global, Inc, Electro-Sensors, Inc, and officers, directors and shareholders of Electro-Sensors, Inc. listed on Schedule 1-A

electro220769_8k_FINAL 06/14/2022 04:19 PM

Exhibit 10.1

Execution Version

VOTING AGREEMENT

This VOTING AGREEMENT (this "<u>Voting Agreement</u>") is made and entered into as of June 10, 2022 by and among Mobile X Global Inc., a Delaware corporation ("<u>Mobile X</u>"), the undersigned shareholder (the "<u>Shareholder</u>") of Electro-Sensors, Inc., a Minnesota corporation (the "<u>Parent</u>") and the Parent. Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

- A. Concurrently with the execution of this Voting Agreement, Mobile X, the Parent and Mobile X Newco, Inc., a Delaware corporation ("Merger Sub") are entering into an Agreement and Plan of Merger (as it may be amended, supplemented or otherwise modified from time to time, the "Merger Agreement") that provides, among other things, that the Merger Sub shall be merged with and into Mobile X and Mobile X will continue as the surviving corporation and as a wholly owned subsidiary of the Parent, in a transaction intended to qualify as a tax free reorganization under Sections 368 (a)(1)(A) or 368(a)(2)(E) of the Internal Revenue Code (the "Merger").
- B. The Shareholder is the owner of such number of outstanding shares of the Parent Common Stock indicated on the final page of this Voting Agreement and the Shareholder is also the owner of options to acquire such number of shares as is indicated on the final page of this Voting Agreement.
- C. As a material inducement to and condition precedent of Mobile X to enter into the Merger Agreement, Mobile X desires the Shareholder to agree, and the Shareholder is willing to agree, to vote the Shares (as defined below) so as to facilitate consummation of the Merger and to agree to the other matters set forth herein.

NOW, THEREFORE, in consideration of the foregoing and the mutual promises, covenants and conditions set forth herein, the parties hereto agree as follows:

1. AGREEMENT TO VOTE SHARES

- 1.1 <u>Definitions. For the purposes of this Voting Agreement:</u>
- (a) <u>Shares.</u> The term "Shares" means such number of shares of capital stock of the Parent, including without limitation shares of the Parent Common Stock, owned of record or beneficially by the Shareholder or over which the Shareholder exercises voting power as of the execution by the Shareholder of this Voting Agreement and all additional shares of capital stock of the Parent (including all such additional shares, the "Parent Common Stock") of which the Shareholder acquires beneficial ownership or voting power after the time that the Shareholder executes this Voting Agreement.

- (b) <u>Transfer.</u> The Shareholder shall be deemed to have effected a "Transfer" of a security if the Shareholder directly or indirectly: (i) sells, pledges, encumbers, transfers or disposes of, or grants an option with respect to, such security or any interest therein; or (ii) enters into an agreement or commitment providing for the sale, pledge, encumbrance, transfer or disposition of, or grant of an option with respect to, such security or any interest therein.
- (c) Other. The Shareholder is a "Parent Shareholder" within the meaning of the Merger Agreement. For purposes of the definition of Shares, the terms "beneficial owner" and "beneficial ownership" shall have the meaning given such terms in Section 302A.011 of the Minnesota Business Corporation Act ("MBCA").
- 1.2 Agreement to Vote Shares. The Shareholder hereby irrevocably and unconditionally covenants and agrees that, at all times prior to the termination of the Shareholder's obligations under this Section 1.2 pursuant to Section 3 hereof, at any and every meeting (whether annual or special and whether or not an adjourned or postponed meeting) of the shareholders of the Parent, however called, and on any action or approval of the Parent's shareholders by written consent, the Shareholder will appear at the meeting or otherwise cause the Shares to be counted as present thereat for purposes of establishing a quorum and vote (or cause to be voted, including via proxy), or with respect to any written consent, grant (or cause to be granted) such consent as to, all of the Shares:
- (a) in favor of the approval and adoption of the Merger Agreement and the Merger, the Transactions and the other actions contemplated by or in furtherance of the Merger Agreement;
- (b) against any Acquisition Proposals and against the approval of any proposal or action that could reasonably be expected to lead to an Acquisition Proposal; and
- (c) against any action or proposal that could reasonably be expected to adversely affect Mobile X's rights and benefits under the Merger Agreement.

The obligations of the Shareholder specified in this Section 1.2 shall (i) apply whether or not the Merger or any action described above is recommended by the Parent's board of directors or committee thereof but (ii) shall cease to apply after the termination of Shareholder's obligations under this Section 1.2 pursuant to Section 3 hereof. The Shareholder shall execute and deliver (or cause the holders of record to execute and deliver), and shall use commercially reasonable efforts to do the same within 48 hours of receipt thereof, any proxy card or voting instructions it receives that is sent to shareholder of the Parent soliciting proxies with respect to the proposals described in clauses (a), (b) and (c) of this Section 1.2 (the "Covered Proposals"), which shall be voted in the manner described in Section 1.2. This Voting Agreement is intended to bind the Shareholder as a shareholder of the Parent (and not in any other capacity such as a director or officer of the Parent or any of its Affiliates) and only with respect to the Covered Proposals. Except as provided for in this Section 1.2, the Shareholder shall not be restricted from voting in favor of, against or abstaining with respect to any other matter presented to the shareholders of the Parent. The Shareholder further agrees not to enter into any agreement or understanding with any person that would be inconsistent with or violate any provision in this

Section 1.2. Notwithstanding any other provision in this Voting Agreement to the contrary, in no event shall this Voting Agreement constitute an acknowledgement by Mobile X or Shareholder that the execution and delivery by Mobile X or the Shareholder of this Voting Agreement results in Mobile X acquiring beneficial ownership of any of the Shares.

1.3 Transfer and Other Restrictions.

- (a) Prior to the termination of the Shareholder's obligations under this Section 1.3 pursuant to Section 3 hereof, the Shareholder agrees not to, directly or indirectly (with any Transfer, attempted Transfer or action in violation of this Section 1.3(a) being null and void and of no effect whatsoever):
 - (i) offer for sale, Transfer or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to, or consent to the offer for sale, Transfer or other disposition of, any or all of the Shares or any interest therein except as provided in Section 1.5(a) below;
 - (ii) grant any proxy or power of attorney with respect to the Shares, deposit any of the Shares into a voting trust or enter into a voting agreement or arrangement with respect to the Shares other than in connection with this Voting Agreement; or
 - (iii) take any other action that would make any representation or warranty of the Shareholder contained herein untrue or incorrect or have the effect of preventing or disabling the Shareholder from performing his or its obligations under this Voting Agreement.
- (b) Notwithstanding anything in this Voting Agreement to the contrary, the Shareholder may Transfer all or a portion of the Shares to (i) any other Person who enters into a voting agreement (or similar Contract) with Mobile X on substantially similar terms and conditions as set forth in this Voting Agreement or to any Affiliate of such Person, (ii) any immediate family member (including a trust for such family member's benefit) of the Shareholder, (iii) if the Shareholder is an entity, any shareholder, member or partner of the Shareholder, (iv) any Person if and to the extent required by any order or decree by a Governmental Authority (including by divorce decree), or by will, intestacy or other similar Law or (v) any charitable foundation or organization, in each case, only so long as, prior to and as a condition to effectuating any such Transfer, the assignee or transferee agrees to be bound by the terms of this Voting Agreement and first executes and delivers to the parties hereto a written consent and joinder memorializing such agreement in form and substance reasonably satisfactory to Mobile X and Parent. During the term of this Voting Agreement, the Parent will not register or otherwise recognize the transfer (book-entry or otherwise) of any Shares or any certificate or uncertificated interest representing any of the Shares, except as permitted by, and in accordance with, this Section 1.3(b), receipt of written consent by the parties or as, in the reasonable determination of the Parent, required by law.
- (c) To the extent the Shareholder is, as of the date hereof, party to a contract or agreement that requires the Shareholder to Transfer Shares to another person or

electro220769 ex10-1.htm

4 of 11

entity, the Shareholder will not effect any such Transfer unless and until the transferee agrees to be bound by and executes an agreement in the form of this Voting Agreement with respect to the Shares to be Transferred. Nothing herein shall prohibit the Shareholder from exercising any option the Shareholder may hold in accordance with the terms of the option, provided that the securities acquired upon the exercise will become subject to this Voting Agreement.

1.4 <u>Waivers of Certain Rights and Actions.</u> Shareholder hereby agrees not to commence or participate in, and to take all actions necessary to opt out of any class in any class action with respect to, any Action, derivative or otherwise, against the Parent, Mobile X, or any of their respective Subsidiaries or successors: (a) challenging the validity of, or seeking to enjoin or delay the operation of, any provision of this Agreement or the Merger Agreement (including any claim seeking to enjoin or delay the Closing); or (b) to the fullest extent permitted under Law, alleging a breach of any duty of the Board of Directors of the Parent or Mobile X in connection with the Merger Agreement, this Agreement, or the transactions contemplated thereby or hereby.

1.5 Other Agreements.

- (a) Upon the request of Mobile X, the Shareholder agrees to notify promptly Mobile X of the number of any additional shares of the Parent Common Stock acquired by the Shareholder, if any, after the execution of this Voting Agreement.
- (b) The Shareholder agrees that any shares of Parent Common Stock that the Shareholder purchases or with respect to which the Shareholder otherwise acquires record or beneficial ownership (including (i) any shares of Parent Common Stock that the Shareholder acquires pursuant to the exercise or settlement of any Parent Equity Awards (but excluding shares of Parent Common Stock underlying unexercised Parent Equity Awards (until such time as any such Parent Equity Awards are exercised and the underlying shares of Parent Common Stock are acquired by the Shareholder)) or (ii) pursuant to a stock split, reverse stock split, stock dividend or distribution or any change in Parent Common Stock by reason of any recapitalization, reorganization, combination, reclassification, exchange of shares or similar transaction) after the date hereof and prior to (and until) the termination of this Agreement in accordance with Section 3, shall automatically become, and shall be deemed to be, Shares and will thereafter be subject to the terms and conditions of this Agreement to the same extent as if they comprised Shares on the date hereof.
- (c) The Shareholder further agrees that, from and after the date hereof until the termination of this Voting Agreement in accordance with Section 3, the Shareholder will not, and will not cause any entity under the Shareholder's control to, (i) solicit proxies or become a "participant" in a "solicitation" (as such terms are defined in Rule 14A under the Exchange Act) in opposition to any Covered Proposal, (ii) initiate a stockholders' vote with respect to an Acquisition Proposal or related transaction, (iii) become a member of a "group" (as such term is used in Section 13 (d) of the Exchange Act) with respect to any voting securities of the Parent with respect to an Acquisition Proposal or related transaction, or (iv) take any action that the Parent is prohibited from taking pursuant to the Merger Agreement, subject in each case to Shareholder's right (but without affecting Shareholder's obligations under this Voting Agreement in Shareholder's capacity as a Shareholder) to act in his or her capacity as a director

or officer of the Parent and comply with his or her fiduciary duties or other legal obligations or responsibilities while acting in such capacity as a director or officer of the Parent.

- (d) Shareholder agrees that during the term of this Voting Agreement Shareholder will not, and will not permit any entity under Shareholder's control to, deposit any of the Shares in a voting trust, grant any proxies with respect to the Shares, or subject any of the Shares to any arrangement with respect to the voting of the Shares other than agreements entered into with Mobile X.
- (e) For purposes of clarity, nothing in this Voting Agreement shall in any way prevent or preclude the Shareholder from taking any action or exercising any and all of its rights with respect to the Shares, or any other shares of Common Stock of the Parent owned or controlled by the Shareholder, in connection with an Acquisition Proposal with a third party to the extent that the same shall not result in, and was not procured by, in connection with or as a result of, a breach of this Voting Agreement.
- 1.6 Proxy. Solely in the event of a failure by the Shareholder to act in accordance with such Shareholder's obligations as to voting pursuant to Section 1.2, such Shareholder hereby irrevocably grants to and appoints Mobile X (and any designee thereof) as such Shareholder's proxy and attorney-in-fact (with full power of substitution), for and in the name, place and stead of the Shareholder, to represent, vote and otherwise act (by voting at any meeting of shareholders of the Parent or otherwise) with respect to such Shareholder's Shares solely as and to the extent set forth in this Section 1.6 until the termination of this Voting Agreement in accordance with its provisions, to the same extent and with the same effect as the Shareholder might or could do under applicable law, rules and regulations. The proxy granted pursuant to this Section 1.6 is coupled with an interest and is irrevocable. The Shareholder will take such further action and will execute such other instruments as may be necessary to effectuate the grant of this proxy. The power of attorney granted by Shareholder herein (if Shareholder is a natural person) is a durable power of attorney and shall survive the bankruptcy, death or incapacity of such Shareholder. Notwithstanding the foregoing, this proxy shall terminate upon termination of this Voting Agreement in accordance with Section 3.

2. REPRESENTATIONS AND WARRANTIES OF THE SHAREHOLDER

- Ownership. The Shareholder is the record and beneficial owner of, and the Shareholder exercises exclusive voting power over, the number of shares of the Parent Common Stock indicated on the final page of this Voting Agreement, which are free and clear of any Encumbrances (other than those created by this Agreement and those arising under applicable United States securities laws), proxy, voting restriction or adverse claim. The number of shares of Parent Common Stock set forth on the final page hereto are the only shares of capital stock of the Parent owned of record or beneficially by the Shareholder and, except as set forth on such page, the Shareholder holds no options to purchase or rights to subscribe for or otherwise acquire any securities of the Parent and has no other interest in or voting rights with respect to any securities of the Parent.
- 2.2 <u>Power and Authority; No Conflict.</u> The Shareholder has the requisite power and authority (or, if a natural person, the requisite power and capacity) to enter into this

Voting Agreement and to consummate the transactions contemplated by this Voting Agreement, The execution and delivery of this Voting Agreement by the Shareholder and the consummation by the Shareholder of the transactions contemplated by this Voting Agreement have been duly authorized by all necessary action. This Voting Agreement has been duly executed and delivered by the Shareholder and constitutes a valid and binding obligation of the Shareholder, enforceable against the Shareholder in accordance with its terms, except (i) as the same may be limited by applicable bankruptcy, insolvency, moratorium or similar laws of general application relating to or affecting creditors' rights, and (ii) for the limitations imposed by general principles of equity. The execution and delivery of this Voting Agreement does not, and the consummation of the transactions contemplated by this Voting Agreement and compliance with the provisions of this Voting Agreement will not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation that would result in the creation of any Encumbrance upon any of the Shares owned by the Shareholder under, any provision of applicable law or regulation or of any agreement, judgment, injunction, order, decree or other instrument binding on the Shareholder or any Shares owned by the Shareholder. The execution and delivery of this Voting Agreement by the Shareholder do not, and the performance of this Voting Agreement by the Shareholder will not, require any written, oral or other agreement, contract or legally binding commitment of any third party. If this Voting Agreement is being executed in a representative or fiduciary capacity, the person signing this Voting Agreement has full power and authority to enter into and perform this Voting Agreement. No consent, approval, order or authorization of, or registration, declaration or, except as required by the rules and regulations promulgated under the Exchange Act, filing with, any Governmental Authority or any other Person, is required by or with respect to the Shareholder in connection with the execution and delivery of this Voting Agreement or the consummation by them of the transactions contemplated hereby. As of the date hereof, there is no legal action pending against, or, to the knowledge of the Shareholder, threatened against or affecting the Shareholder that would reasonably be expected to materially impair the ability of the Shareholder to perform its obligations hereunder or to consummate the transactions contemplated hereby on a timely basis.

3. TERMINATION OF OBLIGATIONS

The obligations of the Shareholder pursuant to Sections 1.2, 1.3 and 1.5 hereof shall terminate upon the earliest of (i) the Effective Time and (ii) the date and time of the termination of the Merger Agreement pursuant to Article IX thereof; provided that the provisions of Section 4 shall survive any such termination. Notwithstanding the foregoing, termination of this Voting Agreement shall not prevent any party from seeking any remedies (at law or in equity) against any other party for that party's breach of any of the terms of this Voting Agreement prior to the date of termination in accordance with Section 3. In addition, this Voting Agreement may be terminated in its entirety by Mobile X by providing written notice of such termination to the Parent and the Shareholder, and such termination shall be effective immediately upon delivery of such notice.

4. MISCELLANEOUS

4.1 <u>Severability</u>. If any term, provision, covenant or restriction of this Voting Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, then

the remainder of the terms, provisions, covenants and restrictions of this Voting Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

- 4.2 <u>Binding Effect and Assignment</u>. This Voting Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but, except as otherwise specifically provided herein, neither this Voting Agreement nor any of the rights, interests or obligations of the parties hereto may be assigned by either of the parties without prior written consent of the other. Any purported assignment in violation of this Section 4.2 shall be void.
- 4.3 <u>Amendments and Modification</u>. This Voting Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by the parties hereto.
- 4.4 <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be deemed duly given upon delivery either by commercial delivery service, or sent via facsimile (receipt confirmed), to the parties at the following address or facsimile numbers (or at such other address or facsimile numbers for a party as shall be specified by like notice):

If to Mobile X:

Mobile X Global, Inc. Attention: Peter Adderton, Chief Executive Officer; Justin O'Neill, General Counsel 12 Sail Vista Newport Coast, CA 92657 Email:

with a copy to:

Proskauer Rose LLP Attention: Ben D. Orlanski 2029 Century Park East, Suite 240 Los Angeles, CA 90067 Email:

If to the Shareholder, to the address for notice set forth on the last page hereof, with a copy to Parent at Electro-Sensors, Inc., c/o Ballard Spahr LLP, 2000 IDS Center, 80 South Eighth Street, Minneapolis, MN 55402, Attention: George H. Singer, Email:

Any party hereto may by notice so given provide and change its address for future notices hereunder. Notice shall conclusively be deemed to have been given when personally delivered or when deposited in the mail in the manner set forth above.

- Governing Law; Disputes; Submission to Jurisdiction; Waiver of Jury Trial. This Voting Agreement shall be governed by and construed exclusively in accordance with the laws of the State of Minnesota. Each of the parties to this Voting Agreement (a) consents to submit itself to the exclusive personal jurisdiction of the state courts of the State of Minnesota or, to the extent such court does not have subject matter jurisdiction, the United States District Court for the District of Minnesota (as applicable, the "Chosen Court") in any action or proceeding arising out of or relating to this Voting Agreement or any of the Transactions, (b) agrees that all claims in respect of such action or proceeding shall be heard and determined in any such Chosen Court, (c) agrees that it shall not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such Chosen Court and (d) agrees not to bring any action or proceeding arising out of or relating to this Voting Agreement or any of the transaction contemplated by this Voting Agreement in any other court. Each of the parties hereto waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of any other party with respect thereto. Any party may make service on another party by sending or delivering a copy of the process to the party to be served at the address and in the manner provided for the giving of notices in Section 4.4. Nothing in this Section 4.5, however, shall affect the right of any party to serve legal process in any other manner permitted by law. EACH OF SHAREHOLDER, PARENT AND MOBILE X HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS VOTING AGREEMENT OR THE TRANSACTIONS OR THE ACTIONS OF SHAREHOLDER, PARENT OR MOBILE X IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS VOTING AGREEMENT.
- 4.6 <u>Publicity.</u> The Shareholder acknowledges and represents that it supports the Merger pursuant to the terms of the Merger Agreement and authorizes each of the Parent and Mobile X to make a general statement to that effect (including Shareholder's identity and holding of Shares and the related obligations of Shareholder hereunder) in press releases and public filings that may be made with the SEC, Nasdaq or other regulatory authorities.
- 4.7 Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement was not performed in accordance with its specific terms or was otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Minnesota court, without proof of damages, prior to the valid termination of this Voting Agreement, this being in addition to any other remedy to which they are entitled under this Voting Agreement, and (b) the right of specific enforcement is an integral part of the transactions contemplated by this Voting Agreement and without that right, none of the parties would have entered into this Voting Agreement. Each party agrees that it will not oppose the granting of specific performance and other equitable relief on the basis that the other parties have an adequate remedy at Law or that an award of specific performance is not an appropriate remedy for any reason at Law or equity. The parties acknowledge and agree that any party seeking an injunction to prevent breaches of this Voting Agreement and to enforce specifically the terms and provisions of this voting

Agreement in accordance with this Section 4.7 shall not be required to provide any bond or other security in connection with any such injunction.

- 4.8 <u>Entire Agreement</u>. This Voting Agreement constitutes and contains the entire agreement and understanding of the parties with respect to the subject matter hereof and supersedes any and all prior negotiations, correspondence, agreements, understandings, duties or obligations between the parties respecting the subject matter hereof.
- 4.9 <u>Counterparts.</u> This Voting Agreement may be executed in facsimile, pdf or other electronic means and in or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 4.10 <u>Further Assurances</u>. Shareholder agrees, from time to time, and without additional consideration, to execute and deliver such additional proxies, documents, and other instruments and to take all such further action as the Parent and Mobile X may reasonably request to consummate and make effective the transactions contemplated by this Voting Agreement.
- 4.11 No Agreement Until Executed. Irrespective of negotiations among the parties or the exchanging of drafts of this Voting Agreement, this Voting Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties unless and until (a) the Parent Board has approved, for purposes of any applicable anti-takeover laws and regulations, and any applicable provision of the Parent's organizational and governance documents, the transactions contemplated by the Merger Agreement (including the Merger), (b) the Merger Agreement is executed by all parties thereto, and (c) this Voting Agreement is executed by all parties hereto.
- 4.12 <u>Captions</u>. The captions to sections of this Voting Agreement have been inserted for identification and reference purposes only and shall not be used to construe or interpret this Voting Agreement.

electro220769	8k	FINAL	06/14/2022	04:19 PM

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed by their duly authorized respective officers as of the date first above written.

MOBILE X GLOBAL, INC.

By: /s/ Dan Cohrs
Name: Dan Cohrs
Title: Chief Financial Officer

ELECTRO-SENSORS, INC.

By: /s/ David L. Klenk
Name: David L. Klenk
Title: Chief Executive Officer

THE SHAREHOLDER:

the Shareholder's Address for Notice:

Schedule 1-A

Printed Name: _

Outstanding Shares of the Parent Common Stock Beneficially Owned by the Shareholder:_____

Signature: /s/ signed by each officer, director and shareholder listed on

Outstanding Options to Acquire Shares of Parent Common Stock Owned by the Shareholder:_____

Schedule 1-A to Parent Voting Agreement

Name	Shares	Options	ESOP
John Peterson	350,893		
Jeffrey Peterson	350,893		48
Lynne Peterson	350,893	50,000	
Patricia Peterson	350,893		
Paul Peterson	350,895		
David L. Klenk		100,000	2,979
Joseph A. Martino		52,500	
Scott A. Gabbard		50,000	
Michael Zipoy	10,000	52,500	

electro220769 ex10-2.htm EX-10.2 1 of 11 06/14/2022 04:19 PM

Form of Mobile X Voting Agreement dated June 10, 2022 among Mobile X Ğlobal, Inc, Electro-Sensors. Inc., and the officers, directors and Shareholders of Mobile X Global, Inc. named on Schedule I-B

electro220769 8k FINAL

Exhibit 10.2

Execution Version

VOTING AGREEMENT

This VOTING AGREEMENT (this "Voting Agreement") is made and entered into as of June 10, 2022 by and among Mobile X Global Inc., a Delaware corporation ("Mobile X"), the undersigned shareholder (the "Shareholder") of Mobile X and Electro-Sensors, Inc., a Minnesota corporation (the "Parent"). Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

- Concurrently with the execution of this Voting Agreement, Mobile X, the Parent and Mobile X Newco, Inc., a Delaware corporation ("Merger Sub") are entering into an Agreement and Plan of Merger (as it may be amended, supplemented or otherwise modified from time to time, the "Merger Agreement") that provides, among other things, that the Merger Sub shall be merged with and into Mobile X and Mobile X will continue as the surviving corporation and as a wholly owned subsidiary of the Parent, in a transaction intended to qualify as a tax free reorganization under Sections 368 (a)(1)(A) or 368(a)(2)(E) of the Internal Revenue Code (the "Merger").
- The Shareholder is the owner of such number of outstanding shares of the MX Common Stock indicated on the final page of this Voting Agreement and (if applicable) the Shareholder is also the owner of options to acquire such number of shares as is indicated on the final page of this Voting Agreement.
- As a material inducement to and condition precedent of Parent to enter into the Merger Agreement, Parent desires the Shareholder to agree, and the Shareholder is willing to agree, to vote the Shares (as defined below) so as to facilitate consummation of the Merger and to agree to the other matters set forth herein.

NOW, THEREFORE, in consideration of the foregoing and the mutual promises, covenants and conditions set forth herein, the parties hereto agree as follows:

AGREEMENT TO VOTE SHARES 1.

- 1.1 Definitions. For the purposes of this Voting Agreement:
- Shares. The term "Shares" means such number of shares of capital stock of the Mobile X, including without limitation shares of the MX Common Stock, owned of record or beneficially by the Shareholder or over which the Shareholder exercises voting power as of the execution by the Shareholder of this Voting Agreement and all additional shares of capital stock of the Mobile X (including all such additional shares, the "MX Common Stock") of which the Shareholder acquires beneficial ownership or voting power after the time that the Shareholder executes this Voting Agreement.

- (b) <u>Transfer</u>. The Shareholder shall be deemed to have effected a "Transfer" of a security if the Shareholder directly or indirectly: (i) sells, pledges, encumbers, transfers or disposes of, or grants an option with respect to, such security or any interest therein; or (ii) enters into an agreement or commitment providing for the sale, pledge, encumbrance, transfer or disposition of, or grant of an option with respect to, such security or any interest therein.
 - (c) Other. The Shareholder is a "Company Stockholder" within the meaning of the Merger Agreement.
- 1.2 Agreement to Vote Shares. The Shareholder hereby irrevocably and unconditionally covenants and agrees that, at all times prior to the termination of the Shareholder's obligations under this Section 1.2 pursuant to Section 3 hereof, at any and every meeting (whether annual or special and whether or not an adjourned or postponed meeting) of the shareholders of the Mobile X, however called, and on any action or approval of the Mobile X's shareholders by written consent, the Shareholder will appear at the meeting or otherwise cause the Shares to be counted as present thereat for purposes of establishing a quorum and vote (or cause to be voted, including via proxy), or with respect to any written consent, grant (or cause to be granted) such consent as to, all of the Shares:
- (a) in favor of the approval and adoption of the Merger Agreement and the Merger, the Transactions and the other actions contemplated by or in furtherance of the Merger Agreement; and
- (b) against any action or proposal that could reasonably be expected to adversely affect Parent's rights and benefits under the Merger Agreement.

The obligations of the Shareholder specified in this Section 1.2 shall (i) apply whether or not the Merger or any action described above is recommended by the Mobile X's board of directors or committee thereof but (ii) shall cease to apply after the termination of Shareholder's obligations under this Section 1.2 pursuant to Section 3 hereof. The Shareholder shall execute and deliver (or cause the holders of record to execute and deliver), and shall use commercially reasonable efforts to do the same within 48 hours of receipt thereof, any proxy card or voting instructions it receives that is sent to shareholder of the Mobile X soliciting proxies with respect to the proposals described in clauses (a), (b) and (c) of this Section 1.2 (the "Covered Proposals"), which shall be voted in the manner described in Section 1.2. This Voting Agreement is intended to bind the Shareholder as a shareholder of the Mobile X (and not in any other capacity such as a director or officer of the Mobile X or any of its Affiliates) and only with respect to the Covered Proposals. Except as provided for in this Section 1.2, the Shareholder shall not be restricted from voting in favor of, against or abstaining with respect to any other matter presented to the shareholders of the Mobile X. The Shareholder further agrees not to enter into any agreement or understanding with any person that would be inconsistent with or violate any provision in this Section 1.2. Notwithstanding any other provision in this Voting Agreement to the contrary, in no event shall this Voting Agreement constitute an acknowledgement by Parent or Shareholder that the execution and delivery by Parent or the Shareholder of this Voting Agreement results in Parent acquiring beneficial ownership of any of the Shares.

3 of 11

electro220769_ex10-2.htm

1.3 Transfer and Other Restrictions.

- (a) Prior to the termination of the Shareholder's obligations under this Section 1.3 pursuant to Section 3 hereof, the Shareholder agrees not to, directly or indirectly (with any Transfer, attempted Transfer or action in violation of this Section 1.3(a) being null and void and of no effect whatsoever):
 - (i) offer for sale, Transfer or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to, or consent to the offer for sale, Transfer or other disposition of, any or all of the Shares or any interest therein except as provided in Section 1.5(a) below;
 - (ii) grant any proxy or power of attorney with respect to the Shares, deposit any of the Shares into a voting trust or enter into a voting agreement or arrangement with respect to the Shares other than in connection with this Voting Agreement; or
 - (iii) take any other action that would make any representation or warranty of the Shareholder contained herein untrue or incorrect or have the effect of preventing or disabling the Shareholder from performing his or its obligations under this Voting Agreement.
- (b) Notwithstanding anything in this Voting Agreement to the contrary, the Shareholder may Transfer all or a portion of the Shares to (i) any other Person who enters into a voting agreement (or similar Contract) with Parent on substantially similar terms and conditions as set forth in this Voting Agreement or to any Affiliate of such Person, (ii) any immediate family member (including a trust for such family member's benefit) of the Shareholder, (iii) if the Shareholder is an entity, any shareholder, member or partner of the Shareholder, (iv) any Person if and to the extent required by any order or decree by a Governmental Authority (including by divorce decree), or by will, intestacy or other similar Law or (v) any charitable foundation or organization, in each case, only so long as, prior to and as a condition to effectuating any such Transfer, the assignee or transferee agrees to be bound by the terms of this Voting Agreement and first executes and delivers to the parties hereto a written consent and joinder memorializing such agreement in form and substance reasonably satisfactory to Mobile X and Parent. During the term of this Voting Agreement, the Mobile X will not register or otherwise recognize the transfer (book-entry or otherwise) of any Shares or any certificated interest representing any of the Shares, except as permitted by, and in accordance with, this Section 1.3(b), receipt of written consent by the parties or as, in the reasonable determination of the Mobile X, required by law.
- (c) To the extent the Shareholder is, as of the date hereof, party to a contract or agreement that requires the Shareholder to Transfer Shares to another person or entity, the Shareholder will not effect any such Transfer unless and until the transferee agrees to be bound by and executes an agreement in the form of this Voting Agreement with respect to the Shares to be Transferred. Nothing herein shall prohibit the Shareholder from exercising any option the Shareholder may hold in accordance with the terms of the option, provided that the securities acquired upon the exercise will become subject to this Voting Agreement.

4 of 11

1.5 Other Agreements.

electro220769 ex10-2.htm

- (a) Upon the request of Parent, the Shareholder agrees to notify promptly Parent of the number of any additional shares of the MX Common Stock acquired by the Shareholder, if any, after the execution of this Voting Agreement.
- (b) The Shareholder agrees that any shares of MX Common Stock that the Shareholder purchases or with respect to which the Shareholder otherwise acquires record or beneficial ownership (including (i) any shares of MX Common Stock that the Shareholder acquires pursuant to the exercise or settlement of any Company Stock Options (but excluding shares of MX Common Stock underlying unexercised Company Stock Options (until such time as any such Company Stock Options are exercised and the underlying shares of MX Common Stock are acquired by the Shareholder)) or (ii) pursuant to a stock split, reverse stock split, stock dividend or distribution or any change in MX Common Stock by reason of any recapitalization, reorganization, combination, reclassification, exchange of shares or similar transaction) after the date hereof and prior to (and until) the termination of this Agreement in accordance with Section 3, shall automatically become, and shall be deemed to be, Shares and will thereafter be subject to the terms and conditions of this Agreement to the same extent as if they comprised Shares on the date hereof.
- (c) The Shareholder further agrees that, from and after the date hereof until the termination of this Voting Agreement in accordance with Section 3, the Shareholder will not, and will not cause any entity under the Shareholder's control to, (i) solicit proxies or become a "participant" in a "solicitation" (as such terms are defined in Rule 14A under the Exchange Act) in opposition to any Covered Proposal, or (ii) take any action that the Mobile X is prohibited from taking pursuant to the Merger Agreement, subject in each case to Shareholder's right (but without affecting Shareholder's obligations under this Voting Agreement in Shareholder's capacity as a Shareholder) to act in his or her capacity as a director or officer of the Mobile X and comply with his or her fiduciary duties or other legal obligations or responsibilities while acting in such capacity as a director or officer of the Mobile X.
- (d) Shareholder agrees that during the term of this Voting Agreement Shareholder will not, and will not permit any entity under Shareholder's control to, deposit any of the Shares in a voting trust, grant any proxies with respect to the Shares, or subject any of the Shares to any arrangement with respect to the voting of the Shares other than agreements entered into with Mobile X.

1.6 Proxy. Solely in the event of a failure by the Shareholder to act in accordance with such Shareholder's obligations as to voting pursuant to Section 1.2, such Shareholder hereby irrevocably grants to and appoints Parent (and any designee thereof) as such Shareholder's proxy and attorney-in-fact (with full power of substitution), for and in the name, place and stead of the Shareholder, to represent, vote and otherwise act (by voting at any meeting of shareholders of the Mobile X or otherwise) with respect to such Shareholder's Shares solely as and to the extent set forth in this Section 1.6 until the termination of this Voting Agreement in accordance with its provisions, to the same extent and with the same effect as the Shareholder might or could do under applicable law, rules and regulations. The proxy granted pursuant to this Section 1.6 is coupled with an interest and is irrevocable. The Shareholder will take such further action and will execute such other instruments as may be necessary to effectuate the grant of this proxy. The power of attorney granted by Shareholder herein (if Shareholder is a natural person) is a durable power of attorney and shall survive the bankruptcy, death or incapacity of such Shareholder. Notwithstanding the foregoing, this proxy shall terminate upon termination of this Voting Agreement in accordance with Section 3.

2. REPRESENTATIONS AND WARRANTIES OF THE SHAREHOLDER

- Ownership. The Shareholder is the record and beneficial owner of, and the Shareholder exercises exclusive voting power over, the number of shares of the MX Common Stock indicated on the final page of this Voting Agreement, which are free and clear of any Encumbrances (other than those created by this Agreement and those arising under applicable United States securities laws), proxy, voting restriction or adverse claim. The number of shares of MX Common Stock set forth on the final page hereto are the only shares of capital stock of the Mobile X owned of record or beneficially by the Shareholder and, except as set forth on such page, the Shareholder holds no options to purchase or rights to subscribe for or otherwise acquire any securities of the Mobile X and has no other interest in or voting rights with respect to any securities of the Mobile X.
- 2.2 <u>Power and Authority; No Conflict.</u> The Shareholder has the requisite power and authority (or, if a natural person, the requisite power and capacity) to enter into this Voting Agreement and to consummate the transactions contemplated by this Voting Agreement. The execution and delivery of this Voting Agreement by the Shareholder and the consummation by the Shareholder of the transactions contemplated by this Voting Agreement have been duly authorized by all necessary action. This Voting Agreement has been duly executed and delivered by the Shareholder and constitutes a valid and binding obligation of the Shareholder, enforceable against the Shareholder in accordance with its terms, except (i) as the same may be limited by applicable bankruptcy, insolvency, moratorium or similar laws of general application relating to or affecting creditors' rights, and (ii) for the limitations imposed by general principles of equity. The execution and delivery of this Voting Agreement does not, and the consummation of the transactions contemplated by this Voting Agreement and compliance with the provisions of this Voting Agreement will not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation that would result in the creation of any Encumbrance upon any of the Shares owned by the Shareholder under, any provision of applicable law or regulation or of any agreement, judgment, injunction, order, decree or other instrument binding on the Shareholder or any Shares owned by the Shareholder. The execution and delivery of this Voting

6 of 11

Agreement by the Shareholder do not, and the performance of this Voting Agreement by the Shareholder will not, require any written, oral or other

agreement, contract or legally binding commitment of any third party. If this Voting Agreement is being executed in a representative or fiduciary capacity, the person signing this Voting Agreement has full power and authority to enter into and perform this Voting Agreement. No consent, approval, order or authorization of, or registration, declaration or, except as required by the rules and regulations promulgated under the Exchange Act, filing with, any Governmental Authority or any other Person, is required by or with respect to the Shareholder in connection with the execution and delivery of this Voting Agreement or the consummation by them of the transactions contemplated hereby. As of the date hereof, there is no legal action pending against, or, to the knowledge of the Shareholder, threatened against or affecting the Shareholder that would reasonably be expected to materially impair the ability of the Shareholder to perform its obligations hereunder or to consummate the transactions contemplated hereby on a timely basis.

3. TERMINATION OF OBLIGATIONS

The obligations of the Shareholder pursuant to Sections 1.2, 1.3 and 1.5 hereof shall terminate upon the earliest of (i) the Effective Time and (ii) the date and time of the termination of the Merger Agreement pursuant to Article IX thereof; provided that the provisions of Section 4 shall survive any such termination. Notwithstanding the foregoing, termination of this Voting Agreement shall not prevent any party from seeking any remedies (at law or in equity) against any other party for that party's breach of any of the terms of this Voting Agreement prior to the date of termination in accordance with Section 3. In addition, this Voting Agreement may be terminated in its entirety by Parent by providing written notice of such termination to Mobile X and the Shareholder, and such termination shall be effective immediately upon delivery of such notice.

4. MISCELLANEOUS

electro220769 ex10-2.htm

- 4.1 <u>Severability.</u> If any term, provision, covenant or restriction of this Voting Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, then the remainder of the terms, provisions, covenants and restrictions of this Voting Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.
- 4.2 <u>Binding Effect and Assignment.</u> This Voting Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but, except as otherwise specifically provided herein, neither this Voting Agreement nor any of the rights, interests or obligations of the parties hereto may be assigned by either of the parties without prior written consent of the other. Any purported assignment in violation of this Section 4.2 shall be void.
- 4.3 <u>Amendments and Modification.</u> This Voting Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by the parties hereto.
- 4.4 <u>Notices.</u> All notices and other communications hereunder shall be in writing and shall be deemed duly given upon delivery either by commercial delivery service, or

sent via facsimile (receipt confirmed), to the parties at the following address or facsimile numbers (or at such other address or facsimile numbers for a party as shall be specified by like notice):

If to Mobile X or any Shareholder:

Mobile X Global, Inc. Attention: Peter Adderton, Chief Executive Officer; Justin O'Neill, General Counsel 12 Sail Vista Newport Coast, CA 92657 Email:

with a copy to:

Proskauer Rose LLP Attention: Ben D. Orlanski 2029 Century Park East, Suite 240 Los Angeles, CA 90067 Email:

If to the Parent, to Electro-Sensors, Inc., c/o Ballard Spahr LLP, 2000 IDS Center, 80 South Eighth Street, Minneapolis, MN 55402, Attention: George H. Singer, Email:

Any party hereto may by notice so given provide and change its address for future notices hereunder. Notice shall conclusively be deemed to have been given when personally delivered or when deposited in the mail in the manner set forth above.

4.5 Governing Law; Disputes; Submission to Jurisdiction; Waiver of Jury Trial. This Voting Agreement shall be governed by and construed exclusively in accordance with the laws of the State of Minnesota, except that the General Corporation Law of the State of Delaware and the case law interpreting such law shall govern all aspects of this Voting Agreement that are essentially part of, derive from or are dependent upon the internal affairs of Mobile X. Each of the parties to this Voting Agreement (a) consents to submit itself to the exclusive personal jurisdiction of the state courts of the State of Minnesota or, to the extent such court does not have subject matter jurisdiction, the United States District Court for the District of Minnesota (as applicable, the "Chosen Court") in any action or proceeding arising out of or relating to this Voting Agreement or any of the Transactions, (b) agrees that all claims in respect of such action or proceeding shall be heard and determined in any such Chosen Court, (c) agrees that it shall not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such Chosen Court and (d) agrees not to bring any action or proceeding arising out of or relating to this Voting Agreement or any of the transaction contemplated by this Voting Agreement in any other court. Each of the parties hereto waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of any other party with respect thereto. Any party may make service on another party by sending or delivering a copy of the process to the

party to be served at the address and in the manner provided for the giving of notices in Section 4.4. Nothing in this Section 4.5, however, shall affect the right of any party to serve legal process in any other manner permitted by law. EACH OF SHAREHOLDER, PARENT AND MOBILE X HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS VOTING AGREEMENT OR THE TRANSACTIONS OR THE ACTIONS OF SHAREHOLDER, PARENT OR MOBILE X IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS VOTING AGREEMENT.

- 4.6 <u>Publicity.</u> The Shareholder acknowledges and represents that it supports the Merger pursuant to the terms of the Merger Agreement and authorizes each of the Parent and Mobile X to make a general statement to that effect (including Shareholder's identity and holding of Shares and the related obligations of Shareholder hereunder) in press releases and public filings that may be made with the SEC, Nasdaq or other regulatory authorities.
- 4.7 Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement was not performed in accordance with its specific terms or was otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Minnesota court, without proof of damages, prior to the valid termination of this Voting Agreement, this being in addition to any other remedy to which they are entitled under this Voting Agreement, and (b) the right of specific enforcement is an integral part of the transactions contemplated by this Voting Agreement and without that right, none of the parties would have entered into this Voting Agreement. Each party agrees that it will not oppose the granting of specific performance and other equitable relief on the basis that the other parties have an adequate remedy at Law or that an award of specific performance is not an appropriate remedy for any reason at Law or equity. The parties acknowledge and agree that any party seeking an injunction to prevent breaches of this Voting Agreement and to enforce specifically the terms and provisions of this voting Agreement in accordance with this Section 4.7 shall not be required to provide any bond or other security in connection with any such injunction.
- 4.8 <u>Entire Agreement.</u> This Voting Agreement constitutes and contains the entire agreement and understanding of the parties with respect to the subject matter hereof and supersedes any and all prior negotiations, correspondence, agreements, understandings, duties or obligations between the parties respecting the subject matter hereof.
- 4.9 <u>Counterparts.</u> This Voting Agreement may be executed in facsimile, pdf or other electronic means and in or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 4.10 <u>Further Assurances.</u> Shareholder agrees, from time to time, and without additional consideration, to execute and deliver such additional proxies, documents, and other instruments and to take all such further action as the Parent and Mobile X may reasonably request to consummate and make effective the transactions contemplated by this Voting Agreement.

- 4.11 No Agreement Until Executed. Irrespective of negotiations among the parties or the exchanging of drafts of this Voting Agreement, this Voting Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties unless and until (a) the Mobile X Board of Directors has approved, for purposes of any applicable provision of the Mobile X's organizational and governance documents, the transactions contemplated by the Merger Agreement (including the Merger), (b) the Merger Agreement is executed by all parties thereto, and (c) this Voting Agreement is executed by all parties hereto.
- 4.12 <u>Captions.</u> The captions to sections of this Voting Agreement have been inserted for identification and reference purposes only and shall not be used to construe or interpret this Voting Agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed by their duly authorized respective officers as of the date first above written. MOBILE X GLOBAL, INC. By: /s/ Dan Cohrs Name: Dan Cohrs Title: Chief Financial Officer ELECTRO-SENSORS, INC. /s/ David L. Klenk Name: David L. Klenk Title: Chief Financial Officer THE SHAREHOLDER: Signature: /s/ Signed by Shareholders on Schedule 1-B Printed Name: the Shareholder's Address for Notice: Outstanding \underline{Shares} of the Mobile X Common Stock Beneficially Owned by the Shareholder:_

10

by the Shareholder:

Outstanding Options to Acquire Shares of Mobile X Common Stock Owned

Schedule 1-B to Company Voting Agreement

Equity			
Investor	Common Shares		
Peter Adderton	3,000.00		
Stuart Bird	713.33		
Peter Hill	777.00		
Dan Cohrs	356.67		

EX-99.1 electro220769_8k_FINAL

Exhibit 99.1

06/14/2022 04:19 PM

1 of 5

Electro-Sensors and Mobile X Global Announce Merger

Public company to launch Mobile X, a disruptive new wireless service provider

Minnetonka, MN and Los Angeles, CA – June 13, 2022: Electro-Sensors, Inc. (Nasdaq: ELSE), a leader in industrial monitoring systems, and Mobile X Global, Inc., one of the world's first connectivity-as-a-service companies, announced today that they have entered into definitive agreements to merge and re-name the combined company Mobile X Global, Inc.

Mobile X Global, Inc. will launch Mobile X, a new wireless brand positioned to disrupt the wireless industry. The merger partners also expect Mobile X Global's cloud-native platform to enable synergies between Mobile X and Electro-Sensors' wireless industrial sensor business.

The definitive merger agreement and related transactions have been approved by each company's board of directors and are subject to approval by the shareholders of each company. The directors, officers, and major shareholders of each company, who collectively own a majority of the shares of each company, have entered into agreements obligating them to approve the transactions.

Shares of the combined company are expected to continue to trade on the Nasdaq Capital Market under the ticker symbol "MOBX" upon closing of the merger, expected in the third quarter of 2022.

Launch of Mobile X

In October 2021, Mobile X Global announced a network access agreement to enable the launch of a new AI-powered mobile business for wireless customers. The company will use its proprietary CaaS-AI (Connectivity-as-a-Service-AI) cloud-native platform, XO.1, and its network agreement, to provide personalized wireless data, voice, and messaging services, initially to customers nationwide across the U.S., with a commercial launch expected later this year. The company expects to further expand internationally under the Mobile X brand, leveraging the XO.1 platform capabilities to create a seamless global connectivity experience for its customers.

Mobile X intends to employ artificial intelligence, machine learning, and real-time usage information to learn each customer's unique connectivity requirements and provide optimized individual pricing, supported by a premier app-enabled experience. The seamless experience and potential cost savings for the majority of customers are expected to lead the industry, enabling customers to pay only for the services they use. The XO.1 platform will allow customers to seamlessly switch across multiple global networks, with one number and one service that extends beyond borders.

"This merger, like the one that I engineered to create Digital Turbine, will position us for rapid growth," stated Peter Adderton, founder and CEO of Mobile X Global. "We believe the new combined company will also provide the vehicle for future expansion, so that we can serve a new generation of borderless consumers and use our proprietary XO.1 platform to help connect the billions of IoT devices expected worldwide. Mobile X customers will get unprecedented levels of customized service and only pay for what they use. We expect to help millions of wireless customers save money at a time when they really need it."

Merger Structure and Ownership

After the merger, Mobile X Global shareholders are expected to own approximately 76% of the combined company, Electro-Sensors approximately 11%, and new equity investors approximately 13%, all based on current ownership of the two companies and \$20 million of new equity on the terms in a commitment letter described below. In addition to their continuing interest in the combined company, Electro-Sensors shareholders as of a record date to be determined before the closing will receive special cash dividends expected to total approximately \$18 million, with the actual amount of the dividends subject to adjustment based on the transaction expenses, working capital balance, and any indebtedness of Electro-Sensors at closing.

Cash dividends of \$18 million would be approximately \$4.83 per fully diluted share of Electro-Sensors, whose closing price on June 10, 2022, was \$4.85. In addition, the continuing ownership of Electro-Sensors' legacy shareholders will give them the opportunity to participate in the long-term value to be created by the combined company through Mobile X's planned disruptive entry into wireless communications, synergies expected from enhancing Electro-Sensors' wireless sensor business with the capabilities of Mobile X Global's XO.1 platform, the increased scale of the new organization, and the sharing of best practices.

In connection with the merger, a third-party institutional investor has entered into a commitment letter with Mobile X Global to provide equity financing of up to \$20 million upon closing of the merger, subject to diligence and definitive agreements satisfactory to the investor, including an agreement for a \$50 million equity line of credit to be provided by the investor. The equity line of credit will provide significant additional liquidity, at the option of Mobile X Global.

Electro-Sensors Voting Agreement and Special Meeting of Shareholders

In connection with the execution of the merger agreement, Electro-Sensors' directors, officers, and major shareholders, who collectively own a majority of Electro-Sensors' outstanding shares, have entered into agreements to vote their shares in favor of the merger at a special meeting of shareholders to be held before the closing on a date to be announced.

Closing will follow the special meeting of shareholders of Electro-Sensors, consent of shareholders of Mobile X Global, and satisfaction of other customary closing conditions, including the U.S. Securities and Exchange Commission (the "SEC") having declared effective a registration statement, and The Nasdaq Stock Market having approved the listing of the common stock of the combined company.

David Klenk, CEO of Electro-Sensors, said, "We believe this merger will be very good for the customers, employees and shareholders of Electro-Sensors. Our shareholders will receive a significant cash dividend and retain a meaningful share of an exciting platform company that we expect will also create new opportunities for our sensor business."

Other Information about the Merger and Related Transactions

The transaction is structured as a statutory reverse triangular merger under Delaware and Minnesota law, under which a newly formed subsidiary of Electro-Sensors, Inc. will be merged with and into Mobile X Global, Inc., with Mobile X Global, Inc. surviving the merger and becoming a wholly owned subsidiary of Electro-Sensors, Inc. Electro-Sensors, Inc. will reincorporate in Delaware, be re-named Mobile X Global, Inc., and operate both the new Mobile X wireless business and the Electro-Sensors business.

Additional information about the proposed transaction, including a copy of the merger agreement, will be provided in a Current Report on Form 8-K to be filed by Electro-Sensors with the SEC and made available at www.sec.gov. More information about the proposed transaction will also be described in Electro-Sensors' registration statement and related documents relating to the merger, which it will file with the SEC.

Management and Governance

Upon closing of the merger, Peter Adderton, chairman and chief executive officer of Mobile X Global, will become the chairman and chief executive officer of the combined company. Mr. Adderton has decades of experience creating and operating digital platform businesses, including his former roles with Digital Turbine (Nasdaq: APPS), which currently has a market capitalization of approximately \$1.7 billion and mobile virtual network operator (MVNO) Boost Mobile. Mobile X Global's management team has decades of experience operating large telecom providers and rapidly scaling private and public companies. Mobile X Global and Electro-Sensors are committed to retaining Electro-Sensors' existing employees and customers. Senior management of Electro-Sensors is expected to continue and assist in the integration of the combined company and the operation of the existing sensor business.

When the transaction closes, the current directors of Electro-Sensors will resign, and the Board of Directors of the combined company will initially consist of five members appointed by Mobile X Global.

Advisors

B. Riley Securities is financial advisor and Proskauer Rose LLP is legal counsel to Mobile X Global on the transactions. Lake Street Capital Markets is financial advisor and Ballard Spahr, LLP is legal counsel to Electro-Sensors for the transactions.

About Mobile X Global, Inc.

Mobile X Global, Inc., one of the world's first connectivity-as-a-service-AI companies, is a new entrant in the global mobile industry founded by Peter Adderton, who also founded Boost Mobile and Digital Turbine (Nasdaq: APPS). Mobile X Global plans to launch a new mobile brand called Mobile X in the U.S. in 2022, designed to attract displaced and mistreated wireless customers by offering simple, great value prepaid wireless services and experiences. Enabled by a network access agreement with a major carrier and supported by Mobile X Global's proprietary XO.1 cloud platform and an AI-powered on-device application, Mobile X will offer each customer their own unique service and ensure they only pay for what they use.

About Electro-Sensors, Inc.

Electro-Sensors, Inc. is an industry leading designer and manufacturer of rugged and reliable machine monitoring sensors and wireless/wired hazard monitoring systems applied across multiple industries and applications. These products improve processes by protecting people, safeguarding systems, reducing downtime, and preventing waste. Electro-Sensors is proud to be an ISO9001:2015 quality certified company and is committed to providing excellent customer service and technical support. Founded in 1968 and located in Minnetonka, Minnesota, Electro-Sensors provides its loyal customers with reliable products that improve safety and help plants operate with greater efficiency, productivity, and control.

Important Information and Where to Find It

A full description of the terms of the transaction will be provided in a proxy statement/prospectus/consent solicitation statement included in a Form S-4 Registration Statement that Electro-Sensors will file with the SEC. ELECTRO-SENSORS URGES INVESTORS, SHAREHOLDERS AND OTHER INTERESTED PERSONS TO READ, WHEN AVAILABLE, THE PROSPECTUS/PROXY STATEMENT/CONSENT SOLICITATION STATEMENT AS WELL AS OTHER DOCUMENTS FILED WITH THE SEC BECAUSE THESE DOCUMENTS WILL CONTAIN IMPORTANT INFORMATION ABOUT ELECTRO-SENSORS, MOBILE X GLOBAL, AND THE PROPOSED TRANSACTIONS. When final, the definitive proxy statement will be mailed to Electro-Sensors' shareholders as of a record date to be established for voting on the proposed transaction, and a definitive consent solicitation statement will be sent to the Mobile X shareholders. Shareholders will also be able to obtain a copy of the documents (when available), without charge, by directing a request to: Electro-Sensors, Inc., 6111 Blue Circle Drive, Minnetonka, MN 55343. These documents, once available, can also be obtained, without charge, at the SEC's website (www.sec.gov).

Participants in the Solicitation

Electro-Sensors, Inc., Mobile X Global, Inc., and their respective directors and executive officers may be considered participants in the solicitation of proxies by Electro-Sensors, Inc. in connection with the proposed transaction. Information about the directors and executive officers of Electro-Sensors, Inc. is set forth in its Annual Report on Form 10-K for the fiscal year ended December 31, 2021, and its 2022 Proxy Statement, which were filed with the SEC on March 31, 2022. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies from Electro-Sensors' shareholders in connection with the proposed merger will be included in the definitive proxy statement/prospectus that Electro-Sensors intends to file with the SEC.

Non-Solicitation

This press release does not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which the offer, solicitation or sale would be unlawful prior to the registration or qualification under the securities laws of that jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended, or an exemption therefrom.

Forward Looking Statements

This press release includes certain forward-looking statements concerning Electro-Sensors, Mobile X Global and the proposed transactions within the meaning of the Private Securities Litigation Reform Act of 1995, including, but not limited to, statements regarding future financial performance, future growth, and the development of future products and services; the benefits of the proposed transactions, including anticipated growth and synergies; the combined company's plans, objectives and expectations and intentions; the expected timing of the proposed transactions; and future acquisitions. These statements are based on current expectations or beliefs and are subject to uncertainty and changes in circumstances. There can be no guarantee that the proposed transactions described in this press release will be completed, or that they will be completed as currently proposed, or at any particular time. Actual results may vary materially from those expressed or implied by the statements here due to changes in economic, business, competitive or regulatory factors, and other risks and uncertainties affecting the operation of Electro-Sensors as well as the business of Mobile X Global. Many of these risks, uncertainties and contingencies related to Electro-Sensors are presented in Electro-Sensors' Annual Report on Form 10-K and, from time to time, in Electro-Sensors' other filings with the SEC. These and other risks related to the business of Mobile X Global will be presented in the proxy statement/prospectus/consent solicitation statement to be filed with the SEC.

The information here should be read considering these risks and the following considerations: the ability of the merger parties to obtain definitive investment documents and close on the equity investments necessary to complete the merger; the ability of Mobile X to successfully launch its business, attract subscribers, and achieve the levels of customer service, revenues and costs that it currently expects; the ability of the combined company to successfully maintain a Nasdaq Capital Market listing; the ability of the combined company to successfully access the capital markets to finance expansion and acquisitions; the ability of the combined company to identify and acquire appropriate acquisition targets and successfully integrate these companies into its operations; the ability of the combined company to achieve synergies between its legacy sensor business and its new Mobile X business; the conditions to the closing of the merger may not be satisfied or an event, change or other circumstance could occur that could give rise to the termination of the merger agreement; the merger may involve unexpected costs, liabilities or delays, resulting in the merger not being consummated within the expected time period; risks that the announced merger may disrupt current Electro-Sensors plans and operations or that the business or stock price of Electro-Sensors may suffer as a result of uncertainty surrounding the merger; the outcome of any legal proceedings related to the merger; and Electro-Sensors or Mobile X Global may be adversely affected by other economic, business, or competitive factors.

###

MEDIA CONTACT

Aurli Bokovza Illume Public Relations 917.319.8441